School of Design and the Built Environment Curtin University Sustainability Policy Institute

Low carbon transition through innovation at the university-industry-government nexus

Mike Burbridge

This thesis is presented for the Degree of Doctor of Philosophy of

Curtin University

January 2022

Declaration

To the best of my knowledge and belief this thesis contains no material previously published by any other person except where due acknowledgement has been made.

This thesis contains no material which has been accepted for the award of any other degree or diploma in any university.

Human ethics

The research presented and reported in this thesis was conducted in accordance with the National Health and Medical Research Council National Statement on Ethical Conduct in Human Research (2007) – updated March 2014. The proposed research study received human research ethics approval from the Curtin University Human Research Ethics Committee (EC00262), Approval Number RDHU-49-16.

Signature of candidate:

Signature Redacted

Statement of contributors

I, Mike Burbridge have conceived and coordinated all the written materials submitted as part of this PhD by publication (hybrid). I also undertook the majority of the writing and analysis for each publication.

Signature of candidate:

Signature Redacted

Co-authors' statements

Publication 1

I, Mike Burbridge, contributed 90% to the publication entitled:

Burbridge, M. & G. Morrison (2021)

A systematic literature review of partnership development at the universityindustry-government nexus.

Signature of candidate

Signature Redacted

Date: 11 January 2022

I, as a co-author, endorse that this level of contribution by the candidate indicated above is appropriate.

Co-author: Gregory M. Morrison

Signature of co-author:

Signature Redacted

Date: 11 January 2022

Publication 2

I, Mike Burbridge, contributed 90% to the publication entitled:

<u>Burbridge, M.</u>, G Morrison, M. van Rijn, S. Silvester, D. Keyson, L. Virdee, & C. Baedeker (2016).

Business models for sustainability in living labs. In Living Labs: Design and Assessment of Sustainable Living, pp 391-403.

Signature of candidate

Signature Redacted

I, as a co-author, endorse that this level of contribution by the candidate indicated above is appropriate. I also confirm that the other co-authors provided data for the publication but provided no other contribution towards the publication.

Co-author: Gregory M. Morrison

Signature of co-author:

Signature Redacted

Date: 11 January 2022

Publication 5

I, Mike Burbridge, contributed 90% to the publication entitled:

Burbridge, M. & G Morrison (2016)

Happy Homes – the relationship between homes and mental wellbeing.

Proceedings of the 7th International Conference on Energy and Environment of Residential Buildings, edited by Miller, W., Susilawati, C. and Manley, K. Brisbane: Queensland University of Technology, Australia.

Signature of candidate



Date: 11 January 2022

I, as a co-author, endorse that this level of contribution by the candidate indicated above is appropriate.

Co-author: Gregory M. Morrison

Signature of co-author:

Signature Redacted

Abstract

The industrialised world is continuing along a path of economic transformation. Economic growth is increasingly reliant on the development and exploitation of knowledge. This places those who create knowledge and those who commercialise it in a prominent role, as well as being the focus of government attention. Evidence from best practice has led to the current discourse that suggests that partnerships at the universityindustry-government nexus are required to effectively commercialise knowledge and enable innovation.

This research demonstrates that in the literature there are two typologies of partnership for innovation at the university-industry-government nexus. One is referred to as top-down and involves the creation of largely new structures by management to promote the commercialisation of knowledge. The second form of structure is less well defined but is referred to as bottom-up. Bottom-up structures are initiated from within the partnership and are characterised as being user centred with the adoption of open innovation. One example of the bottom-up artefact is the living lab.

This research finds, through the adoption of a systematic approach to the analysis of latent data, that top-down partnership structures are primarily focused on the economic benefits of innovation. The research demonstrates that a reason for this focus is that top-down structures have their origins from a time when knowledge production was more discipline based (Mode 1) and before the widespread democratisation of knowledge or globalisation of the environment. The research suggests that the phenomenon of top-down partnerships remaining primarily focused on economic growth can be partly explained through path dependence theory.

This research also demonstrates that it is only bottom-up structures that provide the innovation to enable a low carbon and sustainable transition. The research suggests that one reason for this is that bottom-up structures were created in a period when knowledge production was more transdisciplinary (Mode 2) and at a time when the globalisation of society was giving global citizenry a greater voice (resulting in Mode 3 knowledge production) with the globalisation of the environment concomitantly receiving greater policy attention.

vi

The research also demonstrates that there is nothing preventing top-down partnerships from pursuing low carbon or sustainable innovation, which suggests the reasons for lack of sustainable innovation are institutional rather than structural.

This research proposes a theoretical metastructure to enable innovation at the university-industry-government nexus. The metastructure integrates the evolution of partnerships with the globalisation of the economy, society and environment and the changes in modes of knowledge production. This results in three streams of partnership development. Stream 1 involves the development of on-campus structures; stream 2 is the development of campus adjacent structures and stream 3 is the creation of living labs (and derivatives).

The three streams help to explain why top-down structures have a primary economic focus. To address this issue the research proposes an institutional framework to enable Stream 1 and Stream 2 partnership structures to evolve through the application of knowledge, to partnership structures that embrace co-creation and open innovation to enable a low carbon and sustainable transition through innovation at the university-industry-government nexus.

Acknowledgments

They say it takes a village to raise a child; well, the same is true for completing a PhD. I have been fortunate to be showered with love and support during the creative process that has led to the submission of this work.

I must first start by thanking my wife and family for their love and unwavering support over the last few years. Cathy has just been amazing. She has encouraged me to continue with the work, especially at the darkest points, when I have been riddled with self-doubt. But she also somehow kept me grounded and helped me understand the need to focus on the work that I needed to do rather than what I wanted to do. I am so lucky and grateful. Abigail and Greg too have always shown me real encouragement and support. To be honest, I find it difficult to put into words my gratitude to Cathy, Abigail and Greg for their love, support and understanding. I look forward to being able to do more things together.

I also want to thank Greg Morrison for his calm guidance and thoughtful help. I have learnt a lot and I do really appreciate all he has done to help get this across the line. I will always be amazed by his ability to focus and somehow turn on his brain to produce deeply insightful ideas whilst at the same time focusing on the detail. I also would like to thank Peter Newman as this PhD came to life from discussions with him many years ago. I have always valued Peter's positive approach to life and the academic endeavour. I would also like to acknowledge Fritz Kalkbrenner who has inspired me to some of my most creative moments.

I am now a teacher, and I am only now starting to really understand the impact my teachers have had on my life. Nigel Sinclair was my science teacher at Sompting Abbotts in the late 1970s. He taught me that matter can neither be created nor destroyed. The quote maybe wrong, but that sentiment has been with me through all my academic endeavours and still drives my thinking and in many ways helped me to understand sustainable development. John Plowright was my history teacher in 1985 and a truly decent man. He cared about me and somehow helped me to appreciate, as a teenager, the pleasure that comes from learning and gave me the confidence and desire to learn.

viii

There are many others who have offered unwavering support. You know who you are, and I am very grateful for the confidence you inspired in me.

I would also like to thank the Australian Cooperative Research Centre for Low Carbon Living (CRC LCL) for providing my PhD Scholarship, organising opportunities to meet other PhD candidates involved in low carbon-related research, and providing valuable training to inform my research.

List of publications

The following are the publications resulting from the research for this thesis.

Publication 1: Burbridge & Morrison (2021)

Burbridge, M. & G. Morrison (2021)

A systematic literature review of partnership development at the universityindustry-government nexus.

Sustainability 13, no. 24: 13780. https://doi.org/10.3390/su132413780

Publication 2: Burbridge, Morrison, van Rijn, Silvester, Keyson, Virdee, & Baedeker (2016)

<u>Burbridge, M.</u>, G Morrison, M. van Rijn, S. Silvester, D. Keyson, L. Virdee, & C. Baedeker (2016).

Business models for sustainability in living labs. In: Keyson D., Guerra-Santin O., Lockton D. (eds) Living Labs. Springer, Cham. https://doi.org/10.1007/978-3-319-33527-8_30

Publication 3: Burbridge (2017)

Burbridge, M. (2017)

If living labs are the answer – what's the question?

Procedia Engineering, Vol 180 pp 1725 – 1732

Publication 4: Burbridge (2018)

Burbridge, M. (2018)

A Study of Australian Universities' Collective Response to Climate Science.

In: Sayigh A. (eds) Transition Towards 100% Renewable Energy. Innovative Renewable Energy. Springer, Cham. https://doi.org/10.1007/978-3-319-69844-1

Publication 5: Burbridge & Morrison (2016)

Burbridge, M. & G Morrison (2016)

Happy Homes – the Relationship between Homes and mental wellbeing.

Proceedings of the 7th International Conference on Energy and Environment of Residential Buildings, edited by Miller, W., Susilawati, C. and Manley, K. Brisbane: Queensland University of Technology, Australia.

Other published work

Publication: <u>Burbridge, M.</u> (2017)

Australian Universities are not walking the talk on going low carbon.

https://theconversation.com/australias-universities-are-not-walking-the-talk-ongoing-low-carbon-72411

Publication: Wardell-Johnson, G., Wardell-Johnson, A., Bradby, K., Robinson, T., Bateman, P.W., Williams, K., Keesing, A., Braun, K., Beckerling, J. and <u>Burbridge, M.</u> (2016)

Application of a Gondwanan perspective to restore ecological integrity in the south-western Australian global biodiversity hotspot.

Restoration Ecology, 24: 805-815. https://doi.org/10.1111/rec.12372

Table of contents

Abstract	ıstractvi		
Acknowl	edgmer	ıts	viii
List of pu	ublicatio	ins	x
Table of	content	S	xii
List of ta	bles		xvii
List of fig	gures		xviii
List of ab	breviat	ions	xix
Chapter	1 In	troduction	1
1.1	Contex	t	1
1.2	Backgr	ound	1
1.3	Resear	ch positioning	2
1.4	Resear	ch questions and objectives	3
1.5	Thesis	organisation	5
Chantor	2 Su	nthosis of literature review	6
Chapter	2 Jy		0
2.1	Backgr	ound	6
2.2	Theory	of partnership development at the university-industry-	
	govern	ment nexus	8
2.2.	1 Tri	ple Helix	9
2.2.2	2 Qı	ıadruple Helix	10
2.2.3	3 Qı	iintuple Helix	11
2.3	The cre	ation of hybrid organisations	12
2.4	Partne	rships delivering economic, social and environmental outcomes	13
2.5	Develo	ping expertise in partnership development	13
Chapter	3 M	ethodology	15
3.1	Resear	ch methodology	15
3.2	Resear	ch design	16
3.2.	1 Sv	stematic literature review	
		Publication 1, Appendix B.1	17
3	.2.1.1	Discussion of the analysis of latent qualitative data	17
3.2.2	2 Ca	se study	
	— F	<i>,</i> Publication 2, Appendix B.2	18
3.2.3	3 Na	irrative Review and Inductive reasoning	
	_	Publications 3 and 5, Appendix B.3 and B.5	20
3.2.4	4 Qı	antitative data collection and analysis	
	_	Publication 4, Appendix B.4	21
3.2.	5 Ca	se study: semi structured interviews - Chapter 5	22

Chapter	4	Publication results and discussion	. 25
4.1	A sy	stematic literature review of partnership development at the	
	univ	versity-industry-government nexus	26
4.1.	1	Summary of findings	26
4.1.	2	Partnership development at the university-industry-government nexus	27
4.1.	3	Discussion	29
4.1.	4	Progression in research findings	29
4.2	Bus	iness models for sustainability in living labs	30
4.2.	1	Summary of findings	30
4.2.	2	Business case development for low carbon innovation	31
4.2.	3	Discussion	32
4.2.	.4	Progression in research findings	33
4.3	If liv	ving labs are the answer what is the question?	34
4.3.	1	Summary of findings	34
4.3.	2	Partnership development for low carbon innovation	35
4.3.	3	Discussion	36
4.3.	4	Progression in research findings	37
4.4	A st	udy of Australian universities' collective response to climate science	38
4.4.	1	Summary of findings	38
4.4.	2	Developing partnerships based on experience and knowledge	39
4.4.	3	Discussion	40
4.4.	4	Progression of research findings	40
4.5	Нар	py Homes – the relationship between homes and mental wellbeing	41
4.5.	1	Summary of findings	41
4.5.	2	Co-creating partnerships based on knowledge and	
		shared implementation.	42
4.5.	3	Discussion	43
4.5.	.4	Progression of research findings	43
Chantor	- 5	Innovation at the university industry-government nexus	
Chapter	5	The case of Greater Curtin	. 45
F 4			
5.1	вас	kground	45
5.1.	.1	Greater Curtin - Master Plan	46
5.1.	.2	Greater Curtin – student accommodation	47
5.2	Sun	nmary of findings	48
5.3	Disc	cussion of findings	48
5.3.	1	Theme 1 - evolution of the project over time	49
5.3.	2	Theme 2 – engage, integrate and showcase research	
	_	and teaching strengths	51
5.3.	.3	Theme 3 - potential to progress Mission 1, 2 and 3 outcomes	52

5.4	Development of an institutional framework to transition	
	top-down structures	56
5.5	Discussion of proposed institutional framework	60
5.6	Further research	61
		~~~
Chapte	r 6 Synthesis	62
6.1	Research progression addressing the sub-questions	62
6.1	1 What partnership models at the university-industry-government	
	nexus have been successful to deliver innovation?	66
6.1	2 What partnership models at the UIG nexus have been successful to	
	deliver low carbon and/or sustainable innovation?	68
6.1	3 What factors promote or limit partnership development for a	
	low carbon future?	69
6.1	4 How can Australian universities become more effective at partnering	
	with government and industry to drive a low carbon future?	70
6.2	Summary of evidenced research contribution to each sub-question	71
6.2	2.1 Contribution: Sub-question 1	71
6.2	2.2 Contribution: Sub-question 2	72
6.2	2.3 Contribution: Sub-question 3	72
6.2	2.4 Contribution: Sub-question 4	73
Chauta		
Chapte	r / A new metastructure to enable innovation at the university-	
	Industry-government nexus	74
7.1	Realising the potential of a Triple Helix partnership	74
7.2	Summary of current models proposed to drive low carbon innovation	75
7.3	How can universities, industry and government together drive low	
	carbon transition through innovation?	75
7.4	Implications of findings	76
7.5	Limitation of proposal	78
7.6	Development of a metastructure to enable innovation at the university-	
	industry-government nexus	78
7.7	Integrating exogenous factors and modes of knowledge production into	
	the metastructure	79
7.7	7.1 Stream 1: Innovation Partnerships – on campus developments	82
7.7	2.2 Stream 2: Development of campus adjacent structures	83
7.7	'.3         Stream 3: Living labs and derivatives	84
7.8	Summary of the emerging metastructure to enable innovation at the	
	university-industry-government nexus	84
7.9	Integrating the metastructure and institutional framework	85
7.10	Using the metastructure and institutional framework to enable low	
	carbon transition at the university-industry-government nexus	86
7.11	Further research	87

Chapte	r 8	Conclusions and recommendations	
8.1	Sum	nmary of research findings	88
8.2	Orig	ginal and significant contribution	
8.3	Furt	her research	92
Exegesis	s Refe	erences	94
Append	lix A	Extended literature review	106
A.1	The	changing role of knowledge in the economy	
A.1	.1	Knowledge as a management revolution	109
A.1	.2	Realigning the role of universities, industry and government	110
A.1	.3	Entrepreneurial universities	111
A.1	.4	Changing modes of knowledge production	112
A.2	Inno	ovation and government policy	114
A.3	Aus	tralian innovation policy and performance	115
A.4	A ch	nanging climate	116
A.4	.1	Sustainable development - society's response to a changing wor	rld 116
A.5	Inno	ovation and Responsible Research and Innovation	119
A.5	.1	Responsible innovation	119
A.5	.2	Responsible innovation and Australia	121
A.6	Eco	nomics of innovation – the rise of partnerships for innovation	122
A.7	Part	nerships for innovation	127
A.8	Trip	le Helix model	127
A.8	.1	Theoretical foundations of the Triple Helix model	130
A.8	.2	Criticism of the Triple Helix model	134
A.9	Qua	druple Helix	136
A.10	Qui	ntuple Helix	137
A.1	0.1	Criticism of the Quintuple Helix	138
A.11	Trip	le, Quadruple and Quintuple models: Summary	140
A.1	1.1	Partnerships for innovation – the creation of hybrid organisation	ns 140
A.1	1.2	Concluding remarks	142
Append	lix B	Publications	143
B.1	Pub	lication 1	143
B.2	Pub	lication 2	167
B.3	Pub	lication 3	
B.4	Pub	lication 4	
B.5	Pub	lication 5	

Append	dix C Copyright	
C.1	Publication 1	
C.2	Publication 2	210
C.3	Publication 3	213
C.4	Publication 4	214
C.5	Publication 5	217

# List of tables

Table 1.1	Summary of the research sub-questions, their objectives and	
	associated publications	4
Table 2.1	Synthesis of literature reviewed	7
Table 2.2	Changes in the modes of knowledge production	8
Table 2.3	Triple helix model structures	10
Table 2.4	Triple Helix outcome development	12
Table 2.5	Innovation intermediaries in society (Publication 1)	13
Table 3.1	Research methods reported in the publications	15
Table 3.2	Modification of Massey's [56] model for thematic analysis of	
	latent data	19
Table 3.3	Case study participants	20
Table 3.4	Meta questions influencing semi structured interviews	23
Table 3.5	Themes and associated nodes identified in the thematic analysis	23
Table 4.1	Summary of how publications address the research questions	25
Table 4.2	Hypothesis of current relationship between partnerships, globalisation	
	and modes of knowledge production	30
Table 4.3	Summary of individual SusLab purpose	31
Table 4.4	Summary of each SusLab's Business Case	32
Table 4.5	SusLabNWE—informal value to partners in the SusLabs of the Triple Helix	.33
Table 4.6	Collaboration between Australian universities and business	36
Table 4.7	Australian universities commitments to carbon constraint	39
Table 5.1	Projected benefits of Greater Curtin to be realised by 2031	46
Table 5.2	Developing network strengths through Greater Curtin master plan	47
Table 5.3	Proposed institutional framework for partnership development at the	
	UIG nexus to deliver teaching and research outcomes (TRO)	58
Table 6.1	Research progression through the thesis	63
Table 6.2	Partnership structures delivering innovation with economic, social and	
	environmental outcomes	67
Table 6.3	Hypothesis of relationship between partnerships, globalisation and	
	modes of knowledge production	68
Table 7.1	Streams of partnership development in the metastructure	79
Table 7.2	Development of the metastructure to enable innovation at the	
	university-industry-government nexus	80
Table 7.3	Three stream metastructure of partnership development	81
Table 7.4	Hypothesis of partnership response to external stimuli	82
Table 7.5	Stream 1 – on campus structures	83
Table 7.6	Forms of university-industry partnership	83
Table 7.7	Emerging framework for the creation of intermediaries for innovation	
	at the university-industry-government nexus	85
Table 7.8	Relationship between proposed metastrcuture, institutional	
	framework and modes of knowledge production	86
Table 7.9	Relationship between evolution of partnership streams and top-	
	down/bottom-up partnership creation	86
Table 8.1	Summary of evidenced findings from the body of work	89

## List of tables in appendices

Table A.1	Relationship between the literature review, research questions and the	
	exegesis discussion	7
Table A.2	Technological transformations since the 18th century 109	9
Table A.3	The evolving state of the university 112	1
Table A.4	Aspects of learning impacted by transition to an entrepreneurial university 112	2
Table A.5	Modes of knowledge production 113	3
Table A.6	A partial record of the growths and impacts of human activities during	
	the 20th century 117	7
Table A.7	United Nations' Sustainable Development Goals 118	8
Table A.8	OECD's 4 types of innovation 120	D
Table A.9	European RRI - 6 streams of work 122	1
Table A.10	Lundvall's [22] four types of institution important for learning and innovation 12	5
Table A.11	Resources fundamental for economic growth - combining the tangible	
	and reproducible dimensions 126	6
Table A.12	Styles of innovation 122	7
Table A.13	The evolution of the Triple Helix model of innovation 129	9
Table A.14	Themes of reflexive controls	1
Table A.15	Five aspects of a normative balanced Triple Helix model	2
Table A.16	Examples of the Triple Helix model being applied (Publication 1) 133	3
Table A.17	Institutional orders in the evolution of the Triple Helix model	5
Table A.18	The 5-step flow analysis of Innovation under a Quintuple Helix model 139	9
Table A.19	Definition of helices in Triple, Quadruple and Quintuple models 142	1
Table A.20	Extant top down/bottom-up approaches to partnerships for innovation 142	2

# List of figures

Figure 3.1	Interviewee's involvement in student accommodation project	24
Figure 4.1	Innovation intermediaries at the university-industry-government nexus	27

# List of abbreviations

CSIRO	Commonwealth Scientific and Industrial Research Organisation
EC	European Commission
EOI	Expression of interest
EU	European Union
FSP	Future Science Platform
GDP	Gross domestic product
IP	Intellectual property
ISA	Innovation and Science Australia
KPI	Key performance indicator
LL	Living Lab
моос	Massive open online course
NISA	National Innovation and Science Agenda
OECD	Organisation for Economic Cooperation and Development
PCG	Project Control Group (responsible for delivery of Stage 1 Greater Curtin)
RFDP	Request for detailed proposal
RI	Responsible Innovation (Australian vernacular for RRI)
RIFSP	Responsible Innovation Future Science Platform
RRI	Responsible Research and Innovation (RI in Australia)
SDG	Sustainable development goals
SLR	Systematic literature review
STP	Science and Technology Park
SusLab	Sustainability Living Lab
UIG	University-industry-government
ULL	Urban Living Lab
UN	United Nations

## Chapter 1 Introduction

This thesis will investigate low carbon transition through innovation enabled by partnerships for innovation. The research is based on a mixed methods approach and leads, as reported in Chapter 7, to the development of a new metastructure to enable innovation at the university-industry-government nexus.

This introductory Chapter provides the background information that gives rise to the research questions. These questions will be addressed in this thesis through five peer reviewed publications (see Appendix B.1 to B.5), which have been collated and integrated via this exegesis. Further unpublished work is presented in Chapter 5 and an extended literature review (unpublished and presented as Appendix A) also contributed to the findings evidenced in this exegesis.

#### 1.1 Context

In 1959, Peter Drucker [1] wrote about the emergence of a new form of economy – one that is based on knowledge. He did not argue that industrial economies were knowledge free rather that in the emerging knowledge economy, knowledge would become the primary resource for economic growth. He argued that knowledge, in and of itself, only has economic (rather than societal) value when it is used in a task. He predicted many tensions that would arise out of the transformation of knowledge into goods and services. Key amongst these tensions is:

- organisations' desire for autonomy versus society's need to pursue the common good;
- society's desire for stability versus organisations' need to innovate change;
- the strength of specialist knowledge and the needs of the team versus the rising demand for socially responsible organisations.

All of this he predicted would be the challenges for years to come.

#### 1.2 Background

The world's economy has become knowledge intensive, and with the advent of the internet, knowledge is becoming increasingly accessible to a range of stakeholders [2, 3]. In many industrialised economies this has seen the rise in importance of the service

sector for economic output with a reduced emphasis on manufacturing for increasing gross domestic product. In parallel there has been a movement of mass production to less developed economies with a supply of abundant and relatively cheap labour [4].

In such a post-industrial economy, which is increasingly focussed on the commercial exploitation of knowledge, the role of universities in creating and commercialising new knowledge is garnering increased interest in national [5] and international policy circles [6] as well as in academia itself [7]. The process of effectively commercialising new knowledge is seen to be the key to future economic development [8]. It is increasingly being recognised that this economic development needs to be achieved in a manner that is consistent with both the delivery of the United Nations' Sustainable Development Goals [9] and the Paris Climate Accord [10].

There is much in the academic literature concerning the transition to a knowledge economy (see Appendix A for an extended review of the literature and Chapter 2 provides a synthesis of the literature). However, what is missing from the research to date is a model that is designed to enable low carbon and sustainable transition through innovation to help society pursue the common good, deliver the Sustainable Development Goals and enable a low carbon future. A proposed metastructure to help achieve these goals is presented in Chapter 7.

#### 1.3 Research positioning

Before proceeding to the research evidenced in this thesis it is necessary to understand the processes that have led society to the current position.

Since the 1940s there has been a movement by governments to increase the uptake of university research by industry. Initially the focus was on defence and agriculture [7] but over time the purview widened and started to encompass not only policy processes (through funding and incentives, for example the passing in North America in 1980 of the Bayh-Dole Act [11]) but also the creation of structures to help commercialise innovation. This process has been ongoing for over 70 years [12]. In the last decade the globalisation of access to the internet meant that access to knowledge is more freely available [2]. This has led to the acceleration of the development of new forms of intermediaries for innovation and new forms of partnership [13].

2

There is a considerable body of literature that addresses the need to create partnerships between universities, industry and government with the intention to promote innovation for the further economic development of society [12]. There is extensive research looking at how partnerships – or intermediaries for innovation – have developed [12, 14] and how they are producing outputs that contribute towards economic growth. However, there is a much smaller strand of work – although it is growing (see Publication 1 and [12]) – that considers partnerships to enable innovation that is economically viable, socially just and environmentally sustainable. These intermediaries for innovation are a looser agglomeration of a range of partners. In their current form they do not follow set rules but evolve rapidly and currently they are only localised, although some urban areas have developed networks (Publication 1).

To develop innovation that is economically viable, socially just and environmentally sustainable a new metastructure is proposed in Chapter 7.

#### 1.4 Research questions and objectives

The research positioning leads to the overarching research question of this thesis:

How can universities, industry and government together drive low carbon innovation and transition?

Four sub-questions have been developed to support and elaborate on the primary research question to provide a framework for the research. These sub-questions are:

- <u>Sub-question 1</u>: Which partnership models at the university-industry-government (UIG) nexus have been successful in delivering innovation?
- <u>Sub-question 2</u>: Which partnership models at the UIG nexus have been successful in delivering low carbon and/or sustainable innovation?
- <u>Sub-question 3</u>: What factors promote or limit partnership development for a low carbon future?
- <u>Sub-question 4</u>: How can Australian universities become more effective at partnering with government and industry to drive a low carbon future?

To address these questions, five definitive bodies of work have been published with the full text provided in Appendix B.1 to B.5. The objectives derived from the questions, and the resulting publications are shown in Table 1.1. A summary of these publications makes up the main body of this thesis. In addition to these publications, Chapter 5 provides results from an unpublished case study of the development of student accommodation at Curtin University which contributes to the findings evidenced in his thesis.

Sub-questions	Objectives	Associated publications
Which partnership models at the UIG nexus have been successful in delivering innovation?	To identify the development of partnership models to create best practice partnerships To identify partnership structures that are being used to deliver a low carbon future.	Publication 1: A systematic literature review of partnership development at the university-industry- government nexus Published, peer reviewed article
Which partnership models at the UIG nexus have been successful to deliver low carbon and/or sustainable innovation?	To identify best practice in business case development in partnerships a the UIG nexus	Publication 2: Business models for sustainability in living labs Published, peer reviewed book chapter
What factors promote or limit partnership development for a low carbon future?	To identify those issues that prevent effective partnership creation	Publication 3: If living labs are the answer what is the problem? Published, peer reviewed article
How can Australian universities become more effective at partnering with government and industry to drive a low carbon future	To identify Australian universities' performance in partnering with industry and government; To identify the opportunities for universities to partner with industry, or government; and To identify universities' performance in adopting low carbon research outputs	Publication 4: A study of Australian universities' collective response to climate science Published, peer reviewed article Publication 5: Happy homes – the relationship between homes and mental wellbeing. Published, peer reviewed article

Table 1.1Summary of the research sub-questions, their objectives and associated<br/>publications

#### 1.5 Thesis organisation

The research undertaken in this thesis is presented in the format of a thesis by publication, with five published papers, and unpublished data integrated *via* this exegesis. A synthesis of the literature review is presented in Chapter 2; Chapter 3 presents the overall research methodology. The research summary for each publication is presented in Chapter 4 alongside a discussion of the progression of the research through the thesis. Chapter 5 presents a case study on a student accommodation project at Curtin University and develops an institutional framework to enable low carbon and sustainable innovation. Chapter 6 provides a synthesis of the research contribution to the literature. Chapter 7 presents a new metastructure to enable innovation at the university-industry-government nexus. Chapter 8 offers recommendations for further research. Appendix A presents an extended literature review and the publications themselves are presented in Appendices B.1 to B.5.

## Chapter 2 Synthesis of literature review

This Chapter provides a synthesis of the literature reviewed for this thesis. An extended literature review is presented in full in Appendix A and Appendix B.1 - Publication 1.

This synthesis is formed around the sub-questions that are central to the theoretical findings presented in Chapter 4 and Chapter 6. Section 2.1 provides a summary of the underlying theory that has led to this research. Section 2.2 discusses the theory of partnership development at the university-industry-government nexus; Section 2.3 focusses on the creation of hybrid organisations at the university-industry-government nexus and Section 2.4 concerns the creation of partnerships at the nexus to deliver low carbon and/or sustainable innovation. Finally, Section 2.5 considers issues related to the development of the partnerships themselves.

As a compilation of the literature this Chapter provides a synthesis of the context for the development of the theoretical research findings presented in Chapter 4 - 7. A summary of the literature reviewed is provided in Table 2.1.

#### 2.1 Background

Knowledge is a key factor in driving economic development and growth [1] and the exploitation of knowledge has led to five broad technological transformations that have driven economic growth and prosperity since the industrial revolution [15].

The increased importance of knowledge for economic success has led to increased interest into how to commercialise knowledge to deliver the next wave of economic growth [7]. In turn this has led to increased academic and policy interest into how universities (as knowledge creators) and industry (as knowledge users) and government (as guardians of the public purse and public policy) can effectively work in partnership to develop the next wave of economic growth. For universities, this increased policy attention and opportunity has led to the development of increasingly entrepreneurial institutions where entrepreneurialism penetrates each of their three missions [8]:

- Mission 1: teach;
- Mission 2: research; and
- Mission 3: engage (with society)

Section	Literature reviewed	Research question	Thesis narrative
2.2	Partnerships for innovation – models of partnership at the UIG nexus	Which partnership models at the UIG nexus have been successful to deliver innovation?	<ul> <li>Increasingly commercial exploitation of knowledge depends upon a partnership between two or more entities</li> <li>The focus is the delivery of commercial success and economic growth</li> </ul>
2.3	Partnerships for innovation – the creation of hybrid organisations	Which factors promote or limit partnership development for a low carbon future?	<ul> <li>Partnership structures are evolving from a top-down/ bottom-up approach</li> <li>Top-down approaches are an organisational response to the exploitation of knowledge</li> <li>Bottom-up structures are more likely to address economic, social and environmental issues</li> </ul>
2.4	Partnerships for innovation delivering economic, social and environmental outputs	Which partnership models at the UIG nexus have been successful to deliver low carbon and/or sustainable innovation?	<ul> <li>Partnerships that give voice to public interest are more likely to drive local low carbon outcomes</li> <li>However, there is little evidence of these partnerships delivering across the sustainable development goals</li> </ul>
2.5	Developing expertise in creating partnerships for innovation	How can Australian universities become more effective at partnering with Government and industry to drive a low carbon future?	<ul> <li>Australian universities are weak at collaborating for innovation when compared to their OECD counterparts</li> <li>Universities need to partner to develop knowledge and experience</li> </ul>

#### Table 2.1Synthesis of literature reviewed

The deepening of the knowledge economy has also happened at a time when knowledge is becoming increasingly available for use by a wide range of stakeholders [2, 16]. This development, discussed in depth in Appendix A, has also created academic, policy and industry interest leading to the development of theories around open innovation [17] and user centred innovation [18]. The increasingly ubiquitous access to knowledge has also led to changes in the modes of knowledge production (see Table 2.2 developed from [19-21]) where knowledge production has moved from being discipline based to transdisciplinary, theoretical to increasingly applied and from being organisation based to becoming increasingly distributed (initially to partners but ultimately to society).

Knowledge production	Explanation
Mode 1	Discipline based; fundamental
Mode 2	Transdisciplinary; applied and distributed between partners
Mode 3	Transdisciplinary; applied; distributed and democratic

Table 2.2	Changes in	the modes	of knowle	dge	production
			•••••••••••	-0-	

A full discussion of the theoretical approach to innovation in the academic literature can be found in Appendix A.

#### 2.2 Theory of partnership development at the university-industrygovernment nexus

The development of innovation as a key driver of economic prosperity and improving standards of living led to the concept of national innovation systems [20, 22]. The national innovation systems approach focused on the innovation ecosystems in different countries to understand why some countries were more innovative than others [23]. There are several lessons from the national innovation systems approach but key amongst them relates to tangibility and reproducibility of resources. Tangible resources (for example consumption of energy and materials) are reduced in the economic process whereas non-tangible resources (for example knowledge) can be increased through the process [24].

A weakness of the national innovation systems approach is that it considered innovation as a linear process. This might have been true when knowledge production was firmly in Mode 1, when Bush presented his work to President Roosevelt in 1945 [25]. Increasingly a linear system fails to account for the complexity of the innovation process, new modes of knowledge production, shortening timelines between knowledge creation and innovation [26] and the network of relationships to commercialise knowledge [27, 28].

However, the importance of the national innovation systems approach is twofold. Firstly, it highlighted the importance of social capital (for example to reduce transaction costs and build trust [29]) at the heart of innovation. The second is the construction of an innovation system, based on a social construct [30], that evolves over time [28].

These two dynamics are the key to understanding the development of partnerships for innovation and distinguish it from previous models of innovation. Through this lens the Triple [27], Quadruple [21, 31] and Quintuple [32] Helices are an extension of, or evolution from, the national systems approach as they retain a social construct at their core but take on an evolutionary systems approach.

#### 2.2.1 Triple Helix

The Triple Helix model [27] is a model of interactions that takes place between knowledge producers (universities), creators of economic wealth (industry) and promotors of the public interest/regulators (government). Each partner is represented by a helix, and each helix represents a sub-system connecting or interacting with the other helices in a spiral at a variety of scales (local, regional, national and super-national scale) [33]. There are several variants of the Triple Helix to represent their evolution from different economic conditions. These are briefly presented in Table 2.3. A more detailed discussion can be found in Appendix A.

A benefit of the Triple Helix model is its flexibility – where partnerships can be created and managed in a manner that is appropriate for the local circumstance [13, 34]. But the ability to adopt a different version of the Triple Helix in every context (see Table 2.4) is also a weakness as universality means that it is more of a framework than a practical expression of a process to follow to successfully commercialise knowledge [3, 35]. This issue has implications for the research presented in this thesis which is discussed further in Chapters 4 - 7.

9

Triple Helix structure	Diagram	Description
Triple Helix 1: Statist	Government University Industry	Nation state encompasses academia and industry and directs relations between them
Triple Helix 2: Laissez faire	Government University Industry	Separate institutional spheres with strong boundaries between them
Triple Helix 3: Overlapping or balanced	Government University Industry	Generation of knowledge through the overlapping of institutional spheres
Triple Helix 3: Social structure	Government University Industry New organisations arise at intersections	People representing their own views and interests leads to creation of new structures at intersections
Triple Helix 4: Delayed Government	Overlapping merging back to statist	Led by university and industry, but government involvement only happens later, but then takes over

Table 2.3 Triple helix model structures

#### 2.2.2 Quadruple Helix

In recognition of the increasing universal access to knowledge and the progression from Mode 2 to Mode 3 knowledge production the addition of a fourth helix (to represent the codification of knowledge within society) to create the Quadruple Helix has been proposed [21]. The fourth helix represents a development in both the model itself as well as the structure of the model as the fourth helix does not represent an organisation but rather the interests of society resulting from the democratisation of knowledge [21, 36].

The fourth helix represents the importance of engaging with the public to successfully achieve goals and objectives [21]. It is unclear who owns the goals, but the fourth helix is responding to the democratisation of knowledge and represents the interests of the public in a manner different to the government's ability to represent the public interest.

The introduction of a fourth helix also reflects a move in Europe towards open science and Responsible Research and Innovation [37] (see Appendix A) whereby research organisations, and businesses engage in conversations with societal stakeholders and practitioners to ensure that their proposed research direction is broadly acceptable to society. The fourth helix is argued to represent the social structure that underpins the Triple Helix [31].

#### 2.2.3 Quintuple Helix

The Quintuple Helix adds a further helix to the Quadruple Helix. The Quintuple Helix was also proposed by Carayannis [32] arguing that it offers an analytical framework where knowledge and innovation is connected with the environment or natural environments [32]. The fifth helix is represented as a further subsystem akin to, and given equal status of, the other sub-systems: namely university, industry, government and public interest. But the equal weighting given to the environment is also the challenge of the model, as to create innovation that is sustainable requires an understanding of planetary boundaries and an understanding that what is possible locally impacts what is acceptable globally.

As there is no planetary boundary within the Quintuple Helix the issue of finite resources does not interact with decision making and does not offer an explanation as to why a Quintuple Helix model will result in innovation that is any more sustainable than the Triple or Quadruple Helix.

11

Table 2.4	Triple Helix outcome d	levelopment

Author	Outcome sought	
Leydesdorff (1996) [38]	High technology development; global knowledge economy	
Etzkowitz and Leydesdorff (1995) [27]	Wealth creation	
Etzkowitz and Leydesdorff (2000) [20]	New technologies (for example biotechnology and ICT))	
Etzkowitz et al (2000) [28]	Improving regional or national economic performance	
Etzkowitz (2003) [39]	Transition to a knowledge-based economy	
Etzkowitz (2011) [40]	Technological innovation and economic development	
Etzkowitz (2013) [8]	Science based innovation	
Etzkowitz and Zhou (2017) [41]	Developing technologies and applied research	
Cai and Etzkowitz (2020) [42]	Regional economic growth and entrepreneurship; Research	

#### 2.3 The creation of hybrid organisations

Models based on the interaction of various helices give rise to the potential for the creation of hybrid organisations where the helices cross or bridge the boundary between the organisations [27]. The rate of production and the complexity of these organisations is increasing. This issue is discussed further in Appendix A, Publications 1 and 2 and Chapter 6.

The deepening of the knowledge-based economy is affecting how all partners interact - university, industry and government - but also how consumers and citizens interact with other partners in the innovation ecosystem [43]. This has created new power dynamics and new opportunities for individualised partnerships [44] at different scales [45, 46].

In the literature these hybrid organisations, or intermediaries for innovation, are presented as the result of a top-down, or bottom-up, initiative within the partnership (see Table 2.5). Table 2.5 presents a range of organisational responses taken to commercialise knowledge. Top-down responses are presented as being the result of management decision making whereas the initiative for bottom-up intermediaries is less specific [41] but comes from within the organisation (Publication 1).

Publication 1 (see Section 4.1) presented an addition to the top-down/bottom-up hypothesis. This proposal was for an evolution in the development of the intermediaries for innovation that was based on both top-down/bottom-up initiatives but also on the ability to create different sorts of organisations due to exogenous factors. In doing so it suggests a link between the creation of partnerships for innovation at the university-industry-government nexus and the changes in the mode of knowledge production as well as the deepening of the globalisation of the economy. This is discussed in greater detail in Section 4.1 and developed further in Chapter 7.





#### 2.4 Partnerships delivering economic, social and environmental outcomes

There is a thin vein of research concerning the development of partnerships for innovation to deliver low carbon or sustainable innovation. Publication 1 showed that those partnerships that do so are living labs and, in this synthesis of the literature review, represent a bottom-up typology. The literature does not offer insights into why top-down structures do not cater for sustainable innovation but rather focus on economic development with the consequent benefits afforded to economic agents. This issue is discussed in greater detail in Section 4.1, Appendix A and Appendix B.1 – Publication 1.

#### 2.5 Developing expertise in partnership development

The importance of commercialising knowledge as a route to economic growth is widely recognised in the literature [12] and by international development organisations

[6] and came to the fore particularly after the 2008 financial crisis due to the potential to create new jobs [47] and to harness the next wave of economic prosperity [5].

Australia's National Innovation and Science Agenda (NISA) was presented in 2015 and created an agenda to exploit the economic opportunities presented by innovation to drive economic growth and maintain standards of living [5]. The Agenda presented four pillars to support innovation policy – one of those pillars was collaboration.

In 2014, the rate of collaboration between Australian universities and industry was the lowest in the OECD [5, 48] and remained weak in 2021 [49]. Australian industry is also weak at collaborating with other businesses or suppliers for innovation with 25% of those who do collaborate doing so with another organisation within their ownership structure [50]. Business collaboration with publicly funded research is limited, standing at 2% of patent applications [50]. However, collaboration to deliver innovation added \$10.8bn to the Australian economy in 2018 and generated 30,000 jobs, although this is 30% lower than achieved in USA and Israel [51].

The relative weakness in collaboration stands in contrast to public university research output of which 90% is at or above world standard [51] ranking Australia at 23 in the world innovation index [52].

Australia performs well in knowledge creation but poorly in translating that knowledge into new products and services [49]. This weakness has been recognised by Innovation and Science Australia [53] who in 2017, made improving the collaboration between universities and industry as one of their five imperatives [54].

14

# Chapter 3 Methodology

This Chapter outlines the methodology utilised for this thesis as well as the data collection and analysis methods.

#### 3.1 Research methodology

This research adopted a mixed methods approach combining a systematic literature review, narrative literature review, critical analysis, thematic analysis, inductive reasoning, case study and interviews. The method utilised for each publication is outlined in Table 3.1 and discussed in Sections 3.2.1 to 3.2.4.

Publication number	Publication	Method
1	A systematic literature review of partnership development at the university-industry- government nexus	Systematic Literature Review including PRISMA methodology [55] Critical assessment of outputs through articulated, attributed and emergent data [56]
2	Business models for sustainability in living labs	Case study assessment of narrative business cases Inductive reasoning
3	If living labs are the answer – what's the question?	Narrative review and qualitative inductive reasoning developing initial theoretical framework for research
5	A study of Australian universities' collective response to climate science.	Qualitative and quantitative review of universities' response to climate science based on detailed deconstruction of annual reports, mission statements and corporate strategies for all Australian universities Document analysis consisting of detailed textual and balance sheet analysis
5	Happy homes – the relationship between homes and mental wellbeing	Narrative review Inductive reasoning Qualitative business case development
6	Unpublished: Interviews for campus development	Case study of the development of student accommodation.
7	Unpublished literature review	Narrative literature review

Table 3.1Research methods reported in the publications

Mixed methods research has been referred to as the third research paradigm after quantitative and qualitative research [57] and this research is based on a mixed methods approach. The research design is such that it deductively analyses the theory and then applies it to case studies and in turn refines the theory in the light of the findings. It follows the mixed methods tradition with a combination of data and data sources that are interdependent to develop an outcome that is integrative [58]. The sources of data used in the publications, and the heretofore unpublished data presented in this thesis meet Bazeley's [58] test of being interdependent, mutually informing and used for a common purpose.

The data used is derived from many sources where qualitative data weighs more heavily in the design and execution of the research. Qualitative research refers to a wide variety of approaches to and study of natural social life and the outputs include new insights and understandings of social complexity and the evaluation of policies and programs. Hence qualitative research is grounded in the non-fictional realm of social reality [59]. But quantitative data has also been used to test assumptions and findings.

There is a definitional debate in the literature about what constitutes a mixed methods approach. Many academics argue that mixed means a mixture of qualitative and quantitative data (for example see [57]). This thesis meets Creamer's [60] test of using qualitative and quantitative data and integrating them in a meaningful way.

However, there is a parallel approach that posits the important point is how the data is used and integrated and that a variety of data sources is more important than a simple split between quantitative and qualitative data (for example see [58]).

This thesis draws inspiration from both approaches to the methodology and it draws strength from the mixture of qualitative and quantitative data sources which are integrated deriving gains in knowledge when different types of data are actively and systematically engaged.

#### 3.2 Research design

This research was designed to remove the sequentiality of data collection but to keep a clear focus on the research outcome. The research design was integrative and reflexive with a focus on the research outcome as well as research process. This has led to a degree of real author angst but also unexpected creativity, energy and clarity which

16

in turn has led to a discursive and analytical assessment of the theoretical underpinnings to the Triple, Quadruple and Quintuple helix, informed by findings at various stages of gestation, which has resulted in a significantly stronger and more fundamental contribution to society's knowledge which is outlined in Chapter 6 and Chapter 7.

This reflexivity has been strengthened by the mixed methods approach¹ to this research since different forms of data, at different stages have impacted in a non-sequential yet focussed and meaningful way to the development of the final research outcomes.

#### 3.2.1 Systematic literature review – Publication 1, Appendix B.1

Systematic literature reviews (SLR) originated in health studies but are now used in the social and engineering sciences. They are also consistent with the mixed methods approach of this thesis as they take qualitative and quantitative findings and integrate them in a meaningful manner [61]. The SLR offers a methodology to interrogate a body of data in a manner that is repeatable, and evidence based [62] to produce valid and reliable results [63].

A systematic literature review "is a review of a clearly formulated question that uses systematic and explicit methods to identify, select, and critically appraise relevant research, and to collect and analyse data from the studies that are included in the review" [55]. In 2005 an international meeting endorsed the use of Preferred Reporting Items for Systematic reviews and Meta-Analyses (PRISMA) for use in SLRs [64]. PRISMA focuses author attention on the way to ensure transparent and complete reporting of data in SLRs [64]. PRISMA therefore helps to manage bias in the selection of articles for inclusion and the reporting of data in SLRs [55]. PRISMA is deemed applicable to all SLRs and not just ones carried out in health [55]. The SLR undertaken for this research complies with the PRISMA statement [64].

#### 3.2.1.1 Discussion of the analysis of latent qualitative data

The SLR is a way of making sense of a large body of data in a manner that manages author bias with regard to the selection of articles [62]. However, this still leaves the interpretation of the results open to author bias. To further manage author bias in the interpretation of the search results a method developed by Massey [56], initially proposed

¹ See Bazeley (2015) or Creamer (2018) for a fuller discussion of the benefits of integrative data in mixed methods research
for use in interpreting the results from focus groups, was adopted. Using this process, a more transparent and repeatable process for the interpretation of results was developed [56].

In concert with Massey [56] the three levels of data should be viewed as a hierarchy, where the potential for bias is inverted according to its position in the hierarchy with articulated data being subject to least bias (and therefore most repeatable) whereas attributional and emergent data are the least transparent due to the necessary engagement with the SLR authors' views and experiences. It is important to note that bias is not eliminated, but it is minimised and managed to increase the replicability of the work. It is also important to note that the quality of findings increase from articulated (low) to attributed (medium) to emergent (high).

The use of Massey's [56] methodology in an SLR has not been done before. There were significant, and somewhat unexpected benefits that came out of using the process (see below). It should be noted however that it did create a significant extra burden of work, but the results were sufficiently strong to prove its usefulness. The key benefits were:

- The analysis of the attributional data (i.e., the theoretical underpinning for each article) provided insights into the epistemological foundation for the work. This foundation was exposed due to the methodology followed and is a clear benefit in adoption. In this SLR it was demonstrated that the partnerships for innovation were looked at through an economic lens, which contributed to a key finding. This benefit was unexpected but has proved to add real insights into the study's conclusions.
- The section on emergent data provided a useful discussion of findings and emerging hypotheses in a manner that was discreet and did not interfere with the presentation of the observable data. This benefit was unexpected but has proved to be very useful.

The use of the methodology adapted for SLRs [12], whilst involved and timeconsuming delivers tangible benefits to the study and might be adopted more widely.

#### 3.2.2 Case study – Publication 2, Appendix B.2

This case study involved an in-depth analysis of business cases for four active sustainability living labs (known as SusLabs) which were part of the SusLab Northwest Europe network (SusLabNWE). Case study research methodology is broadly based with few limitations in terms of what is being studied [65]. The case study in this thesis is defined as a cross case study [66] or multiple-case study [67].

Levels of data	Massey definition [56]	Revised definition	Source of data	Potential for bias	Quality of findings
Articulated data	Information that is expressed in response to, or specifically addresses the questions posed	Information that is stated directly by the original research authors as the result of their research. this is the original author's statement of results achieved and reported in the article.	Data is articulated by one or more study in the SLR	Lowest	Low
Attributional data	Comments and discussion that relate to <i>a priori</i> hypotheses or theories that the evaluator brings to the study	unchanged	Data is a co- location of ideas from 2 or more articles and is supported by extant models and theories	Medium	Medium
Emergent data	Information that contributes to new insights and hypothesis formulation and is the unanticipated product of individual comments and exchanges amongst group members.	Information that contributes to new insights and hypothesis formulation and is the unanticipated product of connections arising from different articles	Data that is the result of the consideration of articulated and attributional data. It is not necessarily supported by an extant body of work	Highest	High

Table 3.2Modification of Massey's [56] model for thematic analysis of latent data

The case study involved a detailed and intensive study of the business cases for four active sustainability living labs in northwest Europe (see Table 3.3). The business cases reviewed were done so at a single point in time to understand how each SusLab was approaching the engagement of partners and their strategy for longer term viability. The cases selected were sustainability living labs which were primarily focused on reducing energy consumption and increasing energy efficiency. In the extant literature the focus on partnerships to deliver sustainable innovation is scarce (Publication 1, [12, 14]) hence these four viable sustainability living labs provided an opportunity to look in depth at business case development to deliver economic, social and environmental outcomes, as well as to understand what motivated the development of each business case and then draw conclusions. It is not clear if the four business cases are a bounded unit or representative of the field [66] – this is difficult to judge in such a confined and niche field.

Table 3.3	Case study	participants

Organisation	Business model
HSB Living Lab, Sweden	Knowledge to business and business to business
SusLab NRW, Germany	Business to Business and Business to consumer
SusLab Living Lab, UK	Demonstration of concept
Concept House Village, Netherlands	Sponsorship

However, when compared to the wider living lab literature these four cases represent "cases that are different on all independent variable except the one of interest to the researcher" and as such they provide a strong basis for generalisation [68]. But what is more apparent is that studying the four cases has led to the development of theories and causation that are congruent to the mixed methods approach of this thesis and the evidence is used in the conclusions drawn in Chapter 6 and Chapter 7.

## 3.2.3 Narrative review and inductive reasoning – Publications 3 and 5, Appendix B.3 and B.5

Publication 3 is explorative in nature. It was written to explore some of the barriers to the adoption of best practice in partnership development. It was designed to explore the performance of particularly Australian universities in partnering with industry and it was also designed to start the process of thinking about how to create innovation at the university-industry-government nexus that supports the delivery of sustainable development

Publication 3 was transdisciplinary by design and inspired by behavioural economics and specifically theories developed to explain how humans think and make decisions [69] to see if these could be applied to how organisations make

20

decisions. Publication 3 illustrates that there is a rich seam of work in this space (in the footsteps of *The ecology of commerce* [70]) but that ultimately, although fascinating and potentially mould breaking, decision making is outside the remit of this thesis and is for future research.

Publication 5 was designed to establish the forms of issues that a university student accommodation construction project could address in partnership with industry and government. It was designed as a narrative literature review to explore the issues of mental health and accommodation to identify what forms of design interventions are available to either stabilise or improve mental health.

#### 3.2.4 Quantitative data collection and analysis – Publication 4, Appendix B.4

The results of semi-structured interviews (see Chapter 5) highlighted the difficulty practitioners were having in managing a complex process with interrelated but competing themes. To see if this was an issue specific to the Greater Curtin project or a more generic issue, research was undertaken on Australian universities collective response to climate science.

There are 43 Universities in Australia. All Australian university websites were accessed between 18 – 20 November 2016 to download their 2015 annual report and financial statements and their forward-looking corporate strategies (dates ranging up to 2025). Also, all government/university performance agreements (known as Mission Based Compacts) were accessed from the Department of Education and Training website. A limit of 30 minutes was allocated to each university to find and download the data. Where the data was unavailable online (either not found or not published) it was requested by email.

All available annual reports and strategies were read and the public commitments in those reports (for example published statements, targets and performance) to partnership and adopting research were recorded and tabulated. Research on climate change was taken as the example for adoption of research due to the agreed status of the science [71], and the applicability of it to campus development (and therefore to capital investment in campus projects).

21

The documents were also searched through keywords (carbon, sustainable, greenhouse, climate change and variants) to understand the importance of sustainability and climate change for each university. They were also searched to understand the importance of collaboration within the university, to the university (words searched were cross-, inter-, multi-, trans-disciplinary and variants) as well as externally with industry, society and government.

Each University's capital spend, committed for 2016², was recorded to understand the potential of the sector to use their own capital spend to adopt and commercialise their own research (for example a university with no forecast capital spend would find it more difficult to integrate teaching and research into campus development).

Due to the incomplete nature of performance data within the university publications, publicly available performance data was accessed to develop an indicative understanding of the rate of change of emissions from Australian universities and thereby to understand their success in translating research into impact.

#### 3.2.5 Case study: semi structured interviews - Chapter 5

Chapter 5 details the results of a series of semi-structured interviews, undertaken in 2016 with practitioners. The interviews explored the practitioners' views on the process of integrating education and research outcomes throughout the design and construction process for a university student accommodation project. Interviews were semi-structured consisting of 10 meta-questions (see Table 3.4) with the opportunity for the interviewee and interviewer to then discuss issues that arose or needed further elucidation. Interviewees were chosen to represent a cross section of senior academics, business managers and consultants (see Figure 3.1) to test hypotheses developed for Publications 4 and 5. Interviews were conducted in Autumn 2016 and lasted for approximately one hour. All interviews were recorded using *Livescribe* and then manually transcribed. To protect individual's privacy all participants have been anonymised. The results of the interviews were then coded and analysed. The themes and associated nodes identified in the thematic analysis are set out in Table 3.5.

² All annual reports detail money spent during the past year. They are also required to report on capital expenditure for the forthcoming 12 months and 5 years. Hence 2015 annual report contain data for 2016.

	Question
1	How has innovation, leadership and teaching and research excellence been planned into Stage One?
2	Can you identify any specific examples of Innovation, Leadership, Teaching, Research that have been included in the EOI process?
3	How much has Curtin spent so far developing the Stage One proposal?
4	How much of this is for professional services (by which I mean consultancy advice)?
5	How many of Curtin's academics have been consulted on the development of the Stage One proposals?
6	Can you identify any specific examples of Innovation, Leadership, Teaching, Research that have been achieved to date?
7	How much has Curtin budgeted to spend by early 2017?
8	Can you identify any specific examples of Innovation, Leadership, Teaching, Research that are intended to be delivered by early 2017 and 2019?
9	Can you identify the teaching and research outcomes associated with that expenditure?
10	How does the activity and the planned activity demonstrate leadership by Curtin?

Table 3.4Meta questions influencing semi structured interviews

Table 3.5 Themes and associated nodes id	dentified in the thematic analysis
------------------------------------------	------------------------------------

Theme	Node
Loadorship	Using exemplars
	Developing teams
	Integrating research
Developing opportunity	Integrating teaching
	Outreach
Pick management	Learning from others
Kisk management	Learning by doing
Developing partnerships	Business to business
	Academic to business
	Knowledge
Communication	Participation
	Ease of engagement

Figure 3.1 Interviewee's involvement in student accommodation project



The purpose of the interviews was to test the findings of Publication 5 and to understand how deeply embedded research, teaching, leadership and innovation were into the accommodation project. The interviews also went deeper to understand the extent to which the project was being used to create partnerships for innovation and collaboration, either in practice or in future stages of development.

These data have not been previously published but they are the third part of a triptych of publications (Publications 4, 5 and Chapter 5) exploring the factors that promote or limit low carbon innovation at the university-industry-government nexus (see Sections 4.4. and 4.5 for a detailed discussion).

### Chapter 4 Publication results and discussion

This Chapter provides a summary of each publication and thereby mapping a progression in thinking and the development of the research evidence for the thesis. Table 4.1 summarises how the published and unpublished work address the research questions.

Sub-questions	Objectives	Associated publications
Which partnership models at the UIG nexus have been successful in delivering innovation?	To identify the development of partnership models to create best practice partnerships To identify partnership structures that are being used to deliver a low carbon future.	<b>Publication 1</b> : A systematic literature review of partnership development at the university- industry-government nexus Published, peer reviewed article
Which partnership models at the UIG nexus have been successful in delivering low carbon and/or sustainable innovation?	To identify best practice in business case development in partnerships at the UIG nexus	Publication 1: A systematicliterature review of partnershipdevelopment at the university-industry-government nexusPublication 2: Business modelsfor sustainability in living labsPublished, peer reviewed bookchapter
What factors promote or limit partnership development for a low carbon future?	To identify issues that prevent effective partnership creation	<b>Publication 3</b> : If living labs are the answer what is the problem? Published, peer reviewed article
How can Australian universities become more effective at partnering with Government and industry to drive a low carbon future?	To identify Australian universities' performance in partnering with industry and Government; To identify the opportunities for universities to work with business, or Government in partnership; To identify universities' performance in adopting low carbon policies	Publication 4: A study of Australian universities' collective response to climate science Published, peer reviewed article Publication 5: Happy homes – the relationship between homes and mental wellbeing. Published, peer reviewed article
How can Australian universities become more effective at partnering with government and industry to drive a low carbon future?	To identify the process to develop a construction project to deliver student accommodation	<b>Unpublished</b> : Case study of the development of student accommodation. See Chapter 5

Table 4.1Summary of how publications address the research questions

The Chapter is arranged to provide a summary of the paper, the findings associated with each paper; a discussion of the findings and then concluding remarks which place the papers in the wider context of this exegesis. The publications have been reproduced in full in Appendix B.1 to B.5.

# 4.1 A systematic literature review of partnership development at the university-industry-government nexus

This article, Publication 1, was published in a peer reviewed journal (Appendix B.1). It directly addresses sub-question 1 and includes a contribution to sub-question 2 (see Table 4.1).

- <u>Sub question-1</u>: Which partnership models at the UIG nexus have been successful in delivering innovation?
- <u>Sub question-2</u>: Which partnership models at the UIG nexus have been successful in delivering low carbon and/or sustainable innovation?

#### 4.1.1 Summary of findings

Publication 1 has made the following contribution to the theory of partnership development at the university-industry-government nexus.

- Builds on top-down/bottom-up analysis discussed in the literature and demonstrates that the evolution of partnerships for innovation are driven by factors within the partnership but also by factors external to the partnership;
- Describes 3 stages in the evolution of partnerships for innovation, which can be concurrent: namely the development of on-campus structures; campus adjacent structures and living labs;
- Finds only living labs (and derivatives) tackle economic and social issues; a smaller subset tackle economic, social and environmental issues;
- Due to the adaption of Massey's methodology [56] for analysis of latent data (see Section 3.2.1.1), Publication 1 demonstrates that the majority of research in the literature is grounded in economic theories; and
- Development of a hypothesis linking evolution of partnerships for innovation, globalisation of economy, society and environment and changes in modes of knowledge production.

#### 4.1.2 Partnership development at the university-industry-government nexus

Publication 1 reveals that the innovation intermediaries created by university, industry and government are created through an internal dynamic, and these are either management led (top-down) or led by an entrepreneurial individual or group (bottom-up) [34, 41]. These types of intermediaries for innovation, as identified in the literature, are shown in Figure 4.1 (Publication 1).



Figure 4.1 Innovation intermediaries at the university-industry-government nexus

It is clear from the literature that the deepening knowledge-based economy is affecting not only how universities, industry and government interact, but also how consumers and citizens [43] interact with other partners in the innovation ecosystem. The pace of change affects all sectors of society (university, industry, government as well as citizens), and this dynamic relationship is rapidly changing, which is leading to new forms of intermediaries that are highly individualized [44]. This, in turn, leads to the opportunity for innovation at different scales and under differing dynamics and delivering different outcomes [46, 72].

Various forms of intermediaries are being created because of decisions made within the partnership, but also being facilitated by external opportunities and stimuli (exogenous factors). It is this overlay of externally changing dynamics that has led to new forms of intermediaries for innovation emerging [73] and that are being led by, or include, different actors [74] or different power dynamics and approaches [72, 75].

Additionally, new models for innovation are being adopted by different sectors—including the public sector [76] or cities [77]. It is this change in dynamics, structure and power relationships that is leading to the nascent creation of innovation

that is seeking to deliver economic, social and environmental enhancements at the same time (Publication 2 [78]).

These external stimuli (the deepening of the knowledge economy; increasing globalisation of the access to the internet) has led to the description of an evolution in partnerships for innovation at the university-industry-government nexus. This is detailed in full in Publication 1 (see Appendix B.1). The research shows the temporal evolution of three types of structures to promote partnership development and the commercialisation of knowledge:

- On campus structures. These are typically created as a management response to encourage greater partnership between the university and industry or government. The intervention ranges from access to university library facilities [79], to the creation of posts such as knowledge transfer officers [80] or academic liaison officers [81], or the creation of space for incubators [82]
- Campus adjacent structures. This involves the creation of science and technology parks (STPs) (and derivatives) adjacent, or close to a university, or cluster of universities. This process started in North America [7] and was adopted in the UK and Europe [83-85]. There is much debate about the efficacy of STPs [86] but Publication 1 does reveal the STP inhabitants place value on informal links with universities [79, 87, 88].
- Development of living labs. Living labs are partnership structures focussed on user engagement and open innovation [89]. The impetus for the development and increasing adoption of living labs was the ability for a range of stakeholders to become involved in the process of innovation [90].

Publication 1 also revealed that much of the research undertaken into partnerships for innovation is, epistemologically, through the following three broad lenses:

- Economic theories: most papers (over 100 of the 132 reviewed in Publication 1) looked at the issue of partnerships for innovation with an economic theoretical underpinning analysed through theories of innovation, economic geography, planning and transitions.
- Social theories: the main focus in the social theories was how individuals interact with or contribute to innovation.

 Theories of learning: this was a shallow vein of work but there were elements looking to see how innovation could be used to deliver learning to students, and how organisations can learn by doing.

Publication 1 also revealed, in concert with other studies (for example see [14]) that there was limited research looking into the delivery of innovation that seeks to balance and integrate economic, social and environmental interests at the same time. Further it revealed that the only partnership structures, as identified in Publication 1, to do so were living labs and their derivatives. This is discussed further in Section 6.1.1.

#### 4.1.3 Discussion

Publication 1 revealed several interesting insights. The first was the finding that factors exogenous to the partnership were driving partnership development whilst the literature has viewed the development of new partnership structures as the result of endogenous factors (that is due to decisions taken within the partnership). This finding is new to the literature and led to the development of potentially sequential creation of partnership structures at the university-industry-government nexus. The temporal nature of these partnerships is discussed in detail in Section 6.1.1. and developed in Chapter 7.

The second finding is that, in Publication 1, the only partnership structure to seek to deliver sustainable outcomes is the living lab (and its derivatives). The literature does not reveal why the living lab structure is the preferred structure to deal with sustainable development. This too is discussed further in Section 6.1.1. There are two hypotheses: one is that it is because the research undertaken is driven through economic theories with limited articles being underpinned with a sustainable perspective; the other is that the living lab approach is based on user engagement and open innovation which allows greater visibility to, and expression of, a community of interests in low carbon transition and sustainable development.

#### 4.1.4 Progression in research findings

The use of the adapted Massey methodology [56] to augment the SLR (PRISMA) [55] methodology was powerful (see Section 3.2.1.1). Analysis of attributed data led to the finding that the majority of the research presented for analysis through the SLR methodology was grounded in economic theories.

Discussion of the emergent data led to the development of a hypothesis that links the evolution of partnerships for innovation at the university-industry-government nexus to the globalisation of the economy, society and environment as well as the changing modes of knowledge production. This is set out in Table 4.2 developed further in Section 6.2 and progressed to the presentation of a metastructure in Chapter 7.

The limited focus in the literature on partnerships for low carbon innovation and business cases to develop low carbon partnerships is also addressed in Publication 2.

Table 4.2Hypothesis of current relationship between partnerships, globalisation<br/>and modes of knowledge production

Evolution of partnerships for innovation	Potential partnership response to trends in globalisation		Mode of knowledge production	
On campus				Modes 1 and 2
Campus adjacent	Economy	Society		Modes 2 and 3
Living labs		Society	Environment	Mode 3

#### 4.2 Business models for sustainability in living labs

Publication 2 is a book chapter with seven co-authors (see Appendix B.2). It directly addresses sub-question 2 and includes a contribution to sub-question 1.

- <u>Sub-question 2</u>: Which partnership models at the UIG nexus have been successful in delivering low carbon and/or sustainable innovation?
- <u>Sub-question 1</u>: Which partnership models at the UIG nexus have been successful in delivering innovation?

#### 4.2.1 Summary of findings

Publication 1 (see Section 4.1) showed that the innovation partnership structure to deliver economic, social and environmental outcomes at the same time was the living lab. Therefore, the objective of this publication was to look at the development of business cases for four sustainability living labs (SusLabs) to understand the drivers of their development. The four SusLabs were focussed on the development of energy efficient products, services or systems but all had different businesses cases. Publication 2 sought to understand the reasons for this difference.

Publication 2 has made the following contribution to the theory of partnership development at the university-industry-government nexus:

- Each business case was different due to the impact of three factors:
  - The principles or purposes underpinning the SusLab's creation;
  - The response of each entity to pressures or opportunities they faced during operation; and
  - The response to internal or external opportunities which meant the business cases evolved over time
- The article shows how the Triple Helix model can be implemented locally and can be used in partnerships for innovation that seek to deliver innovation consistent with sustainable development; it shows how path dependence theory helps to understand the decision-making process to develop business cases; and it shows the value members of the European Interreg project (SusLabNWE) placed in being part of a wider consortium to assist with learning from experience and from doing.

#### 4.2.2 Business case development for low carbon innovation

The book chapter had three significant findings. Firstly, business cases were different because although each SusLab was focussed on energy efficiency they each had a different founding principle or purpose. The purpose was either development of products, services or systems but also there was a focus on knowledge creation through research and business to business development. Further detail is provided in Table 4.3.

SusLab	Purpose
HSB Living Lab	Attracting companies who will commit to research and development for their own and societal interest
SusLab NRW	Testing products and services in a real home environment
SusLab Living Lab in London	Procured projects to demonstrate potential of collaboration
Concept House Village	Procured projects to demonstrate potential of collaboration. Students involved in design and build

Table 4.3	Summary of individual SusLab purpose
-----------	--------------------------------------

Secondly the gestation of each SusLab was different and arose due to different internal or external pressures and/or opportunities and this meant that each business model evolved differently. A summary is provided in Table 4.4.

SusLab	Business model
HSB Living Lab	Knowledge to business and business to business
SusLab NRW	Business to business and business to consumers
SusLab, London	Demonstration project
Concept House Village	Sponsorship

Table 4.4Summary of each SusLab's business case

Thirdly each business case, like each SusLab, is evolving. Neither each SusLab nor their associated business case is static but changes as opportunities arise. The change can be iterative (for example SusLab NWR identified a new stream of income and was rethinking its business case as a result). The changes can also be more fundamental. For example, Concept House Village was created with a backdrop of public funding, but its new business model is focussed on being independent of public funding and to accelerate the real market application of its prototyping with a view to becoming a regional focal point for sustainable building.

#### 4.2.3 Discussion

The study of the individual cases is interesting and confirms the importance of three extant theories or models. It shows how the Triple Helix model [27] can be implemented to benefit consumers and how the model can be applied to innovation driving sustainable outcomes. It demonstrates that path dependence theory is relevant and does influence decision making, though not by repeating the same decisions, but by using the same prompts to help make decisions (this issue is further considered in Section 5.4). Further it demonstrates the value of co-creation and including end users, for product development.

But the other benefit of the SusLabNWE consortium is unexpected, and that is the value each partner places on being part of the network. These benefits are set out in Table 4.5. This is a finding that is shared with Publication 1 (discussed in Section 4.1.3)

which found that informal linkages with a science and technology park are important to park inhabitants. In Publication 1 such informal connections ranged from social activities to assistance with business processes (like human resources) which whilst not mission critical was important none the less. It appears that all SusLab partners valued learning by doing but also learning from other people's experiences.

Value	Benefit
Communications	Document ongoing work on SusLab website, including findings from HSB Living Lab, and the Living Lab at the Institute of Sustainability Finalise SusLab book and promote methods at scientific and professional forums as well as at network events
Academic	Work with network to share and co-develop new SusLab methodologies Publication of results and methodology in academic journals Continue to develop SusLab sensor tool kit to support new work Continue to develop the SusLab tools such that the toolkit can be easily deployed in the field by sustainable building researchers and practitioners Each SusLab has a measuring and monitoring element and all four can be connected to a single data store and analysis tool to enhance rigour and ensure learnings are shared
Outreach	Maintain network of living labs linked to SusLab website Link SusLab work to regional networks Leverage SusLab network combined with new partners to develop new joint projects Value/weight to partners from increased potential to attract further funding (public and private) for SusLab and product development Enhanced business and academic brand value

Table 4.5	SusLabNWE—informal value to partners in the SusLabs of the Triple Heli	х
		•

#### 4.2.4 Progression in research findings

The findings in this publication are consistent with the findings of Publication 1 (see Section 4.1.1) namely that:

- The Triple Helix model is relevant to a range of situations including sustainable innovation;
- Co-creation and open innovation are central to the SusLab approach;
- Being part of a community of interest is important;
- Learning by doing can be augmented by learning from other's experience;

- The factors influencing decisions are endogenous and exogenous to the partnership; and
- Publication 2 has also shown the importance of path dependence for decision making and the potential to learn from other's experience as well as by doing (this is addressed in Publication 4 see Section 4.4.2 as well as Section 5.4). There is a clear link between the expression of path dependency and the endogenous/exogenous factors affecting partnership creation presented in Publication 1.

A transdisciplinary attempt to tackle the issue of decision making, rather that the influences on the decision is included in Publication 3 (Section 4.3 and Appendix B.3: Publication 3) but as discussed in the Methods (see Section 3.2.3), decision making, rather than factors that impact on decisions, falls outside the remit of this thesis.

#### 4.3 If living labs are the answer what is the question?

This article, Publication 3, was published in a peer reviewed conference proceedings (Appendix B.3). It directly addresses sub-question 3.

• <u>Sub-question 3</u>: What are the factors that promote or limit partnership development for a low carbon future?

This publication sought to understand the performance of Australian universities in partnering for innovation as well as the performance of business in partnering with universities to commercialise research. It sought to develop key themes in the research program.

#### 4.3.1 Summary of findings

Publication 3 has made the following contribution to the theory of partnership development at the university-industry-government nexus:

- Australian universities are weak, when compared to OECD counterparts, at partnering with industry for innovation;
- Industry is weak at partnering with Australian universities, as well as customers and suppliers;
- Universities do not make effective use of their campuses to drive partnership development;

- The rate of external change makes partnership development complex and uncertain;
- There is little evidence of the development of innovation partnerships to drive sustainable innovation at scale; and
- Proposed that SDGs should be used as the framework to drive partnerships at the university-industry-government nexus.

#### 4.3.2 Partnership development for low carbon innovation

Continuing the theme from Publication 2 (see above and Appendix B.2: Publication 2), Publication 3 considered different forms of exogenous change detailed in the foresight literature. As such it continued to investigate the consequences for the economy and society of the further deepening of the knowledge-based economy and the likely impact on future employment prospects (where future jobs will be increasingly skill or knowledge based) and the consequent impact on wider society. It also looked at the globalisation of the environment and the continued vexed issue of the "un-managed commons" [91] and the impact of environmental change for society and the economy.

Publication 3 sought to develop the research finding in Publication 2 (see Section 0) about the importance of path dependency to decision making and to look at how organisations institute change. The paper, ultimately unsuccessfully, sought to link human judgement and decision making [69] to an organisational setting, an issue further discussed in Section 8.3.

This publication showed that Australian universities were weaker than their OECD counterparts at partnering with industry to collaborate for innovation [5, 48]. It also revealed that even though Australian industry funds public research in universities at a level that is higher than the OECD average, collectively Australian industry is also generally weak at collaborating with universities, customers and suppliers for innovation [48].

Australian universities and Australian business have a complex relationship in terms of how they collaborate with each other for research and innovation, and this is set out in Table 4.6 which is reproduced from Publication 3.

	Industry	Universities	Customers/ students	Suppliers	Use foresight	Research user?
Industry	Collaboration between industry for innovation is weak and localised [92]	Industry fund universities to undertake research at level greater than the OECD and EU28 average.[48]	Industry is weak at collaborating for innovation with its customers [93]	Industry is weak at collaborating for innovation with its suppliers [93]	Industry weak at using strategic foresight [94]	User, and active funder of research [48]
University	Collaboration for innovation is amongst the worst in the OECD [48]	Universities are good at collaborating with universities [95]	Investment by universities and students/ alumni is the focus of increased attention (pers comm)	No evidence of universities collaborating with suppliers for innovation	Unclear as to whether universities use (rather than just research) foresight (although they do research scenarios)	Unclear if universities use the research they produce (see section 4.3)

 Table 4.6
 Collaboration between Australian universities and business

#### 4.3.3 Discussion

This paper contributed further to the development of thinking regarding factors external to the partnership and their impact on partnership development. The changing economic, social and environmental orbit is impacting universities, industry and government individually and collectively and this effect on collaboration needs to be further considered.

This complexity (as also detailed in Publication 1) of evolving environment means that the development of partnerships between universities, industry and government is fraught with complexity as the rate of change in the broader environment within which they operate is increasing and becoming increasingly complex (Publication 1). It also suggests that partnerships will struggle to adapt due to the impact of making decisions based on previous similar decisions (this issue of path dependency is addressed in Section 5.4). This complex environment also provides a potential hypothesis to be tested into why partnerships for economic, social and environmental innovation happen predominantly in living lab arrangements (see Section 4.1.2), and importantly locally. There is scant evidence from Publication 3 or from Publication 1, or 2 of such partnerships happening at genuine scale.

The lack of evidence of scale raises the important issue of how to create partnerships for innovation that will drive the delivery of sustainable development. Hawken [70] argued the market mechanism can provide the solution through competitive efficiency, and his proposal was through government intervention and Pigouvian tax where the cost structure is such that the totality of external costs is considered within the marketplace. This would then drive the delivery of social and environmentally optimal development.

Publication 3 made a different proposal (which was not available to Hawken [70]), and that is the sustainable development goals should guide the delivery of partnerships for innovation. The SDGs are the shared expression of what the global collective is seeking to deliver for the global citizen and could be used to drive partnerships for innovation at the university-industry-government nexus. This issue is further considered in Chapter 6 and Chapter 7.

Publication 3 also raised the issue of how universities in particular use resources available to them to further teaching and research outcomes (Mission 1 and 2) as well as engagement (Mission 3). This started the process of thinking about the potential of campus re-development to focus on institutional settings to drive all three missions as well as the commercialisation of knowledge in partnership with industry and government.

#### 4.3.4 Progression in research findings

Publication 3 raised the issue of how universities can make greater use of their own infrastructure to drive partnerships for innovation. The focus on the institutional setting is addressed in a triptych of articles. Publication 4 addresses the issue of physical infrastructure; Publication 5 and Chapter 5 address the issue of how to incorporate teaching and research into the heart of such redevelopment.

As discussed above, Publication 3 sought to apply Kahneman's [69] work on human judgement and decision making to an organisational setting. This initial foray, although

37

unsuccessful, was intuitively and logically sound and has potential, but further work is required. It would closely align to a seam of work rooted in evolutionary, environmental and ecological economics with a strong lineage to Hawken [70], Meadows [96], Hardin [91], Boulding [97], Daly [98] and others. Although outside the remit of this thesis it is worthy of further research.

#### 4.4 A study of Australian universities' collective response to climate science

This article, Publication 4, was published in a peer reviewed conference proceedings (Appendix B.4). It directly addresses sub-question 4.

• <u>Sub-question 4</u>: How can Australian universities become more effective at partnering with government and industry to drive a low carbon future?

#### 4.4.1 Summary of findings

This publication is the first of three articles that looks at how Australian universities use their infrastructure (Publication 4) or leverage (Publication 5 and Chapter 5) to develop partnerships for innovation that are based on both the experience of implementing knowledge created and using investment in campus infrastructure to develop new knowledge. Publication 4 focuses primarily on the experience of implementing change based on knowledge; Publication 5 and Chapter 5 on using investment in campus infrastructure to develop teaching and research outcomes (TRO).

Publication 4 has made the following contribution to the theory of partnership development at the university-industry-government nexus:

- Three Australian universities committed themselves to a carbon reduction target in line with climate science;
- For the 10 Australian universities who were required to report emissions data, their emissions rose by 4.63% between 2010/11 and 2014/15;
- Together Australian universities were committed to investing over \$1.5bn in capital infrastructure on their campuses in 2016;
- Eight universities (20%) committed to using their campuses as living labs, or similar, to deliver teaching and research outcomes; and
- Revealed limited evidence of universities willingness to partner with themselves (to use their campus infrastructure to drive teaching and research outcomes).

#### 4.4.2 Developing partnerships based on experience and knowledge

Publication 2 (see Section 4.2) highlighted the importance of partnerships being able to use experience and knowledge to commercialise research. Publication 4 develops this theme by considering the experience of Australian universities at adopting and implementing climate research. The purpose of the publication was to see if universities, having developed the research into climate science and translated that research into policy proposals had experience of implementing the research on their campuses. The theory would suggest that they would then be better able to take this knowledge and experience into a partnership and become a more informed member of that partnership.

This study therefore investigated four issues. The first was how universities collectively had put in place strategies to reduce emissions in line with climate science. The second was to understand the scale of capital investment that was being undertaken, in 2016, into campus re-development. The third was to look to see what relationships universities had developed internally to link academic researchers to university administrators and finally the paper sought to understand the extent to which Australian universities collectively were prepared to use their campus as test beds for innovation.

The research, based on accounting and textual analysis of all Australian universities 2015 annual reports, mission based compacts (these are performance agreements with Government) is summarised in Section 4.4.1. Table 4.7 presents Australian universities commitment to carbon constraint.

	Number of universities	Universities
Universities with a science-based commitment	3	7%
Universities with a carbon reduction commitment and published target	12	29%
Universities who make statements but do not provide evidence as to what success looks like	15	37%
Universities who do not mention carbon or emissions	11	27%
Total	41	100%

Table 4.7 Australian universities commitments to carbon constraint

#### 4.4.3 Discussion

The evidence from this study suggests that universities are not yet fully realising their potential to partner with industry based on experience. The paper reveals the following issues that need to be considered.

Australian universities invested \$1.5bn in 2016 on capital infrastructure on their campuses, of this, eight universities committed to using their campuses as a living lab to develop and deliver teaching and research outcomes (Missions 1 and 2 – see Section 2.2). This means approximately \$1bn of capital spending in 2016 not being explicitly linked to drive teaching and learning outcomes (see Chapter 5). One university committed to using their campus as a living lab (or equivalent) in their forward-looking strategy.

97% of climate scientists agree [71] that climate change is the result of anthropogenic emissions. In 2015, only three universities had made a commitment, in line with climate science, for an absolute reduction in carbon emissions while over 25% of universities had not mentioned carbon or emissions in the documentation analysed.

Finally, the article found limited evidence for university willingness to partner with themselves. This is an important issue as universities are being pressured into developing more effective partnerships with industry. To be an effective partner universities need to have experience and knowledge of the difficulties of implementing research.

#### 4.4.4 Progression of research findings

Publication 4 revealed the following issues that Australian universities, as institutions, need to be cognisant of when considering their capital investment in their campuses:

- An opportunity is being lost if all infrastructure spending is not being used to drive teaching and research outcomes, and/or to showcase how to adopt research;
- Universities have developed climate science, have advocated for society to adopt science-based targets but had not developed the practical experience themselves of implementing that knowledge; and
- If universities are to become experienced, transdisciplinary and knowledgeable partners for innovation they need to become more adept at using opportunities on campus for partnership development as well as delivering teaching and research outcomes. The evidence suggests that this is not a job just for academics but rather for institutions.

40

These issues are further developed in Publication 5 and Chapter 5 culminating with an institutional framework to help universities, industry and government better understand the opportunities to leverage knowledge to create partnerships for innovation.

#### 4.5 Happy Homes – the relationship between homes and mental wellbeing.

This article, Publication 5, was published in a peer reviewed conference proceedings (Appendix B.5). It directly addresses sub-question 4.

• <u>Sub-question 4</u>: How can Australian universities become more effective at partnering with government and industry to drive a low carbon future?

#### 4.5.1 Summary of findings

This is the second of three publications looking to see how universities use their infrastructure (Publication 4) to leverage partnerships for innovation (Publication 5). The objective of this publication was to look at how universities could become more effective partners with industry using mental health and student accommodation as a case study to illustrate the potential to drive research and partnership through campus redevelopment. The findings from Publications 4 and 5 are developed in Chapter 5.

Publication 5 has made the following contribution to the theory of partnership development at the university-industry-government nexus:

- Found little evidence in the self-help guidance of how to maintain or promote mental health in accommodation for the mentally-well;
- Through analysis of the literature developed a series of initiatives to be tested to help maintain mental wellbeing when inside accommodation;
- Proposes that there needs to be a focus on transdisciplinary (to integrate all relevant disciplines) and translational (to test and develop practical interventions that work) research;
- Needs to be a clearer focus on the social pillar of sustainable development through meeting the needs of the building resident by changing building design to improve mental health outcomes;

- Found that there is a focus in the literature on how to use building fabric to deliver desirable policy outcomes (for example climate targets) with building users being considered after policy analysis;
- Study shows the potential benefits of undertaking transdisciplinary and translational research to create an innovative partnership for innovation at the universityindustry-government nexus; and
- Identified three waves of research into mental wellbeing and proposed the impact of the home on mental wellbeing should be the fourth wave of research.

#### 4.5.2 Co-creating partnerships based on knowledge and shared implementation.

The research was conducted through an analysis of secondary sources to identify which measures and initiatives were available to ensure that the design of student accommodation was conducive to promoting positive mental health outcomes. More specifically it was looking at how the design of the living space itself (rather than the environment within which the accommodation sits) could contribute to positive mental outcomes or prevent a decline in student mental health. One purpose of the paper was to provide up to date analysis to Curtin University's student accommodation project coordinator on the state of the research into student accommodation and mental health (this is discussed further in Chapter 5).

The other driver of the research was consideration of how a partnership between the university and industry could be developed to prototype and test the effectiveness of new design proposals by creating a partnership to commercialise knowledge as part of the student accommodation project. As such the research sought to develop an understanding of how an on-campus student accommodation project could be used to develop a partnership at the university-industry-government nexus.

The paper identified three phases in the development of research into mental health and mental wellbeing. These start with the control of contagion in the 1800s to an understanding that mental health is impacted by the environment in the early 1900s to deeper engagement in healthy lifestyles and quality of life in the late 1900s. Pre COVID-19, the paper proposed the impact of the home on mental wellbeing to become the fourth wave of research.

#### 4.5.3 Discussion

There was little in the literature about what mentally healthy people can do within their homes to protect and improve their mental wellbeing (either in new, or existing homes). The focus of the literature about housing and mental health is three-fold:

- Focus on those marginalised in society (be they at the intervention end of the mental health continuum or in need of housing assistance);
- Focus on the issues that will trigger a decline in physical and/or mental health of home dwellers; and
- Focus on descriptive research (i.e., we did this and that happened) rather than translational research (i.e., translating the evidence into wider action)

The work revealed that there is little in the literature to help individuals or organisations design accommodation in a way to promote or maintain mental health outcomes in the accommodation itself for mentally healthy people. It proposed that there needed to be a focus on transdisciplinary and translational research. Transdisciplinary because the study revealed the difficulty faced in capturing all the relevant disciplines; and translational because there is a continued and evidenced need to start developing interventions that can help people in the self-help/social support end of the continuum.

Secondly, there needs to be a focus on the needs of the building user, rather than the building fabric (there was evidence of using building fabric to achieve governmental goals like reducing emissions).

Thirdly, there needs to be a clearer focus on integrating the social pillar of sustainable development into the current focus on economic and environmental pillars.

#### 4.5.4 Progression of research findings

The purpose of this pre-COVID-19 study was to investigate both the state of design for student accommodation as well as to consider the issues that an entrepreneurial university or institution would need to consider in order to forge stronger relationships between it and potential partners. This study shows the potential benefit to a university, industry partners and wider society of the development of enduring partnerships based on the delivery of a shared outcome (improving mental health through better accommodation design) but where each partner will have different desirable outputs (the university outputs would be research; the industry outputs would be a better quality product).

Publication 5 helped to clarify the opportunities open to universities and other institutions to develop partnerships for innovation that cut across all elements of the university (Mission 1: teaching; Mission 2: research and Mission 3: engagement) to partner effectively with industry and government. The findings in Publications 4 and 5 are further developed in Chapter 5 and used to produce an institutional framework to help understand the potential of partnerships for innovation.

The advent of the COVID-19 pandemic, and the impact that isolation has had on mental health across the globe, has underlined the potential value of the proposal.

## Chapter 5 Innovation at the university-industrygovernment nexus. The case of Greater Curtin

This Chapter is a continuation of the enquiry presented in Publications 4 and 5 (see Sections 4.4 and 4.5 and Appendix B.4 and B.5) looking into the use of infrastructure development (Publication 4) and potential opportunities (Publication 5) to develop partnerships for innovation that are based on both the experience of implementing knowledge created and using campus infrastructure to develop new knowledge.

This Chapter continues to address sub-questions 3 and 4.

- <u>Sub-question 3</u>: What factors promote or limit partnership development for a low carbon future?
- <u>Sub-question 4</u>: How can Australian universities become more effective at partnering with Government and industry to drive a low carbon future?

As discussed in section 3.2.5 the data for this Chapter has not been published.

#### 5.1 Background

In 2013 Curtin University presented *Creating the City of Innovation – a vision for Greater Curtin* [99] alongside a campus master plan [100-103]. The vision set out the University's desire to create a "truly globally competitive knowledge centre … and to make a difference to citizens, visitors and the wider community – while ensuring a greater tomorrow for all" [99]. Creating a city of innovation set out a vision to benefit not only the university but also industry for the benefit of Perth and Western Australia [100]. The document was divided into four sections:

- Greater Opportunity which described the economic benefits of the plan;
- Greater Community which discussed the benefits of physical and virtual networks of innovators;
- Greater Learning set out the steps to create a focus on learning and a lifelong framework for the exchange of knowledge; and
- Greater Connections set out how well physically connected the city will be to attract investment but also as a destination.

The concept was to build a city with the sole purpose to innovate, and it would do this through the built environment but also through incidental learning, thinking and new ideas [100]. The documentation foresaw a process that would start in 2013 "even before there are bricks and mortar we will build a city with inspired thought, creativity and knowledge". The projected benefits of Greater Curtin [99-103] are summarised in Table 5.1.

Projected benefits to be realised by 2031			
Greater opportunity	\$4.5bn (including \$317m exports) annually	762,000m ² of floorspace	20,000 jobs created
Greater community	20,000 residents in total	8,000 resident students	73,000 daily visitors
Greater learning	44,000 full time equivalent students	6,700 full time equivalent staff	2,300 staff dedicated to research
Greater connections		54% less CO ₂ than Perth average	76% less water consumed than Perth average

Table 5.1	Projected benefits of Greater Curtin to be realised by	2031
-----------	--------------------------------------------------------	------

#### 5.1.1 Greater Curtin - Master Plan

Alongside *Creating the City of innovation* [99] Curtin University also published its three-part master plan [100-103] which set out spatial strategies to support Curtin's vision and the aspirations set by the WA Government Strategic Plan for Perth [104] to guide the university from being an "isolated suburban campus into a major node of activity and a city-wide community asset" [100].

The master plan was developed by a transdisciplinary project group consisting of Curtin Properties, AECOM, Arup, Block Branding, CBRE, Donaldson and Warn, Pracsys and Syrinx Environmental PL. The master plan set out how the 114-hectare campus would be developed over the succeeding 20 years to create a place where "innovation and opportunity co-exist for the taking" [100]. It described the process of forging strong relationships with industry and government to ensure the goals and targets met the demands and changing face of university education and delivery [100].

"[T]he master plan seeks to establish a flexible framework that the university and its partners can work with to guide the evolution of an urban centre with a strong identity, high levels of vitality, community interaction and opportunities for growth, prosperity and strong partnerships." [100] p10

The Master Plan was conceived around developing a knowledge economy hub in Australia to deliver what Curtin University referred to as four key network strengths [100]. These are set out in Table 5.2.

Network	Strength
Education and innovation network	Designed to meet the future needs of students by creating and delivering courses that closely align with industry and community expectations To become the centre of research and innovation in WA blending new practices for entrepreneurship with industry partners
Social and cultural network	A place where art and culture meet technology and innovation. It seeks to become a creative hub for Perth
Urbanisation network	To serve as a living laboratory where research and education are both visible and accessible to all
Business and research network	Greater Curtin will showcase the network benefits to industry and researchers of access to fresh ideas and collaborators

 Table 5.2
 Developing network strengths through Greater Curtin master plan

#### 5.1.2 Greater Curtin – student accommodation

The first stage of the Greater Curtin program was the development of a student accommodation project to accommodate 1,500 students on campus. The process to deliver the student accommodation project was to seek expressions of interest (EOI) from suitably qualified companies. These were then filtered and a request for detailed proposals (RFDP) were requested from preferred bidders.

The data collection stage for this Chapter took place after the EOI had been advertised, but before the RFDP had been issued. As the interviews took place during the procurement process no interviewee was able to provide detail on individual bidders. This did not impact upon the verisimilitude of the interviews as they focused on how the accommodation project could or would be used to drive the delivery of teaching and research outcomes as well as the development of partnerships for innovation, rather than an investigation into the procurement process *per se*.

#### 5.2 Summary of findings

Chapter 5 has made the following contribution to the theoretical underpinning of partnership development at the university-industry-government nexus:

- Three themes emerged from the interviews:
  - projects will evolve over time as the potential to deliver teaching and research outcomes is better understood and widely shared;
  - there is a lack of consensus between internal partners as well as external advisors around how to engage, integrate and showcase teaching and research outcomes into the project;
  - provides evidence that the university was not maximising the potential of the project to drive Mission 1: teaching, Mission: 2 research, or Mission 3: engagement to develop partnerships for innovation;
- Proposes an institutional framework to illustrate the potential to progress partnerships, overtime, to deepen the engagement (Mission 3) of university and industry in the delivery of teaching and research outcomes (Missions 1 and 2) through capital infrastructure projects.

#### 5.3 Discussion of findings

The purpose of the interviews was twofold. One was to discuss how the Greater Curtin Stage 1 project had been developed and how different stakeholders within Curtin University were involved. The other element was to discuss emerging constructs from the literature, particularly the two elements that are evident in all bottom-up structures (namely open innovation and user engagement – see Publications 1 and 2 for further discussion) to see if the stakeholders in the project had realised the potential of the project to deliver teaching and research outcomes to enable partnerships for innovation at the university-industry-government nexus.

The findings from the interviews reveal three themes. The first is a general agreement that Greater Curtin, and specifically stage one of Greater Curtin, is the start of the process rather than the end model and the project is likely to evolve with each iteration. It is a process of learning by doing and that, depending on the outcome, the process could change and be adaptable (this is consistent with findings in Publications 1 and 2). The second theme concerns a lack of consensus about how and when to engage

academics, when to value, showcase and develop teaching and research outputs, and when to integrate research strengths into the project. The third theme concerns the creation of partnerships for innovation where there was some evidence of the university collectively not maximising the opportunity of the project to deliver across Mission 1: teaching, Mission 2: research and Mission 3: engagement.

#### 5.3.1 Theme 1 - evolution of the project over time

The interviews reveal a general level of consensus that the student accommodation project represented an important part of the Greater Curtin journey. There was a general level of support amongst all interviewees for the project and what it was seeking to achieve. There was agreement that the project has potential and that it is important to start the process of delivering it.

<u>External</u>: One of the issues the university suffered from is that it was such a big project, and the university didn't know where to start from.

Different views were offered regarding the engagement of the broader university in the project. The Project Control Group, made up from internal and external appointees who co-ordinated the development of the Greater Curtin project, were clear that academics had not been involved in the formulation of the project, but that academic involvement would be valuable.

<u>External</u>: We haven't reached out to the university with this project yet, but I can't see any reason why you couldn't incorporate it. The factual position is that we haven't done that at this stage. Could it have been useful? Yes, it could have been.

<u>Professional administrator</u>: Now is a period of no engagement because we are going through a procurement process rather than talking about what it's going to be.

This disconnect was recognised by the academic administrators who were looking to develop a process to effectively engage all parts of the university with an interest in the project. <u>Academic administrator</u>: I think there is a gap and it's the formalisation of an internal group that is representative of the different parts of the university in terms of contributing to Greater Curtin.

<u>Academic administrator</u>: Have we had involvement in Greater Curtin innovative precinct? Not really. They are very enthusiastic, but we're not involved formally in the process. I think they probably only have high level concepts.

However, despite the lack of a formal grouping or process it was felt that the academics were having an influence over the development project.

<u>Academic administrator</u>: My understanding as the result is that the RFP ... or EOI ... required a response to how they would engage the research, the academic community in the development. I think we are influencing it already.

There was also a node in the discussions about the initial ambition of the project. This is consistent with the idea that the university had to start somewhere and learn by doing to understand the potential of the project. The university recognised both that the process was important and whilst ambitious, there was more that could be delivered from future stages.

*External*: A large part of what I think we are doing is reasonably vanilla.

<u>Academic</u>: He [project lead] said this project has marginal financial outcomes which restricts how adventurous we can be.

The interviews revealed a predisposition, particularly amongst the administrators, towards learning by doing, rather than learning from other people's experience. In part this was due to a recognition that the scale of the project was such that the University needed to collectively understand the potential.

<u>External</u>: It's not a straightforward concept. It's a process right? How do you run a process to make sure you have the opportunity to get these sorts of things in?

<u>External</u>: I think the challenge for others is how do we build more of the university into this space? You can't let everyone have a go as you'd get

overwhelmed, but I think now is the time to start to get those ideas to build them into the next phase.

<u>Academic administrator</u>: The question is at what point do you bring the rest of the organisation along so they can start directing their resources appropriately?

#### 5.3.2 Theme 2 – engage, integrate and showcase research and teaching strengths

Whilst the interviews revealed a strong level of support for the Greater Curtin project and an appreciation that the project was the start of a process there was not unity over the best way to showcase, include or use the project to leverage teaching or research outcomes (Mission 1 and Mission 2). The academic interviewees (both research focussed and administrators) argued that more needed to be done to better integrate their work into the project, but also acknowledged that the process was not straightforward.

<u>Academic</u>: A general point is that sometimes institutions don't make use of inventions, the stuff that there is in place in their back yard and sometimes it's for good reasons – you know it's not that they haven't considered it; they have but sometimes it's because they don't value what they have produced, it's kind of you can't just do it because it's from you. If the same thing came out of Harvard, you'd say wow!

<u>Academic</u>: That's certainly an option we could discuss with potential builders about the opportunity to provide opportunity for research and work integrated learning and clearly that is something we would be motivated to do. The only issue about that is that research is inherently risky – you don't do it if you know the answer.

The interviews revealed the development of a conversation about how to showcase teaching and research strengths. This was a conversation more evidenced in the interviews with administrators than academics. Also, the administrative side of the university, responsible for project delivery, whilst supporting the approach of using the project to drive mission 1 and 2 outcomes were more circumspect and were not clear in how to unlock the potential.

<u>Professional administrator</u>: Its obviously not just monetary – is this aspirationally what we want to pay for? Culturally research is part of what we do. If we don't provide opportunities for research, we're not doing what we're supposed to be doing.

<u>Academic administrator</u>: Not just universities it's all organisations. It's a major problem with us to market products we work on because some people seem to have the view that the universities are full of great ideas but if only they would let them go, people would be able to come in grab them and develop then. That's not an accurate representation of what's going on. Where we see new products, we put in money and try to develop it piloting, co-fund all that stuff. But to try to get people to change the way they do things and shift behaviours is difficult and takes time. You really do have to demonstrate really robust and significant benefits for the change.

<u>Professional administrator</u>: You know should we, would we, constrain bidders and say look in addition to kind of all these other criteria you have to demonstrate how you are actually using research that we actually produce. We'd never go that far – you would hope there would be enough value being produced here that they would make use of that.

<u>Professional Administrator</u>: We weren't explicit. We felt that we were unable to be explicit because who does the developer ring up? No one in the university – it's no one's role in the university to take the phone call.

#### 5.3.3 Theme 3 - potential to progress Mission 1, 2 and 3 outcomes

There was also the emergence of a theme, revealed in Publications 4 and 5 (see Section 4.4 and 4.5), revolving around the potential of the project to progress teaching and research strengths in partnership with industry. At its heart is a conversation around how and when research and teaching can be leveraged through Greater Curtin. The traditional project approach would be when the buildings are opened, and this is also evidenced in the interviews, as was a disparate view of how or when to engage with teaching and research.

<u>Professional administrator</u>: research yes. it's also about advancing research capacity because part of what Greater Curtin stage 1 will deliver the

52

opportunity to house post-grad students and visitors and, you know, visiting fellows, scholars.

<u>Academic administrator</u>: The other part I haven't mentioned has been looking at trying to engage industries or other research organisations that may wish to be <u>tenanted</u> [author emphasis] within Greater Curtin.

<u>Professional administrator</u>: There are research and teaching space in there [the Stage 1 project].

<u>External</u>: We're giving another point where thought leadership and creativity can occur. We hope we are going to build space for this. So, we are going to facilitate that to occur through the built form and through that built form we are going to create an environment where people can come together and think about good ideas. We want to create areas that are for start-ups. People will come together for that definitive purpose – arguably doesn't exist on one area of the campus now.

<u>Professional administrator</u>: I don't think we do. We all want these opportunities. But no –one knows how to make it happen. No one has a KPI to say you must make this happen. It's difficult to join the dots. And so, if we can't join the dots how on earth does someone else do it.

The discussion evidenced above sets out responses to the issue of how and when to incorporate research into Stage 1. It reveals a range of views and timing for the engagement of teaching and research outcomes. There are four elements that were evidenced:

- Incorporating research into the project is difficult;
- Research will happen once the buildings have been opened or created;
- Once there is space then business can rent it, or the space can be used to drive partnership development; and
- Once the accommodation is built there will be more students on campus to undertake research.
There was acceptance, though mostly implicit rather than explicit, about the potential for the project and the need for deeper thinking about what it could deliver given time.

<u>Professional administrator</u>: We are extremely intelligent, but we don't run a business. What are the blockages? I think it's about the maturity – the business didn't understand the contribution it can make. It thought it was just producing graduates. Well, that's one way of looking at it. But what I am demonstrating with the Master Plan is that if you are developing a knowledge city and there are so many layers that it's not just about developing graduates.

However, this theme of potential was discussed by particularly academics about how to leverage research to develop partnerships that would deliver future stages of Greater Curtin. This latter theme also included a link to learning from other's experience and as such reflects a development in both the *modus operandi* of the project as well as its ambition.

<u>Academic</u>: They are still doing it, but basically, they learnt a lot of lessons in doing that – we could certainly cut something down quite quickly by learning their lessons in building that building.

Interviewer: That's research funding their infrastructure?

Academic: Yes.

<u>Academic</u>: Curtin wants to get external funds – external money is what it is all about. They don't want to be funding research projects within Curtin.

This does suggest a deeper engagement with the research and teaching community to develop an ambitious approach to the project. Both academics and professional staff understood that to do this would take time – and time is a constraint identified in the conversations suggesting that the university needs to create an environment where academics (and administrators) would be permitted to allocate time to developing projects for Greater Curtin.

<u>Professional administrator</u>: While one of the challenges we have, we often have, our researchers say we want to do this that, but the reality is when the rubber hits the road they fall away because they don't have time commitment

<u>Academic</u>: I'd be more than happy to do stuff, but we'd need to give my group time to do it.

Whilst time to develop projects was an issue, so too was the deeper understanding of the potential of all stages of the project, from project ideation and design to ribboncutting, to help deliver partnership development, innovation and delivery of teaching and research excellence.

<u>Interviewer</u>: I'm thinking if the Vice Chancellor gave the Deputy Vice Chancellor Research \$85m today and DVCR said 'I'll have you some research outcomes in 2019' she'd go 'you serious?'. But that's what we are doing with Greater Curtin. We're saying jam tomorrow.

<u>External</u>: I know what you are saying! DVCR could probably get a faster rate of return but over 30 years probably our \$85m will deliver greater benefits.

<u>Interviewer</u>: It's not either, or is it? We'll give you \$85m but as part of the process how do we engage them?

<u>External</u>: I'd answer \$50m is you would have to spend anyway. You need new SOBE anyway. \$30m for relocating the bus station, putting new road with all the facilities that a new road provides and creating the environment. That's how I might use all your true discretionary expenditure. You'd spend \$50m anyway.

<u>Professional administrator</u>: It's the same principle. You've got to try to open your arms. You've got to have partnerships; you've got to have that absolute desire to want to understand. It doesn't mean you have to say yes. You've got to want to understand.

The interviews also revealed an element of distrust between academic and professional staff. This was more strongly vocalised by the professional staff but also evidenced by academics and academic administrators.

<u>Academic</u>: The conversations haven't happened from their end but from my end to them. That's my idea of throwing it out there. I can't do anything more than that. I can only push it.

<u>Academic</u>: I remember saying they have my email, and they know what I want - I've sent it a number of times. I've sent it to Office for Research and Development to start with. The Vice Chancellor was also - I had her on the mailing list as well. Just so you know I got it in front of people's minds.

<u>Professional administrator</u>: As I said what typically happens is – I'll use an example – the non-smoking campaign. How did a particular group that said work our campaign before, during and after it, Sure. But we want \$200k to do it. It happens all the time. So why would I do that?

<u>Professional administrator</u>: Look going back to academics and academic freedoms is a problem because right now I would like to shut down every computer in the place that's not being used but academics say it's my right to leave it on and don't turn it off. Really?

<u>Professional administrator</u>: They [non-academic] said "you're on the road and an emergency vehicle might need to come in". "It's a food truck it can [swear word] move". So, you're closing me down? Yes, you're not complying. I said these are all footpaths there are no roads on campus. So, the guys parked on the edge of the footpath on the grass. So, he's done the right thing and moved off the path. It's not a road. Their argument was flawed – because they thought the university operated like a mine site. Again, culture had to change and then once they saw it was OK to have a food truck in the middle of the campus it made sense.

## 5.4 Development of an institutional framework to transition top-down structures

Publications 1 and 2 demonstrated that top-down structures were focused on innovation to drive economic growth. Publication 2 also found nothing inherent in the living lab structure that predisposes them alone to the creation of sustainable outcomes. However, it did suggest that the wider engagement with society (though open innovation and user engagement) is a reason why living labs are more predisposed to deliver innovation in a manner consistent with sustainable development. These findings, alongside the results of the interviews presented in this Chapter and Publications 4 and 5 are summarised in Table 5.3 in the form of an institutional framework.

The institutional framework illustrates the phases of a procurement cycle against which are plotted the potential for a university, with its partners to deliver Mission 1 (teaching), 2 (research) and 3 (engage) outcomes. The institutional framework illustrates the extent of the potential to progressively develop partnerships to enable innovation at the university-industry-government nexus.

The institutional framework is a temporal yet evolutionary adjunct to the Triple Helix model. It is temporal as it represents sequential points in time; it is evolutionary as it facilitates increasing levels of engagement over time; and it is an adjunct as it represents a developing procurement model for partners to use.

Taken together the framework provides a process for top-down, economic focussed partnerships to widen their remit, embrace those elements of the bottom-up approach that appear to be driving the adoption of sustainable innovation and thereby start to develop partnerships to enable low carbon and sustainable innovation at the university-industry-government nexus.

Based on the case study presented in Publication 5 a potential campus infrastructure project was divided into four stages.

- Stage 1 is the ideation phase where a university decides to invest capital (their own or a potential partners) into an infrastructure project;
- Stage 2 is the procurement phase;
- Stage 3 is the construction phase; and
- Stage 4 is occupancy where the university takes possession of or is given access to the asset.

The institutional framework offers insights into how to develop partnerships for innovation increasingly based on knowledge as well as learning by doing and how, over time, to deliver Mission 1, 2 and 3 outcomes in all aspects of infrastructure development.

The institutional framework demonstrates a progression in the incorporation of partnerships (Mission 3) to drive the delivery of teaching (Mission 1) and research (Mission 2) outcomes in an infrastructure project in a university setting over time.

• Early phase represents the traditional approach whereby a project is conceived, and university administration staff are tasked with project delivery. There is a procurement stage the result of which is the award of a contract to a company who then builds the facility. University academics might be involved in the process at project ideation (if it is an academic related project) and then will have access to the building on completion of construction. The result is that research and teaching can only happen once the ribbon is cut, and this is represented in the coloured box in Table 5.3. The early phase has similarities with a Mode 1 approach to knowledge production (discipline based – but in this case demarcated within the institution rather than based on academic discipline; not applied). This method is loosely referred to in the interviews as "jam tomorrow" procurement where delivery of teaching and research outcomes only happen after construction.

		Potential pa	ortners	
institutional progression	Stage 1: ideation & creation	Stage 2: Procurement	Stage 3: Construction	Stage 4: Occupation
Early phase Mode 1	U – administration	U – administration Industry	U – administration Industry	U – academic
				Delivery of TRO
Middle phase Mode 2	U – administration U – academic	U – administration Industry	U – administration U – academic Industry	U – academic Industry
		Potential for TRO	Delivery	of TRO
Mature phase Mode 3	U – administration U – academic Industry Users	U – administration U – academic Industry	ninistration demic y J	
	Potential for TRO		Delivery of TRO	

Table 5.3	Proposed institutional framework for partnership development at the UIG
	nexus to deliver teaching and research outcomes

Key: U = University; TRO = Teaching and research outcomes (Mission 1 and 2)

• Middle phase represents progression on the Early phase and is evidenced by the findings set out in Section 5.3.2. In the Middle phase academics are more closely involved in the process of project development and delivery (and this involvement would be even if the proposed project was not academic (for example delivery of student accommodation). The Middle phase provides academics with the opportunity to partner with potential bidders either through provision of existing research and knowledge or through the creation of new knowledge or teaching opportunities which would create value for the university and its partner. The middle phase has a closer affinity with Mode 2 knowledge production (transdisciplinary, applied with knowledge shared between partners).

In the Middle phase the potential for teaching and research deliverables could accrue potentially during the procurement phase (for example through research outputs such as this thesis) but certainly during the construction and occupation phases. The Middle phase provides the university with greater value and a clearer focus on Mission 1 and Mission 2. However, as a move away from the traditional approach represented in the Early phase, it would also create its own risk profile which would need to be managed.

Mature phase represents further progression still. It too is evidenced in the interviews (and Publication 5) but less explicitly than the Middle phase. It is, however, theoretically robust. The Mature phase presents a more nuanced approach to project development and delivery and has a close relationship to Mode 3 knowledge production (transdisciplinary, applied and shared with society).

<u>Professional administrator</u>: We are extremely intelligent, but we don't run a business. What are the blockages? I think it's about the maturity – the business didn't understand the contribution it can make.

The Mature Phase is significant because it is more focused on open innovation and the co-creation of projects between businesses and the university through the application of knowledge and experience from this and other projects throughout the project lifecycle. As such the potential to derive teaching and research outcomes are evidenced from Stage 1 (ideation) through to Stage 4 (occupation). Such learnings and knowledge will then be carried forward to the next iteration or project, which will benefit from knowledge created as well as learning by doing. As a progression from Middle phase (above) this iteration would also create its own risk profile which would also need to be managed. As illustrated, it could be developed further through the addition of users into the partnerships.

### 5.5 Discussion of proposed institutional framework

The focus of the framework is institutional as this research suggests that it is the institution, rather than partnership structures themselves that are preventing top-down structures from innovating in a manner consistent with sustainable development (see Section 4.4.4). The framework also associated the modes of knowledge production (see Table 2.2 for detail) to the phases of institutional development. This is an association rather than causal link, but it is a useful addition to the model as it helps to draw attention outside of the partnership structure to the changing way knowledge is produced and applied.

The framework also introduced into the institutional setting those factors (transdisciplinary teams and in the Mature phase users to co-create innovation) that may predispose living labs to be more focused on low carbon transition enabled by innovation at the university-industry-government nexus.

The framework reveals two further significant insights: one is that as the framework matures the ability to derive teaching and research value for the university, society and partners comes earlier in the project life cycle; potentially with each iteration.

The other is that each iteration would benefit from knowledge created either by the university or its partners. This would explicitly be brought to the programs ensuring that lessons from previous experiences are learnt and mistakes avoided. As a risk management regime this too will add value to the university. Risk would not be removed but management would be more knowledge based.

This framework is the result of research into the development of Greater Curtin Stage 1 and is therefore focussed on universities, but it could be developed as an *aide memoire* for industry and government. It is represented here as a linear process, but it too is likely to be a rotating helix.

60

### 5.6 Further research

The benefit of the institutional framework is that it is scalable and transfers knowledge forward so that future partnership development will not need to be focussed on learning by doing within the project but rather learning from what other people (and potentially other partners) have already done. The framework needs to be tested to be validated, but it does offer an evidence-based model for partnership development.

Further work would be useful to understand more deeply the motivators for and influences on project ideation as it is at this stage that the potential to drive economic, social and environmental outcomes are set. Publications 1 and 2 demonstrated that the majority of literature reviewed is focused on the economic benefits of partnerships for innovation. Publication 2 showed that there is a thin stream of work, derived in a bottom-up process, looking into economic, social and environmental outcomes from partnerships for innovation. Publication 3 also proposed that the use of the sustainable development goals be used to drive research. Further research on the process to agree the scope of potential projects would be informative. This issue is developed in Chapter 6.

### Chapter 6 Synthesis

This Chapter brings together the findings from the five publications, as well as Chapter 5 and applies the evidenced findings to the research questions. Section 6.1 presents a discussion of the research progression addressing each of the sub-questions and Section 6.2 provides a summary of the findings evidenced in the research.

### 6.1 Research progression addressing the sub-questions

The progression of the research through the publications and throughout this thesis is set out in Table 6.1. It demonstrates the contribution of each of the publications, as well as previously unpublished work presented in Chapter 5 to answering the research questions.

Table 6.1	Research progression through the thesis
-----------	-----------------------------------------

Publication	Primary research question	Summary of findings	Research progression through thesis
		Development of an evolutionary approach to the development of partnerships at the UIG nexus	Findings developed in Chapter 7
Publication 1 A systematic literature review of partnership development at the university-industry- government nexus	Which partnership models at the UIG nexus have been successful to	Living labs are the partnership structure that address economic social and environmental issues at the same time	Finding further explored in Publication 2
	deliver innovation? Develop partner society knowled	Development of hypothesis linking evolution of partnerships for innovation, globalisation of economy, society and environment and changes in modes of knowledge production	Findings developed in Chapter 7
Publication 2	Which partnership models at the UIG nexus have been successful to deliver low carbon and/or	Business cases differed due to the impact of three factors: why created; response to internal opportunities or external pressures	Findings developed in Section 6.2
Business models for sustainability in living labs	sustainable innovation?	Demonstrated Triple helix can be implemented locally; path dependency helps to understand decision making; value of being part of a consortium to learn from experience and by doing	Issues further examined in Publication 3

Publication	Primary research question	Summary of findings	Research progression through thesis
		Australian universities and industry are weak, when compared to OECD counterparts at partnering for innovation;	Ideas developed in Publications 4, 5 and Chapter 5
Publication 3	What factors promote or limit	Universities do not make the most of their campuses to drive partnership development	See Publication 4
If living labs are the answer what is the problem?	partnership development for a low carbon future?	Rate of external change makes partnership development complex and uncertain	See Chapter 7
		Proposed that SDGs should be used as the framework to drive partnerships at the UIG nexus	See Chapter 7
		Foray into process of organisational decision making	See Chapter 7
Publication 4	How can Australian universities become more effective at	Universities not developing experience implementing knowledge or of developing partnerships	See Chapter 5
universities' collective response to climate science	partnering with Government and industry to drive a low carbon future?	Evidence-based innovation could drive partnerships	See Publication 5
Publication 5: Happy homes – the relationship between homes and mental wellbeing.	How can Australian universities become more effective at partnering with Government and industry to drive a low carbon future?	Example of types of issues that an entrepreneurial university could tackle through partnership using infrastructure development	See Chapter 5

Publication	Primary research question	Summary of findings	Research progression through thesis
Chapter 5: Innovation at the university- industry-government nexus. The case of Greater Curtin	How can Australian universities become more effective at partnering with Government and industry to drive a low carbon future?	Development of three themes: projects evolve; lack of consensus as to how to integrate teaching and research outcomes; and not maximising potential to integrate Mode I,2 and 3 into partnerships for innovation	Proposes an institutional framework for partnerships development, which is used in Chapter 7

## 6.1.1 What partnership models at the university-industry-government nexus have been successful to deliver innovation?

This research has produced two findings to assist with answering this question. Firstly Publication 1 revealed an evolution in the development of partnerships for innovation at the university-industry-government nexus where partnership structures are becoming increasingly complex. This finding builds on, rather than replaces, the top-down/bottom-up presentation of the creation partnerships for innovation current in the literature.

The three stages, as revealed in Publication 1, are:

- Development of on-campus structures. Such structures include technology transfer officers or offices, space for business incubation, access to library services and information and potentially access to capital;
- Development of campus-adjacent structures. This stage is represented by the development of science and technology parks. These are developed proximate to a university, or cluster of universities and are designed to help industry work with universities to commercialise knowledge; and
- Development of living labs (and derivatives). These are partnership structures based on user engagement and open innovation. They are footloose and can be found in a variety of locations, including on-campus and campus-adjacent.

Secondly, Publication 1 also revealed that the partnerships were primarily focussed on delivering economic benefits from the commercialisation of knowledge with over 100 of the 132 articles reviewed being grounded in theories of economic development (see Appendix B.1 for full details).

Publication 1 also showed, in concert with other work (for example see [14]), the relative paucity of research looking at partnerships for innovation delivering economic, social and environmental outcomes at the same time. Publication 1 did however reveal that the only partnership structure that has been used to deliver economic, social and environmental outcomes at the same time is the living lab and its derivatives. This finding is summarised in Table 6.2 and is interesting because partnerships at the university-

industry-government nexus include the government, which is responsible for policy on sustainable development, and has agreed to the adoption of the sustainable development goals, and universities which provide some of the research that government's use to develop policy (for example climate science – this is discussed further in Publication 4).

Table 6.2	Partnership structures delivering innovation with economic, social and
	environmental outcomes

Evolution of Partnerships for Innovation	Effectiveness of partnerships to deliver innov		
On campus			
Campus adjacent	Economy	Society	
Living labs		Society	Environment

Publication 1 also revealed that there are a series of factors, outside the control of the partnership that are influencing the evolution of partnership development and potentially the focus of the partnerships.

The deepening of the knowledge economy and the increasingly ubiquitous access to the internet have accelerated the evolution in partnerships for innovation. Further, moves in Europe towards Responsible Research and Innovation [105] and the engagement of wider society in the development of research and innovation programs suggests a further pressure on the partnerships to evolve. Publication 1 refers to this move as the globalisation of society which is taking place alongside the globalisation of the economy and environment.

These findings led to the development in Publication 1 of a hypothesis (see Table 6.3) that proposed a link between the evolution of partnerships for innovation at the university-industry-government nexus, and the external influence of the globalisation of the economy, society and environment alongside the changing modes of knowledge production. This hypothesis is developed further in Table 6.3 and the development of a metastructure to enable innovation at the university-industry-government nexus is revealed in Chapter 7.

## Table 6.3Hypothesis of relationship between partnerships, globalisation and modes<br/>of knowledge production

Evolution of Partnerships for Innovation	Effectivenes	Mode of knowledge production		
On campus				Modes 1 and 2
Campus adjacent	Economy	Conintry		Modes 2 and 3
Living labs		Society	Environment	Mode 3

## 6.1.2 What partnership models at the UIG nexus have been successful to deliver low carbon and/or sustainable innovation?

As discussed, Publication 1 confirmed results elsewhere (for example see [14]) demonstrating a relative lack of research looking into the use of partnerships for innovation to deliver economic, social and environmental outcomes at the same time. Publication 1, as shown in Table 6.2, did reveal that the only structure used to produce innovation that was consistent with sustainable development outcomes was the living lab and its derivatives.

To answer this sub-question Publication 2 investigated the development of the business cases for four sustainability labs (SusLabs) that were part of the SusLabNWE consortium. It shows that the living lab approach is effective for creating partnerships for innovation to deliver sustainable outcomes, that the Triple Helix model can be applied in a local context and that members value being part of the consortium to share knowledge and reduce the burden of learning by doing.

Publication 2 shows that the factors affecting the development of the business cases were driven by the different objectives of each SusLab and that this reflected how each SusLab was originally created. It showed that business cases evolve according to internal and external pressures and opportunities.

Importantly, Publication 2 also found nothing inherent in the living lab structure that predisposes them alone to the creation of sustainable outcomes. However, it does suggest that the wider engagement with society (which in living labs focused on open innovation and user engagement (see Publication 1)) is a reason why living labs, rather than the other structures researched, are more predisposed to deliver innovation in a manner consistent with sustainable development. This issue is investigated further in Chapter 5.

Other research in this thesis contributed to the understanding of how other structures could be used to drive sustainable innovation. Publication 4 demonstrated that universities were not capturing the potential benefit of their research into climate science to develop partnerships for innovation. Publication 5, and Chapter 5, presented work that demonstrates the importance of the institutional setting to create partnerships for innovation. Publication 3 proposed the use of the sustainable development goals to drive the development of research and innovation at the university-industry-government nexus, as the factor limiting the creation of sustainable innovation does not appear to be structural, but rather institutional.

## 6.1.3 What factors promote or limit partnership development for a low carbon future?

This body of research has found no factors that predispose one type of partnership structure to low carbon, or sustainable innovation. It is posited above that the inclusion of users and open innovation in living labs is a factor in encouraging the development of partnerships for innovation that is more consistent with the objectives of sustainable development. As discussed in Section 4.4.4 and above, there is nothing structural to prevent the top-down structures from being used to deliver low carbon, and sustainable transition through innovation at the university-industry-government nexus.

Publication 3 found that Australian universities and industry, when compared to their peers in the OECD, were weak at collaboration for innovation, even though industry funds public research in Australia at a level higher than their OECD counterparts [48] and that 90% of Australian universities' research outputs are world class or better [51].

To understand further the reasons for poor performance in partnering for innovation Publication 4 investigated Australian universities adoption of climate research and the use of campus redevelopment to drive low carbon transition. It showed that in 2015 7% of universities have adopted a science-based carbon reduction target. Further it demonstrated that 20% of Australian universities were using their campuses to drive innovation in general, with one university committed to doing so in its long-term strategy.

69

Publication 4 presented an interesting question. Universities were committed to investing \$1.5bn capital into campus development in 2016, of this approximately \$500m was explicitly tied to the delivery of teaching and research outcomes. The research did not reveal how the remaining \$1bn would be used. But it suggests that universities were not using their resources effectively to drive Mission 1 and 2 or partnerships for innovation (Mission 3).

Publication 3 also attempted to explore, through a transdisciplinary methodology, the process of decision making (rather than the factors that affect decision making that were raised in Publication 1). This was ultimately unsuccessful although logically sound and worthy of further investigation (see Chapter 8 for recommendation).

## 6.1.4 How can Australian universities become more effective at partnering with government and industry to drive a low carbon future?

Having resolved that there is not one form of partnership structure that is inherently better, or more sustainable than others the research turned to focus on the potential to use campus redevelopment to develop partnerships for innovation.

To do this Publication 5 and Chapter 5 used Curtin University's proposed development of a 1,500-bed student accommodation project to understand how a university could drive teaching and research outcomes, partnerships and innovation through such a project.

Publication 5 scoped the issue of mental health and housing and found that at the non-intervention (i.e. those who are mentally healthy) end of the mental health continuum there is little in the literature reviewed to help an individual use their accommodation to support good mental health. Much of the advice (pre-COVID-19) surrounded the importance of leaving your accommodation to get better, as isolation was likely to be inconsistent with furthering positive mental health. The paper found several issues on which a student accommodation project could be used to both research the links between accommodation and mental health and partner with industry to undertake translational research to test solutions.

The practitioner interviews, described in Chapter 5, led to the creation of an institutional framework to illustrate how universities could progressively engage with

industry over the life cycle of an infrastructure project (see Table 5.3). The significance of this institutional framework is that it offers universities, industry and government an understanding of the potential to use infrastructure projects to drive innovation in partnership. It shows how each partner would achieve tailored benefits through partnering in the project.

Using student accommodation as an example, universities would benefit through research outcomes; business would benefit by being able to offer a better product and government (as the representative of broader societal interest) would benefit through the commercialisation of research outcomes as well as student accommodation designed to support positive mental health.

### 6.2 Summary of evidenced research contribution to each sub-question

The theoretical contribution of the five publications (Appendix B.1 to B.5) and Chapter 5 is detailed below.

The overall research question was: how can universities, industry and government together drive low carbon innovation and transition?

The findings presented so far in this thesis evidence that:

- There is an institutional, rather than structural, weakness in developing partnerships for innovation that contribute towards the delivery of sustainable development;
- The Triple Helix model can be used to facilitate the delivery of sustainable development; and
- The institutional response could be to place the sustainable development goals at the heart of the triple helix, around which each helix revolves.

### 6.2.1 Contribution: Sub-question 1

<u>Sub-question 1</u>: Which partnership models at the university-industry-government nexus have been successful in delivering innovation?

The findings presented in this thesis addressing sub-question 1 evidence that:

- The Triple Helix model is widely cited in the literature and has successfully been applied locally; other models (for example the Quadruple and Quintuple Helix) have unresolved weaknesses;
- The partnership structures are evolving over time and due to external influences (such as the globalisation of the economy, society and environment) becoming increasingly complex; and
- The predominant focus in the partnerships reviewed is on the delivery of economic benefits to society with only a small sub-set seeking to promote economic, social and environmental benefits. There is no evidence of structural reasons for this (for example the Triple Helix model is used to deliver sustainable innovation through living labs).

### 6.2.2 Contribution: Sub-question 2

<u>Sub-question 2</u>: Which partnership models at the university-industry-government nexus have been successful in delivering low carbon and/or sustainable innovation?

The findings presented in this thesis addressing sub-question 2 evidence that:

- The Triple Helix model can be applied locally and to the development of innovation to deliver economic, social and environmental outcomes;
- There is nothing inherent in their structure that suggests only living labs are appropriate to the delivery of sustainable innovation; and
- Business cases evolve due to internal and external opportunities and pressures.

### 6.2.3 Contribution: Sub-question 3

<u>Sub-question 3</u>: What factors promote or limit partnership development for a low carbon future?

The findings presented in this thesis addressing sub-question 3 are that:

- No factors were found that predisposed one type of partnership structure to low carbon, or sustainable innovation;
- The inclusion of users and open innovation in living labs may make it more amenable to the delivery of innovation consistent with the objectives of sustainable development;

- Australian universities and industry, when compared to their OECD counterparts, are poor at collaborating for innovation; and
- There is evidence that universities are not maximising the potential of their research findings or capital investments to develop partnerships for innovation.

### 6.2.4 Contribution: Sub-question 4

<u>Sub-question 4</u>: How can Australian universities become more effective at partnering with Government and industry to drive a low carbon future?

The findings presented in this thesis addressing sub-question 4 are that:

- The potential to use campus redevelopment to progress all 3 university missions (teach, research and engage) is currently underutilised;
- All three missions can be developed using the institutional framework developed as part of this thesis; and
- The institutional framework can be used to further low carbon and sustainable innovation.

### Chapter 7 A new metastructure to enable innovation at the university-industry-government nexus

There has been discussion at length in this thesis concerning the evolution of partnerships for innovation and the development of a hypothesis to link this evolution to the globalisation of the economy, society and environment. There has also been discussion about the top-down/bottom-up focus in the existing literature to explain the creation of partnerships for innovation and how this interacts with the evolution of partnerships for innovation. This Chapter draws these discussions together to develop a metastructure to enable innovation at the university-industry-government nexus.

Section 7.1 to 7.3 presents a theoretical application of the research findings evidenced in Chapter 6 to the existing theoretical models. Section 7.4 discusses the implication of the theoretical contribution and Section 7.5 discusses its limitations. In response Section 7.6 proposes the development of a metastructure to enable innovation at the university-industry-government nexus which is discussed in detail in Section 7.7. Section 7.8 summarises the proposal, with Section 7.9 integrating the metastructure and the institutional framework. Section 7.10 discusses limitations of the proposed metastructure. Finally, Section 7.11 proposes a future research agenda.

### 7.1 Realising the potential of a Triple Helix partnership

This section firstly applies the research findings evidenced in Chapter 6 to the theoretical partnership models for innovation at the university-industry-government nexus which provides a substantive contribution to the overarching research question: how can universities, industry and government together drive low carbon innovation and transition?

Publication 1, along with others, demonstrated that within the literature there is a dearth of articles concerning how to develop partnership structures to further social, economic and environmental benefits to the partners and to society more widely. As discussed extensively the only structure that has focussed on the delivery of sustainable innovation (i.e., innovation that furthers societal interests of economic development, social cohesion and environmental rectitude) is the living lab structure.

As also discussed this thesis found no evidence of structural reasons preventing top-down structures from being used to develop partnerships to innovate in a manner seeking to deliver economic, social and environmental benefits at the same time. This suggests that the reasons for the current focus being on economic benefits are institutional rather than structural.

#### 7.2 Summary of current models proposed to drive low carbon innovation

The existing literature proposes two models to develop partnerships for innovation that help to deliver sustainable innovation through the addition of a fourth and fifth helix to the Triple Helix model (see Section 2.2 and Appendix A.9 and A.10).

Both proposals have limited merit. The addition of a fourth helix – proposed to represent the interests of society is interesting as it raises the issue of how the interests of the global citizen (rather than the locally impacted citizenry) are represented. This key question is, however, currently left unanswered by the addition of a fourth (quadruple) helix (see Appendix A.9 for an extended discussion).

To represent the interests (or otherwise) of the environment a fifth conceptual helix is added to create the Quintuple Helix. The role of the fifth helix is to represent both planetary boundaries and to drive a more environmentally enlightened decision-making process. As discussed, the Quintuple Helix is conceptually weak and does not offer an effective explanation as to how the additional fifth helix would work in practice to drive at the partnership level, innovation that benefits the global citizen as well as the local partners. The conceptual problems with the proposed fifth helix are discussed in more detail in Section 2.2.3. and Appendix A.10. The unresolved problem of the Quintuple Helix is that there remains a scalable disconnect between the local delivery of planetary boundaries but without a mechanism to understand what the boundaries are. This is a restatement of the problem of the un-managed commons [91] that under the Quintuple Helix model remains unanswered.

# 7.3 How can universities, industry and government together drive low carbon transition through innovation?

The evidence presented in this research is that there are a minority of partnership structures that are delivering sustainable innovation. The evidence presented shows that top-down structures are not delivering innovation that is consistent with a low carbon, or sustainable future. This research has also shown that there is nothing inherent in top-down approaches that precludes the adoption of sustainable innovation (see Section 4.4.4).

This research has shown that there is an institutional rather than structural issue that lies at the heart of the challenge to contribute to low carbon and sustainable transition through innovation at the university-industry-government nexus.

Therefore, the challenge for this research is to find a way to create an institutional response so that all partnerships developed (top-down or bottom-up) will contribute to a low carbon and sustainable transition.

One proposal, developed from Publication 3, is to use the sustainable development goals (see Appendix A.4.1) as the guiding principles under which to create the institutional response. It is a surprisingly simple conclusion, which would need further applied research, but applying the sustainable development goals to the high-level triple helix framework would result, in theory at least, in all partnerships being informed by, and contributing to, the delivery of the sustainable development goals.

The advantage of adopting this approach to the broader Triple Helix partnership is that the partners are extremely well informed. Governments unanimously agreed to the adoption of the sustainable development goals and are responsible for policy; universities research and teach the goals and business has a direct business interest in operating sustainably. Making sure innovation happens in a manner that is appropriate for the 21st century and addresses the transdisciplinary issues that beset society would be a significant step towards a sustainable society.

### 7.4 Implications of findings

The proposed modification to the Triple Helix framework model to incorporate the sustainable development goals as a core, unifying proposal around which the triple helices revolve is new to the literature. Including the SDGs as a unifying principle around which to develop low carbon or sustainable innovation ensures that all partners understand what they are seeking to deliver for society in the widest sense, as well as for themselves (in the narrowest sense).

76

This innovation to the model leaves in place the triadic imperative discussed by Etzkowitz, and borrowed from Simmel, Marx and Webber. It also resolves the shortcomings of the Quadruple and Quintuple Helix models (see Sections 2.2 and 2.2.3 and Appendix A.9 and A.10).

The model now addresses the un-managed commons by demanding that partnerships for innovation focus on achieving global priorities by ensuring that local decisions are consistent with those global priorities. However, it is not didactic as it leaves to each local partnership the imperative of how to innovate in a manner that is consistent with and contributes to the delivery of the sustainable development goals. It is not telling each partnership what to do, but rather it is letting each partnership decide how best to contribute towards the delivery of sustainable development as well as their local, national or international partnership.

The approach future proofs the partnerships by utilising an international agreement that was agreed unanimously by all member nations of the United Nations.

The approach helps to increase the relevance of the sustainable development goals by putting them at the centre of policy development and partnership delivery. This creates a uniform basis for innovation that is as relevant to industrial democracies and autocracies and less developed economies and societies.

Some partnerships (a limited number) are already delivering economic and social outcomes from innovation; a smaller number are still innovating to produce economic, social and environmental outcomes at the same time. This proposal would not adversely affect these partnerships, but any unintended consequences would need to be further understood.

Finally, implementing the sustainable development goals as the centre piece to which the partners revolve around also reflects the move toward open science and reflects the move to mode 3 knowledge production where knowledge is produced in a transdisciplinary plane and this transdisciplinary approach will be continued into the delivery of the innovation. Implementation can prepare partnerships for a progression to transdisciplinary innovation, a progression that is already evidenced in the literature. This issue is discussed further in Section 7.6.

77

### 7.5 Limitation of proposal

The proposed addition (see Section 7.3) of the sustainable development goals to become the core of the Triple Helix model is new to the literature. It is a simple yet potentially powerful contribution to the theoretical literature to help create partnerships to develop low carbon and sustainable innovation.

However, whilst an important theoretical contribution it does little to draw together the various theoretical appendages that contribute to the development of partnerships for innovation at the university-industry-government nexus. The proposed metastructure, developed in Section 7.6 helps in this endeavour by demonstrating the interrelationship between the internal and external drivers of partnerships for innovation and within this metastructure the theories and models discussed in this thesis regarding partnership development would sit. The metastructure also helps to understand why top-down partnerships have not been used to drive sustainable innovation and yet living labs have.

### 7.6 Development of a metastructure to enable innovation at the universityindustry-government nexus

As discussed, the development of the actual partnerships heretofore has been seen to be either top-down or bottom-up initiatives and the location of the partnership has not been seen to be important. This research is proposing a new metastructure which focusses on where the partnership is happening and the factors that are influencing partnership development.

Top-down initiatives are management led and reflect the intentions of the broader partnership framework. Top-down initiatives include the creation of technology transfer teams or officers, access for business to library resources, or the creation of incubator space as well as the creation of forms of venture capital funding [39].

In the metastructure this stream of development, as identified in Publication 1 is largely within the physical confines of the university campus and is referred to as oncampus development. Work presented in Publications 4, 5 and Chapter 5 would sit within Stream 1. The next step in the evolution is Stream 2 which is the creation of campus adjacent structures. These are widely referred to in the literature and consist of science and technology parks, innovation parks and derivatives. The creation of stream 2 structures is also seen in the literature (Publication 1) as being a largely top-down response within the Triple Helix structure.

However, the literature also describes a set of responses that are bottom-up. They come from within the partnership, but they are not management led. This bottom-up development of intermediaries for innovation encompasses a more diverse set of structures, which in the literature is labelled as living labs (and derivatives) and represents a third stream of development of partnerships for innovation.

Stream 3 may then be further divided into two; where Stream 3a represents living labs that are focused on innovation to deliver economic and social innovation and Stream 3b (a smaller subset) that focusses on innovation that delivers economic, social and environmental outcomes at the same time. A summary of the development of the three broad streams is presented in Table 7.1.

Stream	Examples
Stream 1: On campus structures	Access to libraries, business incubators; technology transfer offices
Stream 2: Campus adjacent	Science and Technology Parks
Stream 3: Living labs	A rapidly changing partnership structure; possibly unique to each partnership but focused on user engagement and open innovation

Table 7.1Streams of partnership development in the metastructure

# 7.7 Integrating exogenous factors and modes of knowledge production into the metastructure

One important aspect of the proposed metastructure is that it links factors exogenous to the partnerships to the evolution of the partnerships. For example, the interesting point to note about Stream 3 it has arisen at the same time as the development of access to the internet and the changes in knowledge production. The literature debates in detail the changing of modes of knowledge production (see for example [20, 21]) but a contribution of this thesis is to link the evolution of partnerships for innovation to these exogenous factors as well as the development of modes of knowledge production. This finding is developed from the hypothesis presented in Publication 1 and is set out in Table 7.2.

External stimuli	Manifestation in society	Mode of knowledge production	Extant model in the literature	Partnership stream	Spheres of interest
Globalisation of the economy	Deepening of the knowledge economy	Mode 1 and Mode 2	Triple Helix Model	Stream 1 Stream 2	Economic
Globalisation of society	Democratisation of knowledge	Mode 2 and Mode 3	Quadruple Helix Model	Stream 3a	Economic Social
Globalisation of the environment	Sustainable development goals (etc)	Mode 3	Quintuple Helix Model	Stream 3b	Economic Social Environmental

Table 7.2 Development of the metastructure to enable innovation at the universityindustry-government nexus

This three-stream metastructure is an evolutionary addition to the top down/bottom-up model. Importantly it does not replace the top-down/bottom-up emphasis, but rather uses it to demonstrate how the partnerships are evolving (i.e., what factors are driving the development of the partnership structures).

The contribution of the three-stream metastructure is to illustrate developmental progression (see Table 7.3) in the creation and deployment of partnerships for innovation which happen reflexively as opportunities arise (i.e., in response to an opportunity or threat) rather than in the current didactic orientation presented in the literature.

As discussed in Chapter 1 and 2, in detail in Appendix A and Publication 1, and presented in Table 7.2, stream 1 and 2 partnerships for innovation are predominantly driven by the delivery of economic growth. The metastructure suggests this is because Stream 1 and 2 structures were created before the globalisation of society and the democratisation of knowledge [16] and in a period when modes of knowledge production were transitioning from Mode 1 to Mode 2.

Stream 3 structures were created during or after this period and their structures therefore reflect a deeper engagement with users and an emphasis on open innovation. This association is presented in Table 7.3. The fact that Stream 1 and 2 structures remain focused on economic growth may be linked to path dependence theory (see Publication 2) and Stream 1 and 2 partnerships therefore need a process to help them move to structures that more closely reflect the external environment within which they sit. This is the institutional framework presented in Chapter 5.

Stream	Orientation	Date	Location	Innovation outcome
Stream 1	Top down	1960 and onwards	On campus structures	Economic
Stream 2	Top down	1960s (USA) 1980s (UK) 1990s (EU)	Campus adjacent structures	Economic Social (limited)
Stream 3a	Bottom-up	2000s	Living labs (and derivatives)	Economic Social
Stream 3b	Bottom- up	2000s	Living labs (e.g., SusLabs)	Economic Social Environmental

 Table 7.3
 Three stream metastructure of partnership development

The literature shows an increasing interest in partnerships for innovation (Publication 1) and particularly the bottom-up approaches (see also [14]). This level of interest has increased in concert with the increasingly ubiquitous access to the internet and coincides with the development of open science and responsible research and innovation initially launched in Europe [106] but also America and Australia [105]. This research has not found a causal link, but Publications 1 and 2 do suggest a connection worthy of further research (see Table 7.4 and further discussion in Chapter 8).

External stimuli	Innovation model	Stream	Innovation response	Partnership response
Deepening of the knowledge economy	Triple Helix	Stream 1 & 2	Entrepreneurial university [8]	Incubators Venture capital Science and technology parks
Democratisation of knowledge	Quadruple Helix	Stream 2 & 3a	Service dominant logic [107] Open innovation [17] User innovation [14] User centred design [108] Social innovation [109]	Living lab Urban living labs
Sustainable Development Goals	Quintuple Helix	Stream 3b	Frugal innovation [110]	Sus Lab Smart Cities

Table 7.4 Hypothesis of partnership response to external stimuli

### 7.7.1 Stream 1: Innovation Partnerships – on campus developments

Publication 1 presented on-campus structures as largely concerning the creation of new structures. Publications 3, 4, 5 and Chapter 5 have led to a re-evaluation and consequently Stream 1 therefore consists of two parts. Part 1 is the creation of new organisational structures and infrastructure to create partnerships for innovation (as described in Publication 1). Part 2 focusses on institutional processes to enable using the existing infrastructure in a different manner (as described in Publications 3, 4, 5 and Chapter 5). This is presented in Table 7.5.

The literature reviewed for Publication 1 revealed that partnership structures were often binomial (that is an arrangement between a university and industry, but without government's active involvement). This created a theoretical quandary in the literature as it suggests a binomial relationship is not consistent with the Triple Helix relationship. This issue, amongst others, made the Triple Helix too nebulous to be useful in real partnerships on the ground.

For the metastructure this is not an issue as the focus is on partnership creation between different institutions to commercialise the knowledge to create innovation. These partnerships do not require the active involvement of government, and as such the metastructure is a relevant contribution to binomial or trinomial partnerships.

Stream 1		Examples	Author
Part 1	Creation of <b>new</b> structures	Technology transfer offices Academic liaison officer Business access to library services Business incubation	[80] [81] [79] [82]
Part 2	Either using <b>existing</b> infrastructure in a different manner; or rethinking processes	Opening up the campus to new usages Institutionalisation Campus renewal through planning Infrastructure redevelopment	[111] [112] [113] Chapt 5

Table 7.5Stream 1 – on campus structures

For example, the university-industry partnerships structures, as put forward by Universities Australia in Table 7.66, make sense in the metastructure. These forms of binomial partnership can vary from the formal to the informal [51] and may occur in stream 1, 2 or 3 partnership structures. This issue is discussed further in Chapter 8.

Table 7.6	Forms of	university-industry	partnership
-----------	----------	---------------------	-------------

University/business relationship	Type of relationship	
Contract	Formal	
Co-location		
Expert in residence		
Collaborative		
Student placement		
Innovation networks		
Industry advisory groups	Informal	

### 7.7.2 Stream 2: Development of campus adjacent structures

Stream 2 is the creation of off-campus, or campus adjacent structures to further the partnership between universities, industry and government. The impetus for the creation of these off-site structures appears to be university, or government driven. They are geographically proximate and tend to focus on the research strengths of the university.

The effectiveness of such structures in successfully commercialising knowledge created in universities is much debated. There is evidence, presented in Publication 1, of the value park users attach to being part of a community of interest. The informal value park users place on being part of a community is a finding also reflected in Publication 2.

#### 7.7.3 Stream 3: Living labs and derivatives

Living labs are widely discussed in the literature, Publication 1 and Publication 2. As acknowledged in Publication 1 living labs are evolving at such a rate that they are difficult to define, or to develop best practice models. Stream 3 is subdivided into two separate (but co-existing) typologies that reflect the nature of the literature revealed in Publication 1.

Stream 3a focuses on living labs and urban living labs. These are user focussed and co-create innovation. They typically embrace the views of society but do not necessarily help society move towards the delivery of sustainable development.

Stream 3b however occupies a thinner sub-section in the literature (see Publication 1 and [14]) that uses a living lab methodology to explicitly seek to move society towards sustainable development through user centred innovation.

# 7.8 Summary of the emerging metastructure to enable innovation at the university-industry-government nexus

The emerging metastructure to enable innovation at the university-industrygovernment nexus links the evolution of partnership structures, globalisation of the economy, society and environment and the observable changes in modes of knowledge production into one metastructure and this is presented in Table 7.7. Importantly it is reflexive of what is happening in actual partnerships reviewed for this thesis and is theoretically valid for binomial (between universities and industry) and trinomial (between university, industry and government) partnerships.

As discussed above the evolution in partnership development reflects the increasingly universal access to the internet and knowledge. However, whilst the evolution of partnership development might be influenced by factors external to the partnership it does not replace the importance of the top-down/bottom-up approach to partnership development endemic in the literature; indeed, as discussed (see Section 4.1) it builds upon that approach.

One important finding from combining the top-down/bottom-up approach to the evolutionary description of the creation of partnership development is that the top-down approach to partnership development (a finding in Section 4.1.2 and Publication 1) primarily focusses on the delivery of economic outcomes (with flow on benefits to economic actors in the economy), and that it is the bottom-up approach that seeks to deliver innovation in a manner that considers society (Stream 3a) and the three pillars of sustainable development (Stream 3b). This suggests there is a weakness in the partnership creation to properly address low carbon or sustainable development, which may be rectified by the application of the sustainable development goals as the core feature of the Triple Helix model.

Table 7.7Emerging framework for the creation of intermediaries for innovation at<br/>the university-industry-government nexus

External stimuli	Manifestation	Mode of knowledge production	Existing Model	Partnership stream	Response to sustainable development
Globalisation of the economy	Deepening of the knowledge economy	Mode 1 Mode 2	Triple Helix	Stream 1 Stream 2	Economic
Globalisation of society	Democratisation of knowledge	Mode 2 Mode 3	Quadruple Helix ³	Stream 2 Stream 3a	Economic, Social
Globalisation of the environment	Sustainable development goals	Mode 3	Quintuple Helix ³	Stream 3b	Economic, Social Environmental

### 7.9 Integrating the metastructure and institutional framework

The proposed metastructure links partnerships for innovation, the globalisation of the economy, society and environment and the changes to modes of knowledge production. The institutional framework presented in Section 5.4 evidences the evolution of partnerships from an early to mature phase in terms of their ability to deliver Mission 1, 2 and 3 outcomes through open and co-creative partnerships in a manner that will be more predisposed to creating a low carbon and sustainable transition through innovation at the university-industry-government nexus.

³ The Quadruple and Quintuple Helix are included in Table 7.6 as they are presented in the literature as a means of moving towards sustainable innovation. As discussed above both have significant shortcomings.

There is a close relationship between the phases in the institutional framework, the streams in the metastructure and the changing modes of knowledge production and this is presented in Table 7.8. It is through this affinity that the institutional and structural dislocation evidenced through the published works, and analysis presented in Chapters 4-7 are addressed through the metastructure within which the institutional framework resides.

Table 7.8	Relationship between proposed metastructure, institutional framework
	and modes of knowledge production

External stimuli	Manifestation	Mode of knowledge production	Partnership stream	Institutional framework
Globalisation of the economy	Deepening of the knowledge economy	Mode 1 Mode 2	Stream 1 Stream 2	Phase 1
Globalisation of society	Democratisation of knowledge	Mode 2 Mode 3	Stream 2 Stream 3a	Phase 2
Globalisation of the environment	Sustainable development goals	Mode 3	Stream 3b	Phase 3

# 7.10 Using the metastructure and institutional framework to enable low carbon transition at the university-industry-government nexus

The incorporation of the Sustainable Development Goals to the heart of the partnership structure, as proposed in Section 7.3, will help, in theory, to further the potential of innovation to deliver against a wider remit (as summarised in Table 7.9).

	Location	Partnership creation	Predominant contribution to sustainable development
Stream 1	On campus	Top-down	Economic
Stream 2	Campus adjacent	Top-down	Economic
Stream 3a	Foot loose	Bottom-up	Economic and social
Stream 3b	Foot loose	Bottom-up	Economic, social and environmental

## Table 7.9Relationship between evolution of partnership streams and top-<br/>down/bottom-up partnership creation

However, a more fundamental proposal, as evidenced through the research in this thesis, is to apply the institutional framework presented in Chapter 5 (see Table 5.3 and Section 5.4) to Stream 1 and Stream 2 partnerships. The institutional framework is based on evidence revealed in this research that open innovation and user engagement (enabled by the deepening of the knowledge economy, the globalisation of society and the change in modes of knowledge production) are the two elements that are common to living lab partnership structures (see Publication 1)

The institutional framework proposed in Chapter 5, based on interviews with practitioners and evidence from Publications 4 and 5, then develops a process for partners to apply these two traits to the development of existing partnerships. It is important to recognise that applying these traits to partnerships does not mean all partnerships will be living labs. Rather it is responding to the globalisation of society and the changes in knowledge production that is the impetus for the institutional framework (and this is discussed in Section 5.3).

#### 7.11 Further research

The development of the metastructure can enable innovation at the universityindustry-government nexus for the transition towards low carbon and sustainable innovation as it links external influences (globalisation of the economy, society and environment) to the core of the partnership development. This suggests that partnerships at some stage in their evolution will need to respond to the globalisation of the economy, society and environment and the addition of the sustainable development goals will provide evidence-based criteria to drive the development of intermediaries for innovation.

Much applied work will be required to test the metastructure described above, as well as the proposal discussed in Chapter 6 to develop best practice guides and manuals and start the process of delivering innovation at the university-industry-government nexus that will contribute towards the delivery of sustainable development.

87

### Chapter 8 Conclusions and recommendations

This Chapter is structured as follows. Section 8.1 presents a summary of the research findings from each paper (see Table 8.1) to complement the presentation of research findings addressing the substantive question and the four sub-questions presented in Section 6.2 above. Section 8.2 then discusses the contribution of this body of work to the body of knowledge. Section 8.3 then recommends further research.

### 8.1 Summary of research findings

The body of research evidenced in this thesis forms five publications and Chapter 5 is summarised in Table 8.1.

### 8.2 Original and significant contribution

This thesis and associated publications (in Appendix B.1 to B.5) provides an original and significant contribution to the knowledge and understanding of low carbon and sustainable partnerships for innovation at the university-industry-government nexus in the following ways:

- Finds that top-down partnerships are focused on economic development whereas bottom-up partnership structures are more likely to be used to develop social and environmental innovation;
- Finds that there are no structural impediments preventing top-down partnership structures from being used to progress sustainable innovation, which suggests an institutional rather than structural problem;
- Develops a metastructure to enable innovation at the university-industrygovernment nexus consisting of three streams of partnership development: Stream 1: on campus; Stream 2: campus adjacent and Stream 3: living labs;
- The proposed metastructure to enable innovation at the university-industrygovernment nexus links the evolution of the development of partnerships for innovation to the globalisation of the economy, society and environment and the changing modes of knowledge production;
- Finds that there are institutional, rather than structural impediments that appear to mean that top-down structures are focused primarily on economic growth;

Table 8.1	Summary	of evidenced	findings	from the	body of	of work
-----------	---------	--------------	----------	----------	---------	---------

Publication	Finding
Publication 1: A systematic literature review of partnership development at the university-industry- government nexus	<ul> <li>Builds on top-down/bottom-up analysis discussed the literature by demonstrating that the evolution of partnerships for innovation are driven by factors within the partnership but also by external to the partnership;</li> <li>Describes three stages in the evolution of partnerships for innovation, which can be concurrent: namely the development of on campus structures; campus adjacent structures and living labs;</li> <li>Finds only living labs (and derivatives) tackle economic and social issues; a smaller subset tackle economic, social and environmental issues;</li> <li>Demonstrates that most research in the literature is grounded in economic theories; and</li> <li>Developed a hypothesis linking evolution of partnerships for innovation, globalisation of economy, society and environment and changes in modes of knowledge production.</li> </ul>
Publication 2 Business models for sustainability in living labs	<ul> <li>Each business case was different due to the impact of three factors:         <ul> <li>The principles or purposes underpinning the SusLab's creation;</li> <li>How each entity responded to pressures or opportunities they faced during operation; and that</li> <li>In response to internal of external opportunities the business cases evolve over time</li> </ul> </li> <li>The article shows how the Tiple Helix model can be implemented locally and used on innovation that contributes to sustainable development; it shows how path dependence theory helps to understand the decision-making process to develop business cases, and it shows the value members of SusLabNWE place in being part of a wider consortium to assist with learning from experience and from doing</li> </ul>
Publication	Finding
-------------------------------------------------------------------------------------------------------	-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
Dublication 2	<ul> <li>Australian universities are weak, when compared to OECD counterparts at partnering with business for innovation;</li> <li>Business is weak at partnering with Australia universities and as well as customers and suppliers;</li> </ul>
If living labs are the answer what is the question?	<ul> <li>Business is weak at partnering with Australia universities and as well as customers and suppliers,</li> <li>Universities do not make effective use of their campuses to drive partnership development;</li> <li>Rate of external change makes partnership development complex and uncertain;</li> <li>Little evidence of the development of innovation partnerships to drive sustainable innovation at scale; and</li> <li>Proposed that SDGs should be used as the framework to drive partnerships at the UIG nexus</li> </ul>
Publication 4: A study of Australian Universities' collective response to climate science	<ul> <li>Three Australian universities committed themselves to a carbon reduction target in line with climate science.</li> <li>For those 10 Australian universities who were required to report emissions data, their emissions rose by 4.63% between 2010/11 and 2014/15.</li> <li>Together Australian universities were committed to investing over \$1.5bn in capital infrastructure on their campuses in 2016.</li> <li>Eight universities (20%) committed to using their campus' as living labs, or similar, to deliver teaching and research outcomes.</li> <li>Revealed limited evidence of universities willingness to partner with themselves (to use their campus infrastructure to drive teaching and research outcomes)</li> </ul>
Publication 5: Happy Homes – the relationship between homes and mental wellbeing	<ul> <li>Little in the literature reviewed addressing how to maintain or promote mental health in accommodation for the self-help/social support end of the mental health continuum</li> <li>Needs to be a focus on transdisciplinary (to integrate all relevant disciplines) and translational (to develop practical interventions that work) research</li> <li>Needs to be a focus on building resident to drive policy outcomes (current focus is on how to use building fabric to deliver for example climate targets)</li> <li>Study shows the potential benefits of undertaking transdisciplinary translational research to UIG partners.</li> </ul>

Publication	Finding
	Three themes emerged from the interviews:
Chapter 5:	<ul> <li>Theme one reveals that projects will evolve over time as the potential to deliver teaching and research outcomes is better understood and widely shared;</li> </ul>
Innovation at the university-industry- government nexus. The case of Greater Curtin	<ul> <li>Theme two reveals there is a lack of consensus between internal partners and external advisors around how to engage, integrate and showcase teaching and research outcomes into the project; and</li> </ul>
	<ul> <li>Theme three provides evidence that the university not maximising the potential of the project to drive Mission 1: teaching, Mission: 2 research outcomes, or Mission 3: engagement to develop partnerships for innovation;</li> </ul>
	• Proposes an institutional framework to illustrate the potential to progress partnerships, overtime, to deepen the engagement of university and industry in the delivery of teaching and research outcomes through capital infrastructure projects.

- Finds that universities are not realising the full potential of campus redevelopment to further Mission 1: teach, 2: research and 3: engage priorities;
- Develops an institutional framework to help better understand the potential of campus redevelopment (Stream 1) to further partnership development, teaching and research outcomes and the commercialisation of knowledge to deliver economic, social and environmental outcomes; and
- Proposes the use of the sustainable development goals be used at the heart of the Triple Helix model around which the helixes revolve to further the adoption of sustainable innovation.

#### 8.3 Further research

This thesis and the associated body of work has made a significant contribution to the theoretical body of knowledge on low carbon, or sustainable, transition through innovation at the university-industry-government nexus.

The work has also shown that top-down structures have not, in the literature reviewed for this thesis, led to the development of partnerships for sustainable innovation. This is a fascinating insight as both government and universities are highly educated and heavily invested in policy development, teaching and research into all the issues lying at the heart of sustainable development. Further research to understand this institutional issue is required.

The proposed metastructure (discussed in Chapter 7) and institutional framework developed in Section 5.3 help to address the largely elusive tension discussed above. The tension revolves around the formal, or informal, expression of the partnership between universities, industry and government and the actual partnerships that happen under the framework which leads to binomial partnerships even in a Triple Helix structure. The proposed metastructure starts to make sense of this incongruity but further research and exploration will be required.

Even so the apparent inability of Stream 1 and 2 partnerships to drive sustainable innovation needs to be better understood so that the metastructure can be further developed to help ensure local delivery of partnership development remains consistent with global initiatives. The metastructure is a start of this process but further work is needed.

There is a need for further applied and translational research to test the veracity of the theoretical developments proposed. This thesis has proposed an institutional framework for such applied research (particularly through Chapter 5 and Publication 5) which would help understand the potential to further sustainable innovation through campus redevelopment.

The proposal to use the sustainable development goals at the heart of the Triple Helix framework could be tested, possibly as part of the CSIRO's Responsible Innovation Future Science Platform to better understand how to deliver economically, socially and environmentally responsible innovation.

Publication 3 started – albeit unsuccessfully – to investigate the issue of decision making. It sought to apply Kahneman's [69] work on the psychology of human decision making to an institutional setting. This thesis has shown an institutional weakness that needs addressing. There is a rich seam of work that applies ecological concepts to the economy it would be interesting to see if the principles of human decision making can also be applied in an institutional setting. The early foray presented in Publication 3 suggests it might be fruitful work, but much further work is required.

The findings in this thesis would benefit from industry feedback into the utility of both the model developed in Chapter 5 and the metastructure proposed in Chapter 7 to help enable innovation at the university-industry-government nexus.

# **Exegesis References**

- 1. Drucker, P.F., Landmarks of tomorrow: A report on the new. 2011: Transaction Publishers.
- 2. Boeri, A., et al., *Resilient communities. Social infrastructures for sustainable growth of urban areas. A case study.* International journal of sustainable development and planning, 2017. **12**(2): p. 227-237.
- 3. Cooke, P., Regionally asymmetric knowledge capabilities and open innovation: Exploring 'Globalisation 2'—A new model of industry organisation. Research Policy, 2005. 34(8): p. 1128-1149.
- 4. Porter, M.E., *The competitive advantage of nations*. Competitive Intelligence Review, 1990. **1**(1): p. 14-14.
- Department of Prime Minister and Cabinet, National Innovation and Science Agenda, 2015. Commonwealth of Australia, Canberra.
- 6. OECD, The new economy beyond the hype. 2001, OECD, Paris.
- 7. Mitchell, G.A., *Research parks: Instrument, or harbinger of a new university paradigm?* Interchange, 1992. **23**(1-2): p. 99-104.
- 8. Etzkowitz, H., *Anatomy of the entrepreneurial university*. Social Science Information, 2013. **52**(3): p. 486-511.
- 9. United Nations, *Transforming our world*. 2016, United Nations: New York.
- 10. UNFCCC, *The Paris Agreement*. 2015, UNFCCC Secretariat: Bonn, Germany.
- 11. Flening, E., 30 Years After the Bayh-Dole Act: Rethinking the Australian Research Commercialisation Experience. 2010.
- Burbridge, M. and G.M. Morrison, A Systematic Literature Review of Partnership Development at the University-Industry-Government Nexus. Sustainability, 2021.
   13(24): p. 13780.
- 13. Etzkowitz, H., *The new visible hand: an assisted linear model of science and innovation policy*. Science and Public Policy, 2006. **33**(5): p. 310-320.
- 14. Hossain, M., S. Leminen, and M. Westerlund, *A systematic review of living lab literature.* Journal of Cleaner Production, 2019. **213**: p. 976-988.
- 15. Ayres, R., *Technological transformations and long waves. Part I.* Technological Forecasting & Social Change, 1990. **37**(1): p. 1-37.
- 16. von Hippel, E., *Democratizing Innovation*. 2005, The MIT Press.
- 17. Chesbrough, H. and M. Appleyard, *Open Innovation and Strategy.* California Management Review, 2007. **50**(1): p. 57-76.
- 18. Dell'Era, C. and P. Landoni, *Living Lab: A Methodology between User-Centred Design and Participatory Design.* Creativity & Innovation Management, 2014. **23**(2): p. 137-154.

- 19. Gibbons, M., The new production of knowledge : the dynamics of science and research in contemporary societies / Michael Gibbons et al. 1994, London: London: SAGE.
- 20. Etzkowitz, H. and L. Leydesdorff, *The dynamics of innovation: from National Systems and "Mode 2" to a Triple Helix of university–industry–government relations*. Research Policy, 2000. **29**(2): p. 109-123.
- 21. Carayannis, E.G. and D.F.J. Campbell, '*Mode 3' and 'Quadruple Helix': toward a 21st century fractal innovation ecosystem*. Int. J. Technol. Manag., 2009. **46**: p. 201-234.
- 22. Lundvall, B.A., *Why study national systems and national styles of innovation?* Technology Analysis & Strategic Management, 1998. **10**(4): p. 407-421.
- 23. Nelson, R.R., National innovation systems : a comparative analysis / edited by Richard R. Nelson. 1993, New York : Oxford University Press.
- 24. Lundvall, B. and B. Johnson, *The Learning Economy*. Journal of industry studies, 1994. **1**(2): p. 23-42.
- 25. Bush, V., Science the endless frontier. Washington: United States Government Printing Office, 1945. 184 p. Science Education, 1945. **29**(4): p. 218-218.
- 26. OECD, The Knowledge Based Economy, 1996. OECD, Paris
- 27. Etzkowitz, H. and L. Leydesdorff, *The Triple Helix -University-Industry-Government Relations: A Laboratory for Knowledge Based Economic Development*. EASST Review, 1995. **14**(1): p. 14-19.
- Etzkowitz, H., Webster, A., Gebhardt, C. and B. Terra. *The future of the university and the university of the future: evolution of ivory tower to entrepreneurial paradigm*. Research Policy, 2000. **29**(2): p. 313-330.
- 29. Fukuyama, F., *Social capital, civil society and development*. Third world quarterly, 2001. **22**(1): p. 7-20.
- 30. Jensen, M.B., Johnson, B., Lorenz, E. and Lundvall, B., *Forms of knowledge and modes of innovation*. Research policy, 2007. **36**(5): p. 680-693.
- 31. Carayannis, E.G., Mode 3 Knowledge Production in Quadruple Helix Innovation Systems : 21st-Century Democracy, Innovation, and Entrepreneurship for Development / by Elias G. Carayannis, David F.J. Campbell, ed. D.F.J. Campbell and SpringerLink. 2012, New York, NY: New York, NY : Springer New York.
- 32. Carayannis, E., Barth, T., and Campbell, D., *The Quintuple Helix innovation model: global warming as a challenge and driver for innovation.* Journal of innovation and entrepreneurship, 2012. **1**(1): p. 1-12.
- 33. Barcellos-Paula, L., De la Vega, I. and A. Gil-Lafuente, *The quintuple helix of innovation model and the sdgs: Latin-American countries' case and its forgotten effects*. Mathematics, 2021. **9**(4): p. 1-24.
- 34. Etzkowitz, H., *The new visible hand: An assisted linear model of science and innovation policy*. Sci Public Policy, 2006. **33**(5): p. 310-320.

- 35. Shinn, T., *The Triple Helix and New Production of Knowledge: Prepackaged Thinking on Science and Technology*. Social Studies of Science, 2002. **32**(4): p. 599-614.
- 36. Prainsack, B., *Review of Mode 3 Knowledge Production in Quadruple Helix Innovation Systems*. Minerva, 2012. **50**(1): p. 139-142.
- 37. European Union, *Responsible Research and Innovation Europe's ability to respond to societal challenges*, Innovation. 2014, European Union: Brussels.
- 38. Leydesdorff, L. and H. Etzkowitz, *Emergence of a Triple Helix of university—industry government relations.* Sci Public Policy, 1996. **23**(5): p. 279-286.
- 39. Etzkowitz, H., *Research groups as 'quasi-firms': the invention of the entrepreneurial university*. Research Policy, 2003. **32**(1): p. 109-121.
- Etzkowitz, H., Special introduction: Convergence of science parks, centres and clusters: From creative destruction to creative reconstruction in a triple helix regime. International Journal of Technoentrepreneurship, 2011. 2(3-4): p. 210-226.
- 41. Etzkowitz and Zhou, *Innovation incommensurability and the science park*. R and D Management, 2017.
- 42. Cai, Y. and H. Etzkowitz, *Theorizing the Triple Helix model: Past, present, and future.* Triple Helix Journal, 2020: p. 1-38.
- 43. Schoonmaker, M.G. and E.G. Carayannis, *Mode 3: A Proposed Classification Scheme for the Knowledge Economy and Society.* Journal of the Knowledge Economy, 2013.
  4(4): p. 556-577.
- 44. Ståhlbröst, A. and M. Holst, *Reflecting on Actions in Living Lab Research.* Technology innovation management review, 2017. **7**(2): p. 27-34.
- da Silva, L., Villalva, M., de Almeida, M., Brittes, J., Yasuoka, J., Cypriano, J., Dotta, D., Pereira, J., Salles, M., Archilli, G. and J. Campos, *Sustainable Campus Model at the University of Campinas—Brazil: An Integrated Living Lab for Renewable Generation, Electric Mobility, Energy Efficiency,* Monitoring and Energy Demand Management. 2018, Cham: Springer International Publishing: Cham. p. 457-472.
- 46. Seppä, M., From Business Administration to Business Creation: The Case of the Kalevala Global Business Creation School. Technology Innovation Management Review, 2012(June 2012: Global Business Creation): p. 6-11.
- 47. OECD, Collaboration on innovation, in OECD Science, Technology and Industry Scoreboard 2015: Innovation for growth and society, 2015, OECD Publishing: Paris.
- 48. OECD, Science and Technology Outlook. 2014, OECD: Paris.
- 49. Department of Education, *University research commercialisation*. 2021, Commonwealth Government: Canberra.
- 50. Office of the Chief Economist, *Australian Innovation System Monitor*. 2021, Department of Industry, Science, Energy and Resources: Canberra.
- 51. Univerities Australia, *Clever collaborations: the strong business case for partnering with universities*. 2018, Universities Australia: Canberra.

- 52. Global Innovation Index, *Global Innovation Index 2020*. 2020, Cornell University, INSEAD and WIPO: New York.
- 53. Innovation and Science Australia, Performance Review of the Australian Innovation, Science and Research System. 2016: Canberra.
- 54. Innovation and Science Australia, *Australia 2030: prosperity through innovation*. 2017, Commonwealth of Australia: Canberra.
- 55. Moher, D., et al., *Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement*. Journal of Clinical Epidemiology, 2009. **62**(10): p. 1006-1012.
- 56. Massey, O., *A proposed model for the analysis and interpretation of focus groups in evaluation research.* Evaluation and Program Planning, 2011. **34**(1): p. 21-28.
- 57. Denscombe, M., *Communities of Practice: A Research Paradigm for the Mixed Methods Approach.* Journal of Mixed Methods Research, 2008. **2**(3): p. 270-283.
- 58. Bazeley, P and P Brindle, *Mixed methods research*, 2015 SAGE Publications Ltd. London
- 59. Saldaña, J., *Fundamentals of qualitative research / Johnny Saldaña*, ed. C. Ebooks. 2011, New York: Oxford University Press.
- 60. Creamer, E.G., *Striving for Methodological Integrity in Mixed Methods Research: The Difference Between Mixed Methods and Mixed-Up Methods*. Journal of Engineering Education, 2018. **107**(4): p. 526-530.
- 61. Sandelowski, M., et al., *Mapping the Mixed Methods–Mixed Research Synthesis Terrain.* J Mix Methods Res, 2012. **6**(4): p. 317-331.
- 62. Petticrew, M. and H. Roberts, *Systematic reviews in the social science: a practical guide*, 2005. Malden, MA: Blackwell Pub.
- 63. Brien, A., The systematic literature review method: trials and tribulations of electronic database searching at doctoral level in Aoife M. O'Brien & Conor Mc Guckin, ed. C.a. Mc Guckin. 2016: London, SAGE.
- 64. Liberati, A., et al., *The PRISMA statement for reporting systematic reviews and metaanalyses of studies that evaluate healthcare interventions: explanation and elaboration*. BMJ, 2009. **339**(2009).
- 65. Mills, A.J., G. Durepos, and E. Wiebe, Encyclopedia of case study research / edited by Albert J. Mills, Gabrielle Durepos, Elden Wiebe. 2010, Los Angeles, [Calif.]
- 66. Gerring, J.a., *Case study research: principles and practices / John Gerring*. Second edition.. ed. 2017: Cambridge, England, United Kingdom: Cambridge University Press.
- 67. Swanborn, P., *Case study research: what, why and how? / Peter Swanborn*. 1st edition.. ed. 2010: Los Angeles, CA: SAGE Publications, Inc.
- Seawright, J. and J. Gerring, *Case Selection Techniques in Case Study Research: A Menu of Qualitative and Quantitative Options*. Political Research Quarterly, 2008.
   61(2): p. 294-308.

- 69. Kahneman, D., *Thinking fast and slow*. 2011, London: Penguin.
- 70. Hawken, P., *The ecology of commerce : a declaration of sustainability /* Paul Hawken. 1st ed.. ed. 1993, New York, NY : HarperBusiness.
- 71. Cook, J., Nuccitelli, D., Green, S., Richardson, M., Winkler, B., Painting, R., Way, R., Jacobs, P.and A. Skuce, *Quantifying the consensus on anthropogenic global warming in the scientific literature*. Environ. Res. Lett, 2013. **8**(2): p. 24024-7.
- 72. De Silva, M. and M. Wright, *Entrepreneurial co-creation: societal impact through open innovation.* R & D Management, 2019. **49**(3): p. 318-342.
- 73. Steen, K. and E. van Bueren, *The Defining Characteristics of Urban Living Labs*. Technology Innovation Management Review, 2017. **7**(7): p. 21-33.
- 74. Schaffers, H. and P. Turkama, *Living Labs for Cross-Border Systemic Innovation*. Technology Innovation Management Review, 2012. **2**(9): p. 25-30.
- 75. Ayvari, A. and A. Jyrama, *Rethinking value proposition tools for living labs*. Journal of Service Theory and Practice, 2017. **27**(5): p. 1024-1039.
- 76. Gasco, M., *Living labs: Implementing open innovation in the public sector.* Government Information Quarterly, 2017. **34**(1): p. 90-98.
- 77. Lukkari, O., J. Mustonen, and T. Tuikka. *City of Oulu as an innovative service platform.* in 2008 IEEE International Technology Management Conference, ICE 2008. 2016.
- Burbridge, M., Morrison, G. M., van Rijn, M., Silvester, S., Keyson, D. V., Virdee, L., Baedeker, C. and C. Liedtke, *Business Models for Sustainability in Living Labs*. 2016, Cham: Springer International Publishing: p. 391-403.
- 79. Aportela-Rodriguez, I.M. and A.R. Pacios, *University Libraries and Science and Technology Parks: Reasons for Collaboration*. Libri (København), 2017. **67**(3): p. 235-244.
- Fernández-Esquinas, M., C. Merchán-Hernández and O. Valmaseda-Andía, How effective are interface organizations in the promotion of university-industry links? Evidence from a regional innovation system. European Journal of Innovation Management, 2016. 19(3): p. 424-442.
- Bakouros, Y. and E. Samara, Academic Liaison Offices vs. technology transfer units: Could they form a new joint mechanism towards the exploration of Academic/Research results? International journal of innovation science, 2010. 2(4): p. 145-157.
- 82. McAdam, M. and R. McAdam, *The networked incubator: the role and operation of entrepreneurial networking with the university science park incubator (USI).* International Journal of Entrepreneurship and Innovation, 2006. **7**(2): p. 87-98.
- 83. Nielsen, C., *Getting value for money from your science park.* Public Money & Management, 2016. **36**(7): p. 539.
- 84. Minguillo, D. and M. Thelwall, *Research excellence and university–industry collaboration in UK science parks.* Research Evaluation, 2015. **24**(2): p. 181-196.

- Steinthorsson, R., Hilmarsson, E., and H. Janusson, *Towards openness and inclusiveness: The evolution of a science park.* Industry and Higher Education, 2017.
   31(6): p. 388-398.
- 86. Vedovello, C., *Science parks and university- Industry interaction: Geographical proximity between the agents as a driving force*. Technovation, 1997. **17**(9): p. 491-502.
- 87. Motohashi, K., *The role of the science park in innovation performance of start-up firms: an empirical analysis of Tsinghua Science Park in Beijing*. Asia Pacific Business Review, 2013. **19**(4): p. 578-599.
- 88. van Oostrom, M., J.A. Pedraza-Rodríguez, and M. Fernández-Esquinas, Does the Location in a Science and Technology Park Influence University - Industry Relationships?: Evidence From a Peripheral Region. International Journal of Knowledge Management, 2019. 15(3): p. 66-82.
- 89. Chesbrough, H., *Business Model Innovation: Opportunities and Barriers.* Long Range Planning, 2010. **43**(2–3): p. 354-363.
- Baccarne, B., Mechant, P., Schuurma, D., De Marez, L. and P. Colpaert, Urban Sociotechnical Innovations with and by Citizens. Interdisciplinary Studies Journal, 2014.
   3(4): p. 143.
- 91. Hardin, G., *The Tragedy of the Commons*. Science, 1968. **162**(3859): p. 1243-1248.
- 92. OECD, *Meausing Innovation a new perspective*. 2010. OECD Publishing, Paris.
- 93. OECD, OECD Science, Technology and Industry Outlook 2014. 2015. OECD Publishing, Paris.
- 94. Gahan, P., Adamovic, M., Bevitt, A., Harley, B., Healy, J., Olsen, J.E., Theilacker, M., *Leaership at work: do Australian leaders have what it takes?* 2016, Centre for Workplace Leadership, University of Melbourne, Australia.
- 95. Grayson, M. and S. Pincock, *Nature Index 2015 Collaborations*. Nature, 2015. **527**(7577): p. S49-S49.
- 96. Meadows, D.H., The Limits to growth : a report for the Club of Rome's project on the predicament of mankind / [by] Donella H. Meadows [et al.]. 1972, Universe Books, New York.
- 97. Boulding, K.E., The economics of the coming spaceship earth. New York, 1966.
- 98. Daly, H.E., Ecological economics and the ecology of economics : essays in criticism / Herman E. Daly. 1999, Elgar, Cheltenham
- 99. Curtin University, Creating the City of Innovation a vision for Greater Curtin, Curtin University, 2013: Bentley.
- 100. Curtin University, *Curtin Master Plan the vision*. Curtin University, 2013: Bentley.
- 101. Curtin University, *Curtin Master Plan part A*. Curtin University, 2013: Bentley.
- 102. Curtin University, *Curtin Master Plan part B.* Curtin University, 2013: Bentley.
- 103. Curtin University, *Curtin Master Plan part C.* Curtin University, 2013: Bentley.

- 104. Day, J., *Direction plan lays framework for 2031 and beyond*, Department of Planning, 2010, Government of Western Australia: Perth.
- Lacey, J., R. Coates, and M. Herington, Open science for responsible innovation in Australia: understanding the expectations and priorities of scientists and researchers. Journal of Responsible Innovation, 2020. 7(3): p. 427-449.
- 106. European Commission, *Options for Strengthening Responsible Research and Innovation*, D.R.a. Innovation, Editor. 2013, European Union, Luxembourg.
- 107. Ådahl, M., *Commercial consortia*, in Living Labs: Design and Assessment of Sustainable Living. 2016. p. 385-390.
- 108. Artto, K., et al., *The Cuckoo's Nest Approach for Co-Creating Business Ecosystems in Smart Cities.* Technology Innovation Management Review, 2016. **6**(12): p. 26-37.
- Lundström, A. and C. Zhou, Promoting innovation based on social sciences and technologies: the prospect of a social innovation park. Innovation, 2011. 24(1-2): p. 133-149.
- 110. Pisoni, A., Michelini, L. and G. Martignoni, *Frugal approach to innovation: State of the art and future perspectives.* Journal of Cleaner Production, 2018. **171**: p. 107-126.
- 111. Bracco, S., Delfino, F., Laiolo, P.and A. Morini, *Planning & open-air demonstrating smart city sustainable districts.* Sustainability, 2018. **10**(12): p. 4636.
- 112. Ventura, R., Quero, M. and M. Díaz-Méndez, *The role of institutions in achieving radical innovation*. Marketing Intelligence & Planning, 2019. **38**(3): p. 310-324.
- 113. Bergquist, D., Hempel, C. and J. Lööf Green, Bridging the Gap between Theory and Design: A Proposal for Regenerative Campus Development at the Swedish University of Agricultural Sciences. International Journal of Sustainability in Higher Education, 2019. 20(3): p. 548-567.
- 114. Drucker, P.F., *Organizations*. Harvard Business Review, 1992. **20**(7): p. 281-293.
- 115. Powell, W.W. and K. Snellman, *The Knowledge Economy*. Annual Review of Sociology, 2004. **30**(1): p. 199-220.
- 116. Shapiro, C. and H.R. Varian, *Information Rules: A Strategic Guide to the Network Economy*. 1998: Harvard Business Review Press.
- 117. Romer, P.M., *The Origins of Endogenous Growth*. The Journal of Economic Perspectives, 1994. **8**(1): p. 3-22.
- 118. Brynjolfsson, E., *The productivity paradox of information technology*. Commun. ACM, 1993. **36**(12): p. 66-77.
- 119. Solow, R.M., *A Contribution to the Theory of Economic Growth.* The Quarterly Journal of Economics, 1956. **70**(1): p. 65-94.
- 120. Levine, M. and A.G. Levine, *Who Said the Government Can't Do Anything Right? The World War II GI Bill, the Growth of Science, and American Prosperity*. Am J Orthopsychiatry, 2011. **81**(2): p. 149-156.

- 121. Drucker, P., *The rise of the knowledge society*. Wilson Quarterly, 1993. **70**(2).
- 122. Coghlan, D. and M. Brydon-Miller, *The SAGE encyclopedia of action research / general editors, David Coghlan, Mary Brydon-Miller*. Encyclopedia of Action Research. 2014: Los Angeles : SAGE.
- 123. Burbridge, M., *If Living Labs are the Answer What's the Question? A Review of the Literature.* Procedia Engineering, 2017. **180**: p. 1725-1732.
- 124. Hardeman, S. Frenken, K., Nomaler, Z. and A. Walter, *Characterizing and comparing innovation systems by different 'modes' of knowledge production: a proximity approach*. Science & Public Policy, 2015. **42**(4): p. 530-548.
- 125. Castells, M., The rise of the network society. Vol. 1. 2009: Wiley-Blackwell.
- 126. Carayannis, E., Mode 3, in Encyclopedia of Creativity, Invention, Innovation and Entrepreneurship, E.G. Carayannis, Editor. 2013, Springer, New York. p. 40-40.
- 127. Carayannis, E.G., D.F.J. Campbell, and S.S. Rehman, *Mode 3 knowledge production: systems and systems theory, clusters and networks.* Journal of Innovation and Entrepreneurship, 2016. **5**(1): p. 1-24.
- 128. Baccarne, B., Logghe, S., Schuurman, D, and L. De Marez, *Governing Quintuple Helix Innovation: Urban Living Labs and Socio-Ecological Entrepreneurship.* Technology Innovation Management Review, 2016. 6(3): p. 22-30.
- 129. Senate Economics Committee, *Australian Innovation System*. 2015, Senate Printing Unit: Parliament House, Canberra.
- 130. Knowledge Commercialisation Australia, *Survey of commercial outcomes from public research (SCOPR)*. 2020, Knowledge Commercialisation Australia: Canberra.
- 131. Office of the Chief Economist, *Australian Innovation System Report*, Department of Industry, Editor. 2017, Commonwealth Government: Canberra.
- 132. IPCC. Summary for Policymakers. In: Climate Change 2013: The Physical Science Basis. Contribution of Working Group 1 to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. 2013. London and New York: Cambridge University Press.
- 133. IPCC. Fifth Assessment report authors and reviewers. 2014 [cited 1 March 2017];
   Available from: http://www.ipcc.ch/pdf/ar5/ar5_authors_review_editors_updated.pdf.
- 134. IPCC, IPCC 2021: Summary for policy makers. Climate Change 2021: The Physical Science Basis. Contribution of Working Group 1 to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, ed. V. MassonDelmotte, P. Zhai, A. Pirani, S.L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M.I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J.B.R. Matthews, T.K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, and B. Zhou (eds.). 2021: Cambridge University Press.
- 135. IPCC, IPCC, 2014: Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, ed. R.K.P.a.L.A.M. (eds.). 2014, Geneva, Switzerland: IPCC.

- 136. Jones, D., Braganza, K. and Tobin, S, Weather Bureau says hottest, driest year on record led to extreme bushfire season. The Conversation, 2020.
- 137. Deb, P., Moradkhani, H., Abbaszadeh, P., Kiem, A., Engström, J., Keellings, D., and A. Sharma, Causes of the Widespread 2019–2020 Australian Bushfire Season. Earth's Future, 2020. 8(11)
- 138. Collins, L., Bradstock, R., Clarke, H., Clarke, M., Nolan, R., and T. Penman, *The* 2019/2020 mega-fires exposed Australian ecosystems to an unprecedented extent of high-severity fire. Environmental research letters, 2021. **16**(4).
- 139. Williams, J., *Exploring the onset of high-impact mega-fires through a forest land management prism.* Forest Ecology and Management, 2013. **294**: p. 4-10.
- 140. Ward, M., Tulloch, A., Radford, J., Williams, B., Reside, A,.Macdonald, S., Mayfield, H., Maron, M., Possingham, H., Vine, S *et al.*, *Impact of 2019–2020 mega-fires on Australian fauna habitat*. Nat Ecol Evol, 2020. 4(10): p. 1321-1326.
- 141. Crutzen, P.J., *The "anthropocene"*, in Earth System Science in the Anthropocene. 2006, Springer. p. 13-18.
- 142. Burbridge, M., *On being economic with the environment*, Environmental Change Institute. 1995, University of Oxford: Oxford.
- Waters, C. Zalasiewicz, J., Summerhayes, C., Barnosky, A., Poirier, C., Gałuszka, A., Cearreta, A., Edgeworth, M., Ellis, E. amd M. Ellis et al., *The Anthropocene is functionally and stratigraphically distinct from the Holocene*. Science, 2016. 351(6269)
- 144. Brundtland, G.H. and M. Khalid, *Our common future*. 1987: Oxford University Press, Oxford.
- 145. Pearce, D.W., Blueprint for a green economy : a report / by David Pearce, Anil Markandya, Edward B. Barbier ... for the UK Department of the Environment, ed. A. Markandya, E. Barbier, and E. Great Britain. Department of the. 1989, London: London : Earthscan Publications.
- 146. Burbridge, M. *Everyone loves sustainable development, but does anyone know what it is?* The Conversation, 2012.
- 147. United Nations, Transforming our world: the 2030 agenda for sustainable development. 2015, United Nations: New York.
- 148. Sachs, J. From Millennium Development Goals to Sustainable Development Goals. Lancet, 2012. **379**(9832): p. 2206-2211.
- 149. Department of Foreign Affairs and Trade, *Report on the implementation of the sustainable development goals*. 2018, Commonwealth of Australia: Canberra.
- 150. United Nations, *Shaping the future we want*. 2015. United Nations, New York
- 151. Adams, R., Bessant, J. and R. Phelps, *Innovation management measurement: A review*. International Journal of Management Reviews : IJMR, 2006. **8**(1): p. 21-47.

- 152. Baregheh, A., Rowley, J. and S. Sambrook, *Towards a multidisciplinary definition of innovation*. Management Decision, 2009. **47**(8): p. 1323-1339.
- 153. OECD, The Measurement of Scientific and Technological Activities: Guidelines for Collecting and Interpreting Innovation Data: Oslo Manual, W.P.o.N.E.o.S.a.T. Indicators, 2005, OECD: Paris.
- 154. Drucker, P.F., The discipline of innovation. Harvard Business Review, 2002. 80
- 155. Blok, V. and P. Lemmens, The Emerging Concept of Responsible Innovation. Three Reasons Why It Is Questionable and Calls for a Radical Transformation of the Concept of Innovation. 2015, Cham: Springer International Publishing: Cham. p. 19-35.
- 156. European Commission, Horizon 2020 The Framework Programme for Research and Innovation. 2011, European Commission: Brussels.
- 157. Stutcliffe, H., *A report on Responsible Research and Innovation*. 2011, MATTER: Brussels.
- 158. Rosenlund, J., E. Rosell, and W. Hogland, *Overcoming the triple helix boundaries in an environmental research collaboration.* Science & Public Policy, 2017. **44**(2): p. 153-162.
- 159. Finkel, A., Powering up citations. Nature (London), 2014. 511(7510): p. S77.
- 160. Caulfield, T., Harmon, S. and Y. Joly, *Open science versus commercialization: A modern research conflict?* Genome Medicine, 2012. **4**(2): p. 17-17.
- 161. Tait, J., *From responsible research to responsible innovation: challenges in implementation*. Engineering Biology, 2017. **1**(1): p. 7-11.
- 162. CSIRO, Latest from the responsible innovation Future Science Program. 2021, CSIRO: Canberra.
- 163. Smith, A., An inquiry into the nature and causes of the wealth of nations / by Adam Smith. 2009, Champaign
- 164. Gunasekara, C., *Reframing the Role of Universities in the Development of Regional Innovation Systems*. The Journal of Technology Transfer, 2006. **31**(1): p. 101-113.
- 165. Porter, M., The five competitive forces that shape strategy.
- 166. Fukuyama, F., *The great disruption*. 2017: Profile Books.
- 167. Fukuyama, F., *Trust: social virtues and the creation of prosperity / Francis Fukuyama*. Social virtues and the creation of prosperity. 1995, London: Hamilton.
- 168. Pino, R.M. and A.M. Ortega, *Regional innovation systems: Systematic literature review and recommendations for future research.* Cogent Business & Management, 2018. **5**(1): p. 1463606.
- 169. Mathews, J.T., *Power Shift*. Foreign Affairs, 1997. **76**: p. 50.
- 170. Galvao, A., et al., *Triple helix and its evolution: a systematic literature review.* Journal of Science and Technology Policy Management, 2019. **10**(3): p. 812-833.
- 171. Cai, Y., *Implementing the Triple Helix model in a non-Western context: an institutional logics perspective.* Triple Helix, 2014. **1**(1): p. 1-20.

- 172. Etkowitz, H., *The Triple Helix: University-Industry-Government Innovation in Action*. 2008: Taylor and Francis.
- 173. Zhou, C., *Emergence of the entrepreneurial university in evolution of the triple helix.* Journal of Technology Management in China, 2008. **3**(1): p. 109-126.
- 174. Sober, E., *The Principle of Parsimony*. The British journal for the philosophy of science, 1981. **32**(2): p. 145-156.
- Almirall, E., Lee, M. and J. Wareham, *Mapping Living Labs in the Landscape of Innovation Methodologies*. Technology Innovation Management Review, 2012. 2(9): p. 12-18.
- 176. Hughes, H., Wolf, R. and M. Foth, *Informed digital learning through social living labs as participatory methodology: The case of Food Rescue Townsville*. Information and Learning Science, 2017. **118**(9-10): p. 518-534.
- 177. Antonopoulos, C. Papadakis, V., Stylios, C., Efstathiou, M. and P. Groumpos, *Mainstreaming innovation policy in less favoured regions: the case of Patras Science Park, Greece.* Science and Public Policy, 2009. **36**(7): p. 511-521.
- 178. Hladchenko, M. and R. Pinheiro, *Implementing the Triple Helix model: Means-ends decoupling at the state level?* Minerva, 2019. **57**(1): p. 1-22.
- 179. Aslani, A., H. Eftekhari, and M. Didari, Comparative Analysis of the Science and Technology Parks of the US Universities and a Selected Developing Country. Proceedings of the 11th International Conference on Innovation and Management, Vols I and Ii, ed. L. Yang, et al. 2014. 1-7.
- 180. Jongwanich, J., A. Kohpaiboon, and C.-H. Yang, *Science park, triple helix, and regional innovative capacity: province-level evidence from China.* Journal of the Asia Pacific Economy, 2014. **19**(2): p. 333-352.
- 181. Bartelt, V. Urbaczewski, A., Mueller, A., and S. Sarker, *Enabling collaboration and innovation in Denver's smart city through a living lab: a social capital perspective*. European Journal of Information Systems, 2020. **29**(4): p. 369-387.
- 182. Berker, T. and R. Woods, *Identifying and addressing reverse salients in infrastructural change. The case of a small zero emission campus in Southern Norway*. International Journal of Sustainability in Higher Education, 2020. **21**(7): p. 1625-1640.
- 183. Rutter, J., *Policy as if people matter*, M. Burbridge, 2005: unpublished personal communication.
- 184. Prno, J. and D.S. Slocombe, Exploring the origins of 'social license to operate'in the mining sector: Perspectives from governance and sustainability theories. Resources Policy, 2012. 37(3): p. 346-357.
- 185. Carayannis, E., Grigoroudis, E., Campbell, D, Meissner, D. and D. Stamati, *The* ecosystem as helix: an exploratory theory-building study of regional co-opetitive entrepreneurial ecosystems as Quadruple/Quintuple Helix Innovation Models. R & D Management, 2018. **48**(1): p. 148-162.

- Samuelson, P.A., *Economics / Paul A. Samuelson, William D. Nordhaus*. 19th ed..
   2010, Boston: Boston: McGraw-Hill Irwin.
- 187. Hardin, G., *Population, evolution and birth control*. 2nd Edition ed. 1964, San Francisco: WH Freeman and Co.
- 188. Barth, T.D., The Idea of a Green New Deal in a Quintuple Helix Model of Knowledge, Know-How and Innovation. International Journal of social Ecology and Sustainable Development, 2011. 2(1): p. 1-14.
- 189. Craig, E., *Hegelianism*, in *Concise Routledge Dictionary of Philosophy*, E. Craig, Editor.2002, Cambridge University Press: Cambridge.
- 190. De Oliveira Monteiro, S.P. and M.A.P. da Silva Duarte, *Quadruple Helix R&D Growth Models: A Panel Cointegration Analysis Applied to a Sample of OECD Countries, in The Quadruple Innovation Helix Nexus: A Smart Growth Model, Quantitative Empirical Validation and Operationalization for OECD Countries*, S.P. De Oliveira Monteiro and E.G. Carayannis, Editors. 2017, Palgrave Macmillan US: New York. p. 65-93.

*Every reasonable effort has been made to acknowledge the owners of copyright material. I would be pleased to hear from any copyright owner who has been omitted or incorrectly acknowledged.* 

# Appendix A Extended literature review

This Appendix provides an extended literature review for this thesis and for each publication (Appendix B.1 to B.5). It provides a detailed view of the existing literature within which this thesis resides. The extended review provides an exploration of the literature that relates to the rise of the knowledge economy, the changing role of knowledge to drive economic development; the importance of exogeneity to economic theories; changing modes of knowledge creation, the changing role of universities as creators of knowledge, the importance of collaboration and partnership as well as issues specific to the theory of partnership development at the university-industry-government nexus. Table A.1 provides an overview of the relevance of the literature to the overall exegesis discussion. A synthesis of the key issues is provided in Chapter 2.

#### A.1 The changing role of knowledge in the economy

Since the birth of the modern economic system knowledge has been a key factor in creating wealth and delivering economic growth. Economic systems are based on exploiting knowledge to create goods or services. There is no knowledge free economy. It was this exploitation of knowledge and then technological diffusion that led to five technological transformations (see Table A.2) that have driven economic growth and prosperity since at least the 18th century [15].

The point about the economy is that economic growth increasingly depends upon the exploitation of knowledge for the next wave of economic growth [97]. Such an economy has been referred to as post-industrial [15], post-Capitalist [114], learning economy [24] and knowledge economy [115]. In this thesis the term knowledge economy is used.

In a knowledge economy there is a greater reliance on knowledge than physical or natural resources to increase economic activity. A knowledge economy is defined as "the production and services based on knowledge intensive activities that contribute to an accelerated pace of technological and scientific advance as well as equally rapid obsolescence" [115].

The term knowledge economy covers a wide variety of activities, and the literature reveals three broad strands of research. The first is focused on the rise of

new science-based industries, in the early to mid-20th century where cycles of innovation were increasingly rapid and would herald a new product or service. The rise of services, and intangible goods and information was increasingly common [116]. This era saw the formalised development of a professional services industry [115]. Such industries are information rich but the role of knowledge in furthering economic is subject to much debate [117].

Section	Literature reviewed	Research question	Thesis narrative
Appendix A	Changing role of knowledge in the economy Changing role of universities A changing climate Innovation and sustainable development Economics of innovation: the rise of partnerships for innovation	How can universities, industry and government together drive low carbon innovation and transition?	<ul> <li>Knowledge is becoming an increasingly important driver of economic growth</li> <li>The process of innovation is evolving</li> <li>This has led to the creation of different partnership structures</li> <li>The focus of innovation is economic growth</li> </ul>
Section 2.2 and Appendix A	Partnerships for innovation – models of partnership at the UIG nexus	Which partnership models at the UIG nexus have been successful to deliver innovation?	<ul> <li>Increasingly commercial exploitation of knowledge depends upon a partnership between two or more entities</li> <li>The focus is the delivery of commercial success and economic growth</li> </ul>

Table A.1Relationship between the literature review, research questions and the<br/>exegesis discussion

Section	Literature reviewed	Research question	Thesis narrative
Section 2.3 and Appendix A	Partnerships for innovation – the creation of hybrid organisations	Which factors promote or limit partnership development for a low carbon future?	<ul> <li>Partnership structures are evolving from a top- down/ bottom-up approach</li> <li>Top-down approaches are an organisational response to the exploitation of knowledge</li> <li>Bottom-up structures are more likely to address economic, social and environmental issues</li> </ul>
Section 2.4 and Appendix A	Partnerships delivering economic, social and environmental outputs	Which partnership models at the UIG nexus have been successful to deliver low carbon and/or sustainable innovation?	<ul> <li>Partnerships that give voice to public interest are more likely to drive local low carbon outcomes</li> <li>However, there is little evidence of these partnerships delivering across the sustainable development goals</li> </ul>

The second stream of research surrounds the knowledge intensity of industries. The argument runs that knowledge intensive industries create new products and services increasingly rapidly and this gives rise to productivity gains and economic growth. However, there is a paradox as the productivity of particularly the service sector has stagnated whilst the investment in technology has increased by orders of magnitude [118, 119].

Date	Technological transformation	Technological diffusion
1770 - 1800	Move from a dependence on charcoal to coal	Development of transport systems and early mechanisation
1830 - 1850	Application of steam technology to textile and transportation	Diffusion of stationary and moving steam engines
1860 - 1900	Steel, internal combustion engine, electrification and communication. Expansion of the chemicals industry. Sharp increase in manufacturing productivity.	Exploitation of oil and coal to create electricity. Creates the potential for long distance communication and many new industries.
1930 - 1950	Synthetic materials and development of electronics. Emergence of consumer demand	Diffusion of technologies and mass production of electronic and 'white' goods.
1980 -	Convergence of computers and telecommunications	Post-industrial economy (see below)

 Table A.2
 Technological transformations since the 18th century

The third stream of research relates to the role of innovation within firms and is more managerial in nature. This stream of research started with Taylorism but, thanks to Drucker [114] focused on the fifth, or main factor of production – knowledge. He posited that the economy – certainly since the end of World War 2 was increasingly knowledge based. The importance of knowledge to society for economic development has been a policy aim of Government since the 1940s – an early example in the USA is the Bill of Rights which gave every returning serviceman financial support to attend university [120].

Drucker [114] identifies knowledge as being applied to knowledge, which he defines as management. But he also recognises the importance of applying knowledge systematically to define what knowledge is needed, what is feasible and what has to be done to make the knowledge effective. This he refers to as a management revolution, and this thesis is grounded in that revolution.

#### A.1.1 Knowledge as a management revolution

Before the advent of the internet Drucker [1, 114, 121] foresaw the need for educated people to be citizens of the world and foretold the conflict between holders of knowledge who wish to practice their knowledge and managers who see knowledge as a means to the ends of organisational performance. He sees them as representing different poles on a continuum rather than contradictions and he sets out how each need the other.

The intellectual, unless counterbalanced by the manager does his own thing; the manager's world becomes "stultifying and bureaucratic" without the offsetting influence of the intellectual. The relentless rise in the importance of knowledge to individuals, organisations and society gives rise to several tensions [114]:

- Communities need for stability, but organisations need to destabilise;
- An organisation's need for autonomy and society's stake in the common good; and
- The tensions between increasingly specialised specialists and the performance of the team.

He predicted that all of these will be central concerns for years to come. It is the bringing together of these issues that is the essence at the heart of this thesis. How can knowledge be created, used and applied to further the interests of humanity?

"The challenge that faces us now, and especially in the developed, free-market democracies such as the United States, is to make the pluralism of autonomous, knowledge-based organisations redound both to economic performance and to political and social cohesion". [114]

# A.1.2 Realigning the role of universities, industry and government

The importance of the knowledge economy for the next wave of economic growth has led to an increased focus by Government policy makers, universities as well as businesses on the development and economic exploitation of knowledge. This has led to an evolution in the role of the university in society and it is one that continues to evolve. Universities have existed for 800 years, and the current realignment is the beginning of only the third such change in their history (see Table A.3 which is developed from [7, 39, 122]) and it is associated directly with the changing role of knowledge in society and the economy [7].

Date	Principle influences on the university	Main function of university	Main outputs	Mode of knowledge production
1170 – 1330	Church	Education	Lawyers, priests and medical doctors	
1330 - 1830	Church and state	Education	Lawyers, priests, medical doctors and civil servants	
First academic revolution				
1830 - 1980	State	Education Research	Professions, Civil Servants, Enterprise Pure research Applied research	Mode 1 ⁴ research
Second academic revolution				
1980 -	State and enterprise	Education Research Engagement	Key skilled groups Pure research Applied research	Mode 2 & Mode 3 research

	Table A.3	The evolving state of the university
--	-----------	--------------------------------------

# A.1.3 Entrepreneurial universities

The changing role of the university is a response to the intensification of knowledge as a driver of economic growth. This has been recognised by government and has led to the emergence of entrepreneurial universities [8, 39] where the interaction between academia and industry was encouraged as well as an evolution in the form of research undertaken (see Table A.3).

Universities exist today to do three things: teach, research and engage (Publication 3, [123]). Universities continue to evolve in response to their external environment (deepening of the knowledge economy) and internal environment (commercialisation potential of research). America and Europe had different drivers for change. In America

⁴ There is debate about whether Mode 1 precedes Mode 2 knowledge production or not. It is not for this thesis to resolve the question. However, in the university context the institutionalisation of disciplines around 1800 will have led to Mode 1-type knowledge production. Prior to institutionalisation it is reasonable to assume that knowledge production was more in line with Mode 2.

it was a largely bottom-up transformation with universities understanding the potential value of their research; in Europe is has been a largely top-down process with universities being given policy incentives (through funding inducements or cuts) to commercialise research. As such an entrepreneurial, or corporate university functions primarily to promote economic development (Publication 1 [20]) and works towards the capitalisation of knowledge [39].

Increasingly with the emergence of massive open online courses (MOOCs) and online platforms an entrepreneurial university can now apply the ethos of entrepreneurism to all its learning activities (teaching, research and engagement) and not just research. Universities have had to develop new skills, processes and functions (see Table A.4 which is adapted from [93]) to accommodate the evolution to entrepreneurialism [47]. This comes in the form of intellectual property agreements, conflict of interest processes, technology transfer becoming a new function and a blurring of the lines between where universities end, and businesses start [39]. This thesis is placed at the boundary between university, industry, government and society.

### A.1.4 Changing modes of knowledge production

As universities have become more entrepreneurial so too has government, innovation processes and as a result methods of knowledge production.

Knowledge production has become increasingly distributed - a fact that was initially recognised in 1963 [124]. That process has not slowed but rather accelerated leading to a globalisation of knowledge production [125]. This globalisation has led to changing modes of knowledge production.

Aspects of learning	Impact
	Entrepreneurship development in teaching and learning
	Pathways for entrepreneurs
Teaching	Organisational capacity: people and incentives
Research	University/industry – external relationships for knowledge exchange
Engagement	International relationships (academic, business and government)
	Leadership and governance
	Impact realisation (KPIs)

Table A.4 A	Aspects of	learning in	npacted b	v transition to	o an entre	preneurial	universitv
				,			

Mode 1 knowledge production is the type of research that was conducted when universities started to undertake research. It is discipline based and produces theoretical knowledge [19]. Mode 1 research is still relevant today and is referred to as fundamental research [20].

Mode 2 knowledge production is transdisciplinary. In Mode 2 knowledge production there is a continuous interaction between theoretical and applied elements of research and practice so that one informs the other. The research is not fundamental in nature and is characterised by application or use in practice. Typically, more researchers are involved than in Mode 1 from a range of academic disciplines [19]. "Mode 2 creates a novel environment in which knowledge flows more easily across disciplinary boundaries, human resources are more mobile and the organisation of research more open and flexible" [19]. It is also distributed knowledge; distributed between partners but also geographically [124] and this spatial spread is also contributing to the globalisation of knowledge production [125].

Mode 3 knowledge production is a development from Mode 2. The key conceptual development is that it is a multi-level innovation ecosystem [126]. It goes wider than transdisciplinary as it includes various subsystems that when drawn together enable people, culture and technology to interact to innovate across scientific and technological disciplines and public/private sectors [31]. It emphasises a knowledge systems perspective [127]. It has been argued that Mode 3 combines mode 1 and mode 2 by undertaking basic research in an applied manner [127]. Mode 3 knowledge production has been referred to as open innovation democracy [128] or the democratisation of knowledge [16]. See Table A.5 for a summary.

Knowledge Production	Knowledge production elements	Innovation process
Mode 1	Discipline based; fundamental	Linear
Mode 2	Transdisciplinary; applied and distributed between partners	Non-linear
Mode 3	Transdisciplinary; applied; distributed and democratic	Multi- level ecosystem

	Table A.5	Modes of	knowledge	production
--	-----------	----------	-----------	------------

The change in the modes of knowledge production is significant for this thesis for two reasons. Firstly, it is easy to see how conceptually knowledge production can not only transcend disciplinary boundaries but also organisational boundaries. The focus is on the production and sharing of knowledge rather than who creates or owns the knowledge, the focus is on the use of knowledge. This is what von Hippel [16] refers to as the democratisation of knowledge.

The second flows from the first and it is the link to the nascent responsible research initiative (discussed further in this section) where the EU, USA and more recently Australia have started a process to engage society on the merits of research investment in order to reduce the risk of research moving society in a direction that is contrary to the wishes of broader society [106]. These initial moves towards 'Responsible Research and Innovation' reflects the further development a process between universities, industry and government where grant applications increasingly engage funding bodies in language that resonates with the policy, or research impact, or business drivers rather than the vernacular of the academic discipline. The shift to Mode 2 knowledge production was inevitable in a deepening knowledge economy where research funding is competitive and needs to have an economic (and increasingly social) return. Indeed, there is an emerging stream of work on what is referred to as *Globalisation 2* [3] which is looking to the engagement of society in globalisation in a transdisciplinary manner consistent with open science and innovation.

# A.2 Innovation and government policy

The importance of commercialising knowledge as a route to economic growth is increasingly recognised by international development organisations (for example [47]) as well as government (in Australia see for example [49] and [129]).

In the wake of the 2008 financial crisis innovation has become the focus of Government attention due to its potential to generate economic growth, new jobs [47] and to "seize" the next wave of economic prosperity [5]. To do this there is a need to create new forms of partnerships to exploit the potential of new knowledge [92] and vertical, or silo-based policies need to be complemented by a horizontal or whole of government approach to innovation [129] and the development of a more systematic ecosystem approach to innovation [47]. However, despite the emphasis on the

importance of innovation, the OECD gross expenditure on research and development in 2008 - 12 at 1.6% was half of the rate it was between  $2001 - 2008^5$ . China's expenditure on research and development doubled over the same timeframe [93].

#### A.3 Australian innovation policy and performance

Australia launched its National Innovation and Science Agenda in 2015 [5]. It presented a whole of government approach to innovation and stressed the importance of innovation to "seize the next economic wave" to keep Australia competitive, relevant and to maintain standards of living [5]. The *Agenda* presented four pillars that provided the framework for Australia's innovation policy: culture and capital; collaboration; talent and skills; government as an exemplar. It foreshadowed an active role for government in the policy space and the collaboration pillar of the policy reflects the importance of developing partnerships to effectively exploit knowledge.

*"There is an ever-present imperative to capture the commercial value of our research endeavour for our future wellbeing" Alan Finkel,* (in [130])

The Australian government recognised that the rate of collaboration between research and industry sectors in Australia was the lowest in the OECD [5, 93] in the bottom half of the OECD [131] and remains weak [50]. Australian industry is generally weak at collaborating for innovation, even though public research funding in universities at a level that is higher than the OECD average [93]. The collaboration for innovation that does happen is mostly domestic with customers or suppliers. Of those firms that do collaborate 25% do so with a firm within their ownership structure [50]. Collaboration with publicly funded researchers is limited standing at 2% of patent applications [50]. However, even though the percentage is low it is estimated to generate \$10.4bn in 2018 generating 30,000 full time jobs, although this is 30% lower than the United States or Israel [51].

This relative weakness at collaboration stands in contrast to Australian research output, of which 90% is at or above world standard [51] and with a ranking 23rd in the world in the WIPO Global Innovation Index [52] making Australia an innovation leader [50]. Australia performs well in knowledge creation but relatively poorly in translating

⁵ Note – this represents 1.6% of Gross Domestic Product; it does not mean that spending on R&D was halved, but rather the share of spending on R&D did not increase at the same rate as GDP

this knowledge into new products or other innovations [49]. This has been recognised by Innovation and Science Australia who in 2017 made improving the collaboration between universities and industry one of their 5 imperatives [54], which at the time of writing the Australian government is consulting upon [49].

# A.4 A changing climate

Anthropogenic greenhouse gas emissions (GHG) are rising and forcing the climate to change [132-135]. These changes are happening in every region and across the global climate system [134]. These changes will involve increasing heatwaves, more intense storms, changing rainfall patterns, increased (and increasingly frequent) flooding, permafrost thawing, loss of seasonal snow cover, melting of glaciers and ice caps. The impacts of increasing concentrations of greenhouse gases in the atmosphere are significant and long term (ranging from centuries to millennia) [134]. Carbon dioxide is the main driver of climate change [134] and there is a 'near linear' relationship between cumulative anthropogenic carbon dioxide and the global warming they are causing [134]. Concentrations of carbon dioxide have increased by 46% since the industrial revolution and is now at a level that has no equivalent in at least 800,000 years.

It is "unequivocal" [134] that human influence has warmed the atmosphere, ocean and land. This has led to heatwaves, bushfires, flooding and weather extremes. In Australia 2019 was the warmest (1.52°c warmer than previous warmest year) and driest year (11.7% drier than previous driest year) on record [136].

Further, in Australia the 2019/2020 bushfires are the largest on record burning over 10m hectares of land [137] and burnt an unprecedented area of temperate forest, woodland and shrubland across south-eastern Australia [138]. Such mega-fires are increasing in number and severity [139] and climate change making the conditions more conducive for such fires [138]. The cost of the fires is estimated to be \$100bn with 832 native vertebrate fauna impacted of which 21 were already listed with extinction and 70 species lost over 30% of their habitat [140].

#### A.4.1 Sustainable development - society's response to a changing world

Humankind's impact on our natural environment has long been recognised across a range of indicators from a rising population, to increased resource usage and changing of the landscape (see Table A.6 from McNeil in [141]). The current generation is not unique in its efforts to change nature to its will [142]; it is just that the scale of the impact is epoch making which has led to calls for the Anthropocene to be approved as a subdivision of geological time [143].

The impact of these changes on the economy, environment and society has long been debated. Malthus published his essay on the principle of population in 1798 concerning the exponential growth of the population outstripping the linear growth of food production. In 1972 the Club of Rome published "Limits to growth" [96] looking at the issue of exponential growth in a finite world. At the same time the United Nations hosted the United Nations Conference on the Human Environment in Stockholm, Sweden⁶ in 1972. The conference created institutional change (such as the creation of the UN Environment Program) but there was inaction over implementation of the agreed program. However, it did lead to a greater focus on how the economy, environment and society interact.

Impact	Increase factor 1890s to 1990s
Human population	4
World urban population	13
World economy	14
Industrial output	40
Energy use	16
Coal production	7
CO ₂ emissions	17
SO ₂ emissions	13
Water use	9
Marine fish catch	35
Cattle population	4
Irrigated area	5

Table A.6A partial record of the growth and impacts of human activities during the<br/>20th century

⁶ The conference was hosted in Sweden as it had suggested in 1968 that a conference should be hosted to focus on human interactions with the environment.

The issues raised were sufficiently serious to lead to the publication by the United Nations' World Commission on Environment and Development of "Our Common Future" [144] which in turn led to the UN hosting the World Summit of Sustainable Development in 1992, 2002 World Summit on Sustainable Development (Johannesburg) and the UN Conference on Sustainable Development in 2012.

Sustainable development is defined as "development that meets the needs of the current generation without compromising the opportunity of future generations to meet their own needs" [144]. The definition is widely accepted (although there are many variations [145]) but the real issue is not with the definition but with what does good look like [146]? This is an issue that has been resolved with the unanimous adoption in 2015 of the 2030 Agenda for Sustainable Development [147]. The 2030 Agenda for Sustainable Development goals (SDGs) (see Table A.7) with 169 associated targets for economic, social and environmental objectives [147].



# Table A.7 United Nations' Sustainable Development Goals

It therefore defines for the first time what good looks like and defines means of implementation to set humanity on a sustainable trajectory [148]. In Australia the sustainable development goals have been described as "a true global blueprint for a sustainable future for our planet, our communities, our families and our economies" and a "contemporary manifestation of the 'fair go'" [149].

The SDGs are intended to stimulate action - through a collaborative partnership – from 2015 – 2030 to promote the interests of people, planet, prosperity and peace through a revitalised Global Partnership for Sustainable Development [150].

#### A.5 Innovation and Responsible Research and Innovation

The literature on innovation is diverse and there are many definitions of innovation [151]. In developing a multidisciplinary definition of innovation Baregheh et al [152] lists 60 definitions of innovation and proposes an integrative definition of innovation: "Innovation is the multi-stage process whereby organisations transform ideas into new/improved products, service or processes, in order to advance, compete and differentiate themselves successfully in their marketplace" [152].

The OECD define innovation as "the implementation of a new or significantly improved product (good or service), or process, a new marketing method, or a new organisational method in business practices, workplace organisation or external relations". And the also OECD defines four discreet forms of innovation (see Table A.8 [153]).

But as Drucker [154] points out innovation is not limited to an organisational response, but it can be practiced by individuals as well as commercial, not for profit and public sector organisations. Irregardless, at the core of innovation are the concepts of improvement, newness and spread of ideas or technologies for some form of reward. In terms of reward much of the literature is predisposed towards economic reward (Publication 1; [12]), but there is an emerging stream of largely European research looking at 'responsible' innovation where reward can be a mixture of economic, social or environmental outcomes [155].

Innovation itself is a process that starts with an idea (for a new process, product, good or service) and ends with getting some form of reward for that idea. The route between idea and reward conceptually is linear, but increasingly it is a non-linear process (Publication 1, [12, 13, 31]).

### A.5.1 Responsible innovation

Responsible innovation is an emerging stream of research from Europe. It is an approach to research and innovation that looks not only at the research but also the

likely impact of the research and consequent innovation on society to foster a design that produces research and innovation that society is comfortable with [156]. The area is emerging and strategic in nature. It is currently focused on the <u>acceptability</u> of science and technology research and innovation to wider society. The research is at a high level looking largely at focused ethical issues and the social acceptability of the proposed research and innovation [157].

Responsible Research and Innovation is part of Europe's drive for the creation of a research and innovation policy to meet the needs of society by engaging society using inclusive participatory approaches (see Table A.9) [37]. It is the result of a continuation of the pursuit of "open science" in the European Union and as such it is progression from open access [105]. For the EU, Responsible Research and Innovation is about making sure that research and innovation proceeds in an inclusive manner and reflects upon societal needs, moral values [37] and has societal relevance [158]. It is therefore knowledge creation at the Mode 2 or Mode 3 interface.

Type of innovation	Description	
Product innovation	A good or service that is new or significantly improved. This includes significant improvements in technical specifications, components and materials, software in the product, user friendliness or other functional characteristics	
Process innovation	A new or significantly improved production or delivery method. This includes significant changes in techniques, equipment and/or software.	
Marketing innovation	A new marketing method involving significant changes in product design or packaging, product placement, product promotion or pricing.	
Organisational	A new organisational method in business practices, workplace organisation or external relations.	

Table A.8	OECD's 4 types	of innovation
-----------	----------------	---------------

Responsible research and innovation (RRI) goes beyond complying with ethics requirements and research codes of practice [105]. It takes researchers into a space where the research partnership needs to demonstrate and discuss with society a societal benefit (however defined) from the work [159]. As such the science needs to be contextuated [3] and the benefits debated. RRI is about developing individualised

processes to enhance the 'value, benefit and impact that can be created from science for society' [105].

The EU [37] foresees that RRI will ultimately strengthen the knowledge economy (see Table A.9). This move to open science includes the commercialisation of science [160] and the partnership required to commercialise the research [161]. Currently the preponderance of the emerging research is on 'upstream' management of research and innovation partnerships with less attention to 'downstream' implementation and governance [161]. This thesis is concerned with downstream governance issues.

# A.5.2 Responsible innovation and Australia

In Australia, Responsible Research and Innovation is known as *Responsible innovation* and CSIRO created a Responsible Innovation Future Science Platform in 2019. The platform's vision is to "ensure responsible science and technology is designed and delivered for the benefit of all Australians" [162]. The platforms' aims are to:

- Assess the potential risks, benefits and uncertainties of future science and technology; and
- Ensure socially responsible science and technology is delivered for all Australians.

Stream	Description
Choose Together	Co-creation of a response to the grand challenges facing humanity
Unlock the full potential	Ensure that all parts of society are engaged and included (in universities and participatory panels)
Creative learning and fresh ideas	Sharing of ideas and responses through the engagement of all societal actors
Do the right "think" and do it right	Change the view of ethics from being a constraint to helping to deliver high quality relevant research
Share results to advance	Make science knowledge as open as possible and actively share results
Design science with and for society	Development of governance structures to develop responsible research and innovation

Table A.9	European RRI - 6 streams of work
-----------	----------------------------------

Responsible Innovation is a Future Science Platform with the CSIRO. It is aligned with the European approach seeking to encourage public participation in the design of science. Similarly, it is focused on the future of science and technology in both basic and applied research – it seeks to assess the potential risks, benefits and uncertainties of future science and technology [162]. However, the premise behind *Responsible Innovation* in Australia is to make the research and innovation socially acceptable and deliver benefits to society [105]. It is important to note that this does not necessarily mean that the research or innovation will be sustainable or contribute to moving society towards the achievement of the sustainable development goals but rather that a discussion has taken place with society about the benefits of the research to society.

# A.6 Economics of innovation – the rise of partnerships for innovation

Classical economics holds that a nation will export goods in which it is relatively well endowed [4]. However, a nation creates, rather than inherits the most important factors of production such as skilled workforce or a strong scientific base. This reality, demonstrated by Porter [4], demonstrates how nations drive competitiveness not just through the efficient allocation of resources but also through investment in critical areas of non-tangible factors of production.

Driving innovation to enhance the competitive advantage of the nation is therefore a key economic policy issue in both macro and micro-economics. At the macro-level until the 1990s it was assumed that technological progress happened as a result of some external force [22] - possibly akin to the invisible hand [163]. Progress happened, argued classical economists as the result of competition for ideas and the interaction between the factors of production that drove total factor productivity [119].

In an economy where economic success is based on the exploitation of knowledge if that knowledge is exogenous (even if not unimportant [164]) then it leaves an unmanageable and significant weakness in economic theory especially as knowledge and skills are unevenly distributed within regions and nations [22]. Further, if economic growth is now not the result of efficient allocation of resources but rather exploitation of knowledge then the classical approach is not helpful.

An alternative view, developed during the 1980s, was that growth happened as a result of a network of decisions made within an economy. In the endogenous growth theory [117] innovation, or technological change happens because of a deliberate decision by individuals and these decisions are influenced by policy measures [117].

At the micro-economic level innovation and the ability to serve a customer's needs over the longer term determine profitability of a business [165]. The problem is to take an issue that is in essence is a micro-economic problem (different skills being appropriate to different technologies and different geographies) to a national or macro policy context to help drive economic growth.

The key issue here is not the neo-classical efficient allocation of factors of production but the ability of the individual to learn [22] and for society to invest in specialisation [4]. Lundvall [22] argues that the efficient allocation of resources will mean that an economy may grow in the neo-classical sense in the short term by driving out inefficiency, but in the long run efficient allocation of resources will lead to stagnation. The key to future growth will be innovation. But growth will be driven by innovation rather than knowledge *per se*, which takes us straight back to Drucker [114, 121]. The key issue here is that it is the ability to utilise knowledge that drives economic growth rather than knowledge itself. So, the driver of growth will be the ability to learn, as knowledge stagnates. In the same way efficient allocation of resources will drive growth in the short-term; utilisation of a fixed set of knowledge will drive economic growth until the knowledge has been used up. This argument is an economic argument that does not consider social or environmental concerns. However, it is a useful eulogy as it helpfully demonstrates the economic and broader policy rationale behind knowledge and innovation as key drivers of economic growth in modern capitalism.

Consideration of the policy implications brings into focus the issue of institutions. Institutions embody the norms, habits and rules of society [22]. This is particularly true for a knowledge economy based on the exploitation of knowledge for economic growth and rising standards of living. This pathway to growth creates fundamental uncertainty [22] and consequently the role of institutions (as the holders of societal norms) becomes increasingly important (see Table A.10). They become increasingly important because of the break with previous neo-classical assumptions

of perfect knowledge and decision making. It is the addition of social capital into economic development policy that is important [29, 166].

Social capital is "an instantiated informal norm that promotes co-operation between two or more individuals" [29]. But in earlier work he defined social capital as a moral community: that allows members of a given society to trust one another and cooperate in the formation of new groups and associations [167]. In terms of economics the importance of social capital is to reduce transaction costs (thus it helps with the efficient and productive allocation of resources); it is central to the development of trust, but like trust it is difficult to generate with public policy [166, 168]. Interestingly globalisation, as a carrier of ideas, can create social capital [29] and influence through the reduction of proximity even the most powerful states [169].

This development of innovation as a key to the future of economic development led to the development of the concept of national innovation systems [20, 22, 23]. National innovation system research is work looking into the different innovation infrastructure in different countries. It augments the work on competitive advantage [4] by describing the difference in innovative capacity between countries [23]. There is debate about whether a national system of innovation makes sense in an increasingly globalised economy, but the nation state is still the level at which policy is made and developed, welfare levels determined, and the macro-economic context is set [22]. Indeed Porter [4] argues that the nation state is becoming more, not less important. For the purposes of this exegesis the argument is increasingly moot, but the nation state is still able to develop policies to compete on a multi- or international stage.

Type of institution	Example	Importance of the issue
Time horizon of agents	Short term time horizons in some western economies	A short-term mindset will be appropriate for the exploitation of some, but not all knowledge
The role of trust in the economy	Loss of trust through nationalisation or transition from a centralised to market led economy	Trust can be replaced by strong (but expensive) legal institutions
Rationality of agents	Much economic theory assumes that decision making is rational and perfect	The exploitation of knowledge for economic advancement is not an individual endeavour based on instrumental rationality. It is based on communicative rationality
Forms of authority	Trust and authority are necessary if learning is to be efficient	The basis of authority (age, trust, financial) maybe a key determinant in the form and style of innovation.

Table A.10	Lundvall's [22] four types of institution important for learning and
	innovation

The key message to emerge from the national systems of innovation school is the development of thinking that differentiates (but also incorporates) the concepts of tangibility and reproducibility (see Table A.11) [22]. Tangible resources are reduced in the economic process (e.g., consumption of energy and materials) but non-tangible resources (e.g., intellectual capital or knowledge) can be increased in the process [24] It also was the point at which the linearity of innovation processes was questioned.

Linear innovation systems (see Table A.11), inspired by Bush [25] and based on Mode 1 knowledge production and the basis of much government policy post-war, are based on demand pull or technology push. But they fail to adequately account for the complexity of the innovation process or, importantly the complexity of the network of relationships and more importantly was insufficient, alone, to create an effective transfer of knowledge to the marketplace [20]. However, this does not mean that a linear model of innovation is redundant, but rather a non-linear model is a useful addition to the theory [21].
Linear models of innovation suited the institutional paradigms of post-war reconstruction. Knowledge (rather than innovation) was less ubiquitous and institutional demarcation was in place (see Section 2.2.1). In America the linear model became less structured with the creation of land universities and the passage of the Bayh-Dole Act [11]). This period also represented an era where the timelines between knowledge creation and innovation were shortening [26] and an increased focus on the productive use of public resources.

This led to new growth economics based on endogenous growth and a more transdisciplinary approach to economic policy development. It also led to a clearer understanding of the physical limits to growth [96] as well as the non-traded interdependencies, such as social capital development [22]. Lundvall [22] argues that the development of evolutionary economics makes the challenge to sustainability clear: left unimpeded market forces will erode the basis for economic growth.

Table A.11Resources fundamental for economic growth - combining the tangible<br/>and reproducible dimensions

	Reproducible resources	Non-reproducible resources
Tangible resources	Production capital	Natural capital
Intangible resources	Intellectual capital	Social capital

Evolutionary economics or not, the importance of the national systems approach is twofold. One is that it holds social capital development (or depletion) close to its heart. It is the deployment of social capital that reduces transaction costs, creates (or depletes) trust and puts the social dimension at the heart of innovation. The second is the construction of a system of innovation, based on a social construct, that can evolve over time (an endless transition [20]).

These two dynamics are the key to understanding the development of partnerships for innovation and what distinguishes it from previous models of innovation. Through this lens the Triple [27], Quadruple [21] and Quintuple [32] Helices are an extension of, or evolution from, the national systems approach as they retain a social construct at their core but take an evolutionary systems approach. The establishment of innovation around a social construct can be seen, in part at least, as the start of a development of a nonlinear approach to innovation.

Within a national system there are also different styles of innovation with each style representing economically relevant knowledge [24]. The styles of innovation are an important concept as they again reinforce the social dimension (see Table A.12 based on [24, 30]) as well as the start of a distinction between the national (or macro) and the local (or micro) formulation and implementation of innovation policy.

Style	Description	Source of knowledge	Ease pf reproduction
Know-what	Fact based information	Internet, lectures, books	Easy
Know-why	Principle, theoretical or legal law-based information	Internet, lectures, books	Easy
Know-how	Knowledge developed by and kept within a team/firm	Practical experience	Hard (cannot be removed from human and social context)
Know-who	Who knows what and what to do	Social practice	Hard (cannot be removed from human and social context)

Table A.12 Styles of innovation

## A.7 Partnerships for innovation

The systemic approach to innovation offers the benefits of a solution where the sum of the parts does constitute the whole [20]. Left to themselves none of the actors can create a modern competitive economy, take care of the sick or protect the natural environment. It is through the interplay of the participants within the system that these issues can be resolved. It is not an economic solution but rather a quality-of-life solution where decisions are taken to balance and integrate the competing needs of the economic, social and environmental goals.

## A.8 Triple Helix model

The deepening of the knowledge economy, where knowledge provides competitive advantage, creates a new imperative for universities. That imperative is

certainly twofold: one is to continue to develop and create new knowledge but importantly to engage with business and government to seek economic rent from knowledge development. This new interplay led to the development of the Triple Helix model (see Figure 2.1 to Figure 2.4, and a fuller explanation in Table A.13) and the beginning of an evolution of a systems approach to the development of a national or multi-national innovation process [27]. The Triple Helix is a model of interactions between wealth creation (industry), knowledge production (universities) and regulations (government) [170] where each helix represents a knowledge sub-system connecting with each other in a spiral (and at a local, regional, national or super-national scale) [33].

Whilst all variants of the Triple Helix partnerships have the same actors (universities, industry and government) not all are created equal. There are several variants, and these are described in Table A.13 (which is developed from [20, 171]).

The Triple Helix model is useful because it caters for at least three different sources of national variation:

- National industrial sectors are varied and different (for example see [4, 27]);
- Different technologies induce different forms of innovation; and
- Systems of innovation integrate the various functions in a manner that is suitable to the local situation.

These variations can be seen as both exogenous (that is to say they are the result of competitive activities) but within the Triple Helix they are also endogenous (as they are the result of institutional settings [27]). With the evolution of the Triple Helix those endogenous settings are working together resulting in a spiral pattern of linkages [27]. The model embraces the traditional didactic relationships and takes account of the expanding role of knowledge which due to the reflexive nature of human (and thereby organisational) behaviour generates new structures within (for example research centres and institutes) and between each of them (for example hybrid organisations) [38].

Within the Triple Helix model governance arrangements are also changing and becoming increasingly reflexive. They change their positions according to institutional constraints and opportunities. As such governance becomes a nested structure of reflexive controls [38]. What is less clear is what the reflexive controls seek to achieve?

In the literature there are several outcomes of innovation that emerge to drive the delivery of innovation (see Table A.14).

Part of the strength of the Triple Helix model is that it remains relevant in times of uncertainty. This is the key to reflexivity – institutions react according to their best interests (or mutual selections [38]) but within a broader framework. Therefore, the Triple Helix is an evolutionary model – it evolves along a pathway that is considered by the partnership as acceptable according to the self-interested decisions (or institutional constraints and opportunities [38]) of the interested partners (wealth creation, knowledge and regulations [170]). These self-interested decisions need to help move towards an outcome that can never truly be achieved. In many respects it therefore truly represents an infinite triple helix. This evolution is expected in the Triple Helix model. Triple helices are complex and potentially unstable (cf Drucker and Section 1.2) and complex dynamics are expected to occur [27] but the model is sufficiently complex to accommodate chaotic behaviour [20].

	Description	Example
Triple Helix 1: Statist	Nation state encompasses academia and industry and directs relations between them	This might be the case in former Soviet economies or the economies of former eastern Europe. But it can also be recognised in an economy that has a strong government who can direct or incentivise behaviour
Triple Helix 2: <i>Laissez faire</i>	A <i>laissez faire</i> model. Separate institutional spheres with strong boundaries between them	Exemplified by economies that have clear rules about the roles of different organisations (e.g., Sweden and USA in the 1990s)
Triple Helix 3: Overlapping or balanced	Generation of knowledge through the overlapping of institutional spheres. Developed in recognition of the various different forms of government, business and university.	Adopted by countries with an approach that is driven by delivery of desired outcomes.

Table A.13	The evolution of the	Triple Helix	model of innovation
------------	----------------------	--------------	---------------------

	Description	Example
Triple Helix 3: Social structure	New organisational innovations arise from interactions amongst the three helixes.	In a democracy government represents the views of the people. However, in a knowledge democracy the people can represent their own interests and therefore need to be engaged (see Section A5 on responsible research and open science)
Triple Helix 4: Delayed Government-led model	Identified in China this model is led by university and industry, but Government involvement only happens later, but then takes over	Tongji Creative Cluster in China – an industry/academia initiative that then was subsumed into central planning

## A.8.1 Theoretical foundations of the Triple Helix model.

The Triple Helix model is based on the triadic tradition [42] of the social theories of Simmel, Marx and Webber [172]. At the heart of the triadic tradition is acting as another (or strictly acting in the absence of another) – a recurring theme in Etzkowitz's work is "each taking the role of the other" (see [42, 172] and Section 2.2.1 for a discussion of the triadic tradition). The creation of the Triple Helix theory was the result of inductive reasoning of successful practices of regional innovation [42], but the model itself derives inspiration from evolutionary economics, sociology, and policy [173]. There are five aspects that create optimal conditions for innovation and these are set out in Table A.15 (developed from [42]).

The model works on two levels: one is to leverage, or forge relationships between business, universities and government for a common cause (demonstrably economic growth see Chapter 4 and Publication 1) but also to change the institutional settings to help deliver the closer relationships (or develop the social capital) that will drive economic growth [171]. This change of institutional settings, or mutual selections can lead to the creation of niches [38] where localised reflexive responses are created (see Table A.16).

Author	Outcome sought	
Loudordorff [20]	High technology development	
	Global knowledge economy	
Etzkowitz [27]	Wealth creation	
Etzkowitz and Leydesdorff [20]	New technologies (for example biotechnology and ICT))	
Etzkowitz et al [28]	Improving regional or national economic performance	
Etzkowitz [39]	Transition to a knowledge-based economy	
Etzkowitz [40]	Technological innovation and economic development	
Etzkowitz [8]	Science based innovation	
Etzkowitz and Zhou [41]	Developing technologies and applied research	
	Foster regional economic growth and promote	
Cai and Etzkowitz[171]	entrepreneurship;	
	Research and action	

Because each helix is working to its own self-reflexive interest, within a larger overriding framework two further issues arise. One is about the need for clear and effective communication between the helices. If this does not happen, then there is the risk of discord and disruptive evolution [20]. This is an issue that has been recognised by policy makers and why in Europe there is a move to open science, and latterly responsible research and innovation [37] and in Australia responsible innovation [105] (and see Section A.5).

Table A.15	Five aspects of a	normative balanced	Triple Helix model
			mpic richt mouch

Aspect	Explanation
Triadic interactions using an Occam's Razor ⁷ to "remove what is unnecessary" [174] p.145)	The three elements introduce the potential for innovation in innovation with the focus on the reciprocal relations between the three helixes which will be fluid and suffused with (creative) tension.
Take on the role of the other (in terms of adopting new roles)	Institutions doing this differently – for example universities becoming more entrepreneurial and creating new organisations in partnership with other institutions
Evolutionary mechanisms	The model is dynamic with the first two factors creating a non-static space at system and sub-system level (namely market, control and innovation (Leydesdorrf in [42]. The creation of niches further creates dynamic instability and may lead to institutions mutating and changing their genes [42]
Top-down, bottom-up initiatives and lateral coordination	The model is agnostic about where power lies; its strength comes from how power is used. This can be to put in place top-down mechanisms or respond to bottom-up initiatives. This is a development on Cai's [42] argument that government does top down; but in reality, in the Triple Helix any partner can do top down – the key is how the other partners respond
Leadership and capabilities	Triple Helix interactions are enabled by 2 types of conditions – the "sufficient condition of convening authority and the necessary condition of innovation capacity" [42] which links convening authority to individuals who have respect (i.e. social capital) by all partners, but this feels too purist in its approach.

⁷ Occam's Razor precludes over-complication unless the complication provides clarity. See Schaffer (2015) for a fuller discussion of the principle and its potential for application in economics.

The point of this work is to deliver economic growth in a manner that society is comfortable with and in so doing helps to transcend the issue of cross-institutional translation [38]. This issue, though not in these terms, has led to the creation of the quadruple helix (see next section)

As discussed in Section A.5, the work on Responsible Innovation is currently strategic in nature and revolves around Triple Helix stakeholders working together to ensure that the next economic wave, or cycle [15] is consistent with, or at least cognisant of, the wider view of society. Currently this work is about the application of technology (for example artificial intelligence [162] or biotechnology [3]) but there is space for these conversations to also evolve into answering questions such as how can we develop partnerships for innovation that deliver a net-zero economy? And this issue will be discussed further in Section A.10 on the Quintuple Helix and in Chapter 6.

Author	Area of study	Scale of study
Cooke [3]	Biotechnology	Regional
Almiral [175]	Social constructs in technology	Local to regional
Hughes [176]	Social living labs	Local
Antonopoulous [177]	Patras Science Park	Regional innovation
Aportela-Rodriguez [79]	Libraries as a link to business	Local
Hladchenko & Pinheiro [178]	Triple Helix in former Soviet Union	National
Artto [108]	Smart Cities and innovation eco- systems	Local
Aslani [179]	Comparative study of Science Parks	Local (but international comparison)
Jongwanich [180]	Triple Helix in China	Regional
Ayvari [75]	Public, private people partnership	Local
Baccarne [90, 128]	Citizen centric urban planning and socio-ecological entrepreneurship	Local
Lukkari [77]	Innovation service platform	City
Bartelt [181]	Public sector and sustainability	City
Berker and Woods [182]	Repurposing university campus	Campus
Boeri [2]	Community development through living lab	Local

 Table A.16
 Examples of the Triple Helix model being applied (Publication 1)

The Triple Helix model is a model that is evolving and that is its strength as well as its weakness [35]. The Triple Helix does not, strictly, represent a departure from previous innovation models, but is a development based on inductive reasoning (see [7]) based on evidence from successful innovation practices. It does reflect the increased importance of the university in the triad of relationships, but this is a reflection of the changing economic dynamics rather than model design.

The Triple Helix model is ethno-discursive and has created a considerable body of literature, 17 conferences as well as a research peak body (Triple Helix). It has been used by the OECD, the World Bank and national and regional governments [42]. Because of its evolving nature and fluidity, the Triple Helix model has been criticised as too nebulous to implement. However, Cai [171] has used institutional logics to enhance the context sensitivity of the concept to develop an ideal Triple Helix model based on Western proclivities (see Table A.17).

It is interesting to also reflect that the partners in the triple helix are evolving: universities becoming entrepreneurial (see Section A.1.3) and undertaking contextual [3] or responsible research [105] (see Section A.5); the production of knowledge is moving firmly to a Mode 2, and increasingly Mode 3 transdisciplinary process; in Government the political discourse is about "developing policy as if people matter"[183] delivering the sustainable development goals with the focus on the delivery of policy outcomes that work rather than are ideologically pure; and business is increasingly focussed on open innovation [17] and on stakeholders (see, for example, [16]) as well as the development of a social license to operate ([184].

## A.8.2 Criticism of the Triple Helix model

While the Triple Helix model has been widely cited in the literature (for example see [170]) there are questions about its actual impact [35]). There are three consistent strands of criticism of the model

 It is based on a western liberal economic model [33] and pays little attention to national contexts and as such cannot be overlaid onto developed [164] or developing economies without catering for the asymmetric knowledge problem [3] through an elaboration of the context sensitivities [171];

- The model is too general or too strategic [3], lacks a theoretical foundation at the micro (or implementation) level and as such so nebulous as to be more of a think tank than a theory; and [35]
- The model pays insufficient attention to the views of the public or civil society [31] or the natural environment [185].

Irregardless of the critique, the model has promoted a debate in the literature and adopted in wider society about both the knowledge economy and the most effective ways for government, business and academia to work together to create economic growth to benefit society.

Stages of de	evelopment	Triple Helix activities	Institutional logics
Stage 1	Realisation of the need	Realising the importance of entering a reciprocal relationship between university, industry and government	Shared beliefs on knowledge as a key to economic growth
Stage 2	Intra organisational transformation	Taking the role of the other	Market oriented organisational culture Process orientated management
Stage 3	Interactions between organisations	Growing and innovating through co-operation Generating hybrid organisations	Effective protection for intellectual property Participate in civil society
Stage 4	Institutionalisation of the Triple Helix model	Feedback loops between participants Institutionalised norms of entrepreneurial university, innovation state, knowledge- based growth	Competitive market environment Democratic policy making process

 Table A.17
 Institutional orders in the evolution of the Triple Helix model

## A.9 Quadruple Helix

In consideration of Mode 3 Knowledge production and the need to embed scientific discourse within a community [21] came the proposed the addition of a fourth helix to the Triple Helix model. The role of the fourth helix is to act as a codification of knowledge within society – it too is a reflexive helix but unlike the other helices it does not exist as an institution or a body that is subject to institutional rules. The fourth helix however plays a theoretically important role and is consistent with the European drive for open science. So here the fourth helix represents media and culture based public relations [21]. But it is more than public relations as it is envisaged that it might result in a 'democracy of knowledge' [21].

The fourth helix represents the importance of public, or deployment of social capital for the "successful achieving of goals and objectives" [21] p218. Whilst it is not clear who 'owns' the goals and objectives and the public are influenced by culture and values, but so too are industry, government and academia.

This is the importance of social capital [29] to the debate. What is not clear is why the public's social capital is differently applied or will result in different reflexive positioning than the other helices. The key to answering this question is the appreciation that the Quadruple Helix is the media-based reality construction [31] and represents the social structure that underpins the Triple Helix.

The addition of a fourth helix does bring with it a discussion of the democratisation of knowledge production and innovation systems [36]. What is less clear is why this democratisation of knowledge is not catered for in the government helix (particularly in a Western industrial democracy)? In a sense this is answered by von Hippel [16] who argues that users (both business and individuals) have increased power through the democratisation of innovation (using knowledge) and are able to innovate for themselves (this links back to Drucker and individuals, and organisations innovating). Carayannis et al [185] argue that the Quadruple Helix has the user at its core which is where the strength of the Quadruple Helix lies.

## A.10 Quintuple Helix

The Quintuple Helix adds a further helix to the Quadruple Helix. The fifth helix is the environment or natural environments. In so doing it is argued the Quintuple Helix offers an analytical framework where knowledge and innovation are being connected with the environment or natural environments [21]. The development of the Quintuple helix is consistent with the transition to Mode 3 knowledge production and sets an ecological boundary around the conceptualisation of knowledge production and innovation. This is defined as a co-opetitive spatial and sectoral fractal innovation and entrepreneurial ecosystem (CS2FIE2 [185]). The emphasis is on the innovation process as an ecosystem whereas under the Quadruple Helix model it is referred to as an innovation system [185].

Under this model two definitions of sustainable development are offered. One is the Brundtland definition of "development that meets the needs of the current generation without compromising the ability of future generations to meet their own needs" [144]. Carayannis et al [185] however, introduce a further definition of sustainable development as "a co-evolution of the different systems of society, based on knowledge and a mutual cross-learning that is socially and environmentally sensitive and that is receptive for concepts of a quality of democracy and demonstrate the usefulness of social ecology and the potential for a socio-metabolic transition where material and energy flows are in equilibrium for the benefit of society and nature".

The natural environment(s) of the fifth helix is not a helix nor is it presented methodologically as a boundary within which the innovation processes take place. It is stylised as carrying out the quasi function of a helix as it promotes innovation and under the Quintuple Helix all systems influence each other [32]. The promotion of sustainable innovation is the useful contribution of the quintuple helix. Although criticised for being anthropomorphic (see [42] and next section) the usefulness of the fifth boundary helix is to transposition the discussion from "what society cannot do" due to a planetary boundary to one where the emphasis on "what can society do" but within this model *without* planetary constraint. Irregardless, this is useful as it is the classic representation of the economic problem of unlimited wants constrained by limited resources (see for example [186]). Bringing together

knowledge creation, innovation systems and the environment explicitly within that context is helpful.

There is a 5-step flow analysis to help understand the practical implementation of the Quintuple Helix [32]. The example starts with an investment in *education* and the process is then explained in Table A.18.

## A.10.1 Criticism of the Quintuple Helix

Cai and Etzkowitz [42] describe the addition of natural environments of the society as the fifth helix as "introducing and anthropomorphic fallacy into the model" (p9). They do not however explain the fallacy, although the language chosen by Carayannis [32] does suggest an unnecessary and uncharacteristically loose anthropomorphic sentiment (for example see Step 3 in Table A.18). However, the principle of including representation of a finite constraint (however nebulous) that reflects the reality confronting society (see [97] [187]) is useful. Within this model that constraint, however, is negated by its inclusion as another sub-system rather than as a system boundary.

The key, but unstated in the literature, weakness of the Quintuple Helix model is a return to a more linear approach to innovation, where innovation becomes more of a process than an evolution, with insufficient explanation as to why wealth (being the creation of various forms of capital) will be more sustainable under this model than the Triple Helix or any other model. Further the use of various sobriquets to describe what the fifth helix is presented to achieve is unhelpful (see A19 for definitional issues).

As there is no planetary boundary within the Quintuple Helix the issue of finite resources does not interact with decision making and does not offer an explanation as to why a Quintuple Helix model, though with the addition of a fifth helix, will be any more sustainable then the Triple or Quadruple Helix.

138

Step	Innovation sub- system	Description
1	Education	Inflow of investment into education which creates new impulses and suggestions for knowledge creation, which in turn increases research output in sustainable development. Due to the increase in human capital this output becomes an input into the economic system.
2	Economy	The input into the economic helix increases the value of the knowledge economy through the creation of new jobs etc. In this economic sub-system demand for corporate social responsibility and a sustainable future is increased.
3	Natural Environment(s)	The economic output "communicates with nature that it will be increasingly protected" (p8). The output of this sub-system is to provide more environmental protection and a better quality of life for people. According to Barth [188] the economic capital of know how in this context is sustainability.
4	Media-based and culture-based public	The public receive information capital about how to live a greener lifestyle and the new knowledge created should be implemented by the newly created social capital in the social- based public. The output of this system is information on "wishes, needs, problems or satisfaction of citizens into the political system
5	Political system	The input from the culture-based public (and other sub-systems creates new political and legal capital that makes the likelihood of living in a less unsustainable world achievable. The output from the political system then feeds the other subsystems.

 Table A.18
 The 5-step flow analysis of Innovation under a Quintuple Helix model

According to Cai [42] a further weakness is that the triadic tradition is lost. The triadic tradition removed the linearity from the national innovation system and introduced an unconstrained evolutionary approach. The triadic tradition (*entwicklung*) is powerful. At its core it proposes that philosophy will not contradict experience but rather experience will contribute to philosophy to reveal the true explanation. Triadic represents three but should not be misconstrued as the three helices. The three elements of the triadic tradition are more about process than institutions. The three parts are: in itself; out of itself and finally in and for itself. The triadic tradition is powerful as it focuses on what might become as well as what is [189].

Studies of the quintuple helix are currently limited and led by Carayannis [32, 185]. Baccarne [128] has taken the concept and applied it to an urban living lab concluding that the urban living lab approach is a way to put the model into practice at the level of a single localised innovation development process. It has also been applied, using fuzzy logic to the *ex-ante* implementation of the sustainable development goals in Latin America [33], rather than the decision making that led to the implementation.

## A.11 Triple, Quadruple and Quintuple models: Summary

The Quintuple Helix model is a conceptually useful development as it provides a conceptual extension of the modes of knowledge creation and the systems of how the knowledge is used to innovate. Conceptually the Quintuple model also provides an envelope within which innovation needs to take place. It is also non-judgemental about the weighting between the helixes [33]. Providing this structure does mean that it reconceptualises economic opportunity and the opportunity for knowledge creation and innovation (see [185]. It starts to answer the questions posed by Boulding [97] but does not yet provide the answer to Hardin's [91] search for a technical answer to the tragedy of the commons.

## A.11.1 Partnerships for innovation – the creation of hybrid organisations

As discussed in Section A.8 the Triple Helix model gives rise to the potential for new hybrid organisations due to the interaction between the helices [27]. The deepening of the knowledge-based economy is affecting how all partners interact – government, industry and universities but also how consumers and citizens interact with other partners in the innovation ecosystem [43]. As discussed in Section A.9 the inclusion of the local citizen has led to the conceptualisation of a Quadruple Helix model [31]. But the rate of pace of change, partly driven by the new power dynamics offered through the democratisation of knowledge [21] is giving rise to new forms of individualised partnerships [44] at different scales and different dynamics [46]. This result is as predicted by Etzkowitz [27] in his proposal of the Triple Helix model. In ecological vernacular these would be referred to as niches that are created through some form of mutual serendipity to meet a localised need [38].

Much of the literature (see Table A.19) refers to these hybrid organisations, or intermediaries for innovation as resulting from a top-down or bottom-up initiative. There

are no dissenting voices in the literature to adopting this approach to suggest that it is theoretically invalid (see Section 4.1.2). But the findings of the systematic literature review undertaken as part of the research suggests a further refinement to this paradigm that takes into account the increasingly important role of society in its many forms in developing innovation (Publication 1 [12]). The evidence underpinning the new model has been discussed in Section A.5 on Responsible Research and Innovation which has resulted from the international push towards open science [37], the development in business of a social license to operate [184]; knowledge production moving from Mode 2 to Mode 3 and the take up of open innovation [17] and policy being created "as if people matter" [183].

Model	Authors	Definition of helices	Summary of focus
Triplo Holiy	Etzkowitz and	Universities, industry	Knowledge
пре пенх	Leydesdorff [27]	government relations	economy
Quadruple Helix	Carayannis and Campbell [21]	Media-based and culture- based public Civil Society [185] Citizens ([190] Public environment [33]	Knowledge society
Quintruple Helix	Carayannis and Campbell [32]	Environment Natural environment(s) Social ecology Social, or societal, ecosystem Natural environment(s) of society Ecological Capital (Barcellos- Paula 2021)	Socio-ecological transition

 Table A.19
 Definition of helices in Triple, Quadruple and Quintuple models

This empowerment of people needs to be recognised and this has led to the evolution of the proposed three streams of partnership development which is a revisualisation of the extant top-down/bottom-up approach (see Table A.20) currently endemic in the literature. Both the existing top-down/bottom-up analysis and the three-stream approach are based on the premise that it is through the development of situationally specific partnerships at different scales and under different dynamics that best facilitates the likelihood of innovation happening (the innovation equivalence of Darwinism). This is discussed in full in Section 6.2.





## A.11.2 Concluding remarks

This literature review complements the systematic literature review published for this thesis (Publication 1). Building on the top-down/bottom-up paradigm [34] in the published literature it is demonstrated that partnerships for innovation are the result of progressive changes in knowledge production and are also a result of the democratisation of knowledge and the emergence of society as an important partner in innovation. These external stimuli to the ecosystem of knowledge production and partnerships for innovation have then created their own structures that reflect the changing external stimulus.

This literature review also demonstrates that there has been an evolution in thinking about knowledge production and innovation processes. The pre-eminent model in the literature (and widely adopted in society) is the Triple Helix [27]. This then was developed into a Quadruple Helix (to consider media-based and society-based public participation) [21]. This in turn was developed into the Quintuple Helix to consider sustainable development [32]. This literature review, and the associated systematic literature review (Publication 1 [12]) has introduced the concepts of exogenous stimuli that has necessitated a development in the theoretical models.

However, the literature review has also revealed that the model development (as set out in Table 7.7) suffers from theoretical weaknesses that mitigate its usefulness. These are discussed further in Chapter 4.

## Appendix B Publications

## B.1 Publication 1

Burbridge, M. & G. Morrison (2021) A systematic literature review of partnership

development at the university-industry-government nexus. Sustainability 13, no. 24: 13780. <u>https://doi.org/10.3390/su132413780</u>





#### Review

# A Systematic Literature Review of Partnership Development at the University–Industry–Government Nexus

Mike Burbridge * and Gregory M. Morrison 💿

Curtin University Sustainability Policy Institute, School of Design and the Built Environment, Curtin University, Perth 6102, Australia; greg.morrison@curtin.edu.au

* Correspondence: mike.burbridge@postgrad.curtin.edu.au

Abstract: The increasingly entrepreneurial intent of universities implies the commercialization of knowledge and innovation through the triple helix of interactions between universities, industry and government. However, there remains a lack of clarity concerning best practice partnerships for innovation. This systematic literature review (SLR) provides insights onto the development of partnerships at the university-industry-government nexus and builds on the existing top-down/bottom-up approach for the creation of intermediaries of innovation. The SLR describes the evolution of these intermediaries, which is driven both by criteria set by partners and the globalization of the knowledge economy. This SLR reveals that the partnership structure most likely to further economic and broader societal goals is the living lab with the inherent focus on open innovation and co-creation. This SLR reveals that the living lab structure (and including sustainability labs and urban living labs) is the partnership structure utilized for innovation that addresses economic, social and environmental goals. Two areas are recommended for further research. One concerns the development of a deeper understanding of the relationship between the evolution in the structures of partnerships for innovation and how it is influenced by the globalization of the economy, society and environment, and changing modes of knowledge production. The other is to better understand why the living lab approach to partnership creation is best suited to the delivery of sustainable development objectives and how this learning can be applied to other models of partnership development at the university-industry-government nexus.

Keywords: triple helix; innovation; partnership development; sustainable development

#### 1. Introduction

Since the 1940s, there has been a push to encourage universities and industry to increase their engagement to commercialize research [1]. This started with the use of government procurement to encourage research and the development of innovative products [2] initially in the space, defense and energy sectors [3]. The 1950s and 1960s saw the development of science and/or technology parks to commercialize research [4] in America, which was followed, during the 1980s, in the UK [5].

During the 1980s, universities developed an increasing focus on technology transfer [6] to facilitate engagement with business. However, it was only from the 1990s that universities became directly involved in the world of business, actively seeking to commercialize their knowledge and research [3].

The benefits of collaboration range from the local (commercialization of research and innovation [7]) to the regional (revitalization of regions [8,9]) or national (catalyst for techno-economic development [10]) levels. Collaboration is a key part of modern innovation [11], which in turn is an important part of a well-developed entrepreneurial industrial sector [12].

The triple helix model [13] highlights the importance of a partnership between universities, industry and government to create innovation that meets business objectives in



Citation: Burbridge, M.; Morrison, G.M. A Systematic Literature Review of Partnership Development at the University–Industry–Government Nexus. *Sustainability* **2021**, *13*, 13780. https://doi.org/10.3390/ sul.32413780

#### Academic Editor: Andrea Pérez

Received: 16 October 2021 Accepted: 8 December 2021 Published: 14 December 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). developing and commercializing universities' research outcomes. Including government within the partnership creates outcomes that are socially and economically beneficial [14,15]. Others [16] have proposed the addition of a fourth helix (the quadruple helix) to promote a democratic approach to innovation, where society can provide feedback to create socially acceptable policies and practices. However, this model does not include the explicit consideration of non-market parameters, such as the natural environment [17]. The quintuple helix [18] seeks to address this shortcoming.

However, universities, industry and government struggle to effectively partner particularly at scale—to deliver economic benefits from the commercialization of research outputs [19–21].

The aim of this systematic literature review (SLR) is to identify best practice partnership development to promote innovation at the university–industry–government nexus. To do this, it seeks to understand how universities partner with external organizations to deliver innovation and tracks the development of these partnerships.

This article is structured as follows. Section 2 sets out the methods used in the SLR; Section 3 provides a quantitative analysis of the papers revealed through the SLR; Section 4 discusses these results and puts forward a new understanding of the development of partnerships for innovation at the university–industry–government nexus; and Section 5 proposes a further research agenda.

#### 2. Methods

The SLR offers a stringent methodology to interrogate a large body of data in a manner that is repeatable and evidence based [22]. This removes much of the inherent bias that can befall more traditional literature reviews [23]. However, using repeatable search terms does not in itself remove the potential for author bias [23]. The SLR provides a stringent approach, where the selection of studies is repeatable, but where the interpretation of those studies is left to the authors [24]. In order to develop a repeatable process for the interpretation of results, this study applies a thematic analysis approach developed by Massey (2011), which was initially proposed for use in interpreting the results of focus groups [25]. Massey's [25] process is applied here to further manage author bias by highlighting the epistemological foundation for the work.

This SLR is consistent with PRISMA (preferred reporting items for systematic reviews and meta-analysis) [26] and follows the process set out in Figure 1.

The articles were analyzed using Massey's model for thematic analysis [25]. This approach increases the replicability of the study, as it proposes three levels for data analysis: articulated, attributional and emergent [25]. Categorizing data into these three levels enables a transparent analysis which is repeatable and develops an analysis that is descriptive and based on a systematic approach.

However, Massey's (2011) model is not immediately transferable, as it was designed to be used for analyzing data arising from focus groups rather than from SLRs. Massey's three definitions of each data level are included in Table 1. The three levels of data should be viewed as a hierarchy, where the potential for bias is inverted according to its position in the hierarchy, with articulated data being subject to the least bias (and therefore, most replicable), whereas attributional and emergent data are the least transparent, due to the necessary engagement with the authors' views and experiences in the SLR process.



**Figure 1.** PRISMA—flow of information through the different phases of a systematic review [26]. The number of results is shown in Tables 2–8. # = number.

Table 1. Model	for thematic	analysis of dat	a—based on l	251.
THOTO TH HIDOROF	tor energie	and you or ac	a paoca on	

Levels of Data	Original Definition	<b>Revised</b> Definition	Source of Data	Potential for Bias
Articulated data	Information that is expressed in response to, or specifically addresses the questions posed	Information that is stated directly by the original research authors as the result of their research. This is the original author's statement of results achieved and reported in the articled	Data are articulated by one or more study in the SLR	Lowest
Attributional data	Comments and discussion that relate to a priori hypotheses or theories that the evaluator brings to the study	Unchanged	Data are a co-location of ideas from 2 or more articles and are supported by extant models and theories	Medium
Emergent data	Information that contributes to new insights and hypothesis formulation and is the unanticipated product of individual comments and exchanges amongst group members.	Information that contributes to new insights and hypothesis formulation and is the unanticipated product of connections arising from different articles	Data that are the result of the consideration of articulated and attributional data. They are not necessarily supported by an extant body of work	Highest

#### 2.1. Research Question

The aim of this SLR is to identify best practice partnership development to promote innovation in the university-industry-government nexus. To do this, it seeks to understand how universities partner with external organizations to deliver innovation. Further, it investigates the question of how innovation can be undertaken in a manner that is consistent with sustainable development.

#### 2.2. Identification of Relevant Studies

Having defined the research question, it was then necessary to construct a search string for use in 3 key databases (ProQuest, Scopus and Web of Science). Defining the search terms was an iterative process, but the Boolean search was designed to reveal the state of the literature regarding how universities partner with business (and government) to transfer or develop knowledge to create and/or commercialize innovation.

The search was initially narrowed to include sustainability, carbon, sustainable development (and variants) in order to consider the appropriateness of the research or innovation to move society toward delivery of the sustainable development goals [27] but adding these elements to the search string returned too few articles from the search. A new search focusing on sustainable development and partnerships for innovation at the university–industry–government nexus was undertaken, which revealed a further 7 articles.

The development of the search string was determined from the problem definition and then combined into a Boolean research string (as set out in Table 2). Where the option was given, databases were searched for peer-reviewed articles written in English. There was no temporal constraint imposed.

#### Table 2. Development of Boolean search string.

×c	
Key Areas	String Expression
Activity	Innovati * or research
Location of activity	"campus develop *" OR "living lab *" OR "sus * lab *" OR "tech * park" OR "science park" OR "innovat * park"
Form of activity	collaboration OR "co-creat *" OR partner * OR "triple helix"
Between	university OR academi* OR college AND industry OR business OR commerce

* = root, stem or truncation.

#### 2.3. Process of Filtering Studies in the Systematic Literature Review

The process of filtering studies in this SLR follows the PRISMA phases of a systematic review [26,28], with additional data considered relevant to the quantitative elements of the SLR retained.

#### 2.3.1. Records Identified through Database Searching

The research string was run through all three databases in December 2020. The results for each database are shown in Table 3.

#### Table 3. Records identified through database searching.

Database	Boolean Search Results
Proquest	168
Scopus	151
Web of Science	171

#### 2.3.2. Records Removed

Once the search of Scopus, Proquest and Web of Science was undertaken, the first step was to remove any duplicates within each database. Opinion articles and articles not in English were also removed from each database. The result of this process is shown in Table 4.

Table 4. Removal of duplicates, editorials and non-English papers.

Database	Search Results	Number Removed	Number Remaining
Proquest	168	14	154
Scopus	151	4	147
Web of Science	171	5	165

Having removed internal duplicates, editorials and non-English papers, all articles were reviewed to ensure that they were consistent with the objectives of the SLR. This process involved reviewing the titles and key words of the papers to make sure that the area of study was consistent with the research question. The results for this process are set out in Table 5.

Table 5. Removal of non-relevant papers (by title).

Database	Number of Papers	Number Removed	Number Remaining
Proquest	154	27	127
Scopus	147	2	145
Web of Science	165	35	130

Databases were amalgamated, and duplicates between databases were removed. The results can be seen in Table 6.

Table 6. Removal of duplicates between databases.

Database	Number of Papers	Number Removed	Number Remaining
Combined	402	65	337

2.3.3. Number of Records Screened (and Records Removed)

All abstracts were read to ensure that papers were relevant to the research question. Papers had to address innovation or research between two partners, either at some location or by addressing the issue of the location of the research or innovation activity. Papers that were not consistent with the research objectives of the SLR were removed. See Table 7.

Table 7. Removal of non-relevant papers (by abstract).

Database	Number of Papers	Number Removed	Number Remaining
Combined	337	154	183

2.3.4. Full Text Articles Assessed for Eligibility (Full Text Articles Excluded)

All papers that could be accessed were downloaded. Papers were then read and coded. Papers that were not consistent with the research criteria were at this stage eliminated. See Table 8.

Table 8. Final papers for data synthesis.

Database	Number of Papers	Number Removed	Number Remaining
Combined	183	51	132

2.3.5. Number of Studies Included in the Qualitative Synthesis and Meta-Analysis

The number of papers remaining in this study after the filtering was 132. These were then analyzed quantitatively (Section 3) and qualitatively (Section 4).

#### 3. Results

The data were synthesized through two stages. Firstly, they were analyzed according to the factors detailed in Table 9. This part of the analysis dealt with issues such as publication date, type of study, location of where research/innovation took place, and location of the lead author.

Table 9. Data synthesis-data categorization (based on Eon and Morrison in print).

	Criteria	Description
1	Year of publication	Year that the article was published
2	Country of lead author affiliation	Location of lead author affiliation
3	Type of publication	Publications were differentiated into: Journal Book chapter Conference paper
4	Type of article	Articles were divided into three broad categories: Case studies (where performance of research and innovation at a specific location or locations were described and detailed). Review papers where performance of a series of partnerships were assessed and conclusions drawn about efficacy or partnership models, future policy or research direction. This includes literature reviews Model development identifying different means of assessing the performance of research and innovation partnerships.
5	Country where case study research took place	Location of where the case study took place
6	Physical location of where the innovation took place	Detailing the focus of the research on the physical space. Spaces (or places) were categorized: Nation state Regional (including rural and peripheral areas) City Science and Technology Park Campus Living lab

The second phase (see Section 4) was the synthesis of the data. Data were coded and synthesized into the three categories set out in Section 2.

#### 3.1. Data Synthesis

All papers were categorized according to the criteria set out in Table 9. This categorization was undertaken to establish the year of publication, the location of the lead author, the type of publication and article, and the physical location of the innovation partnership. These data are largely factual, although there is an element of subjective interpretation in the categorization of papers.

#### 3.1.1. Temporal Distribution of Studies

Of the 132 papers included in this study, 106 were published in the past 10 years. The trend in publishing is increasing and the interest in partnerships for innovation is similarly increasing (see Figure 2). This trend has been observed by others (for example [29,30]). A discussion for the drivers of the increased interests in partnerships for innovation, rather than the evidence of the increase in interest, is beyond the scope of this SLR. The



figures for 2020 represent data for 11 months (as the primary searches were conducted in December 2020).

Figure 2. Temporal distribution of studies.

3.1.2. Country of Lead Author Affiliation

The country of the lead author affiliation is set out in Figure 3. The top 5 European countries published 69 studies or 52% of all the studies in the SLR. European authors published 78% of the papers identified in this SLR.



Figure 3. Lead author—country of affiliation.

A breakdown of lead author affiliation by continent is given in Table 10. This reinforces the dominance of Europe for research into partnerships for innovation. The dominance of Europe in the research field is not surprising, given that the co-creation of innovation was a priority of the Finnish Presidency of the European Council in 2006 [31].

Continent	Number of Papers
Europe	97
Asia	13
N. America	9
Australia/Oceania	6
Africa	5
S. America	2

Table 10. Lead author-affiliation by continent.

#### 3.1.3. Type of Publication

Table 11 illustrates the breakdown of the sources for the articles used in this SLR: 85% were journal articles, with 4% being from conference papers.

#### Table 11. Type of publication.

Type of Publication	Number of Papers
Journal article	112
Book chapter	14
Conference paper	6

#### 3.1.4. Type of Article

This SLR considered 132 articles. The type of article is set out in Table 12. For this analysis, we used only three broad categories of articles, as drawing the boundaries is difficult and ultimately adds little to the sum of knowledge. The principle of Occam's razor was applied [30].

#### Table 12. Type of article.

Type of Article	Number of Papers
Case study	103
Review	24
Model development	5

A case study was deemed to be a study looking at a specific partnership that was used for research and/or innovation (these could be site specific, region specific, sector specific, or process specific). A review was defined as a study that considered the performance of a range of case studies to develop a future research agenda, present that state of the field, or make proposals for future policy. Finally, model development was defined as the development of evidence for proposals of new performance matrices or new forms of partnerships. Undertaking even such a simplified classification process requires the application of expertise and is, therefore, subjective.

#### 3.1.5. Case Study—Location and Focus

An analysis was undertaken of the physical and geographic location of the 104 case studies in this SLR. The physical location was defined as the place, or predominant place, where the partnership for innovation took place. Table 13 demonstrates the physical (rather than geographic) location of all 103 case studies reviewed by this SLR.

Focus Area of Case Study	Number of Papers
Science and Technology Park	28
Campus	21
Living Lab	21
City	14
Regions	11
National	6
Business	2

Table 13. Focus area of case studies.

3.1.6. Single Country Case Studies

There were 87 single case studies in this SLR, with the remaining being trans-border or trans-continent. This demonstrates the importance of Europe to research into partnerships for innovation. Figure 4 details the geographic location of the single case studies.



Figure 4. Country of single case study.

3.1.7. Case Studies in More Than One Country

There were 18 studies that were trans-border case studies; details are set out in Table 14.

Trans- Border	Countries	Number of Case Studies	Authors
	The Netherlands, USA.	1	Curvelo Magdaniel, De Jonge [32]
International	Spain, S. Africa, Hungary, Czech Republic, Finland	1	Schaffers, Cordoba [33]
	USA., Greece, Portugal	1	Schoonmaker and Carayannis [34]
	Finland, South Africa, Spain, Sweden	3	Leminen, Westerlund [35] Leminen [36] Leminen, Nyström [37]
	Spain, Hungary, South Africa	1	Guzman, Schaffers [15]

Trans- Countries Border		Number of Case Studies	Authors		
	Spain, Mexico	1	Olvera, Piqué [4]		
- International	USA, Iran	1	Aslani, Eftekhari [38]		
-	USA, UK, Bulgaria	1	Purcell, Henriksen [39]		
- Intra-European -	Sweden, Finland	1	Buhr, Federley [40]		
	Denmark, Norway	1	Nielsen [41]		
	Finland, Belgium	1	Veeckman, Schuurman [42]		
	Portugal, UK	2	Martins [43] Germain-Alamartine and Moghadam-Saman [44]		
	The Netherlands, Sweden, Finland	1	Voytenko, McCormick [45]		
-	Finland, Spain	1	Almirall, Lee [46]		
Asia	South Korea, Taiwan	1	Yun and Lee [47]		

Table 14. Cont.

#### 4. Discussion and Analysis of the Data

The data in this SLR were analyzed using a systematic process and the results are discussed below. Section 4.1 sets out the articulated data—these are data that are revealed by one or more study. The evolution of a non-hierarchical structure emerged to assist with understanding the development of partnerships for innovation at the university—industry–government nexus. Section 4.2 presents the epistemological backdrop to the articles to present a more nuanced understanding of the articulated data. Section 4.3 presents emergent data, which are author insights developed through engagement with the articulated and attributional data.

#### 4.1. Articulated Data

This section presents the thematic results of the data analysis as revealed by the SLR. The aim of this SLR is to reveal best practice partnership development to promote innovation in the university-industry-government nexus. To do this, it seeks to understand how and why universities partner with external organizations to deliver innovation.

This section is divided as follows. Firstly, there is a reprise of the theory of partnership development at the university-industry-government nexus (Section 4.1.1). Following this is a discussion of the intermediaries for innovation created at the university-industry-government nexus and how they are currently portrayed in the literature (Section 4.1.2). Finally, an evolutionary view of the creation of intermediaries for innovation is proposed (Section 4.1.3) that describes the development of partnerships for innovation. These are based not solely on dynamics within the partnerships, but on external factors that have facilitated, or enabled, new forms of intermediaries for innovation (Section 4.1.3) present the evolution of the three forms of partnership revealed in this SLR at the university-industry-government nexus).

#### 4.1.1. Innovation at the University-Industry-Government Nexus

Economies have become increasingly dependent on the exploitation of knowledge for continued economic growth [1,4] and the role of the university is widely debated [48–51]. Universities play a key role in furthering future economic development, due to their missions to educate, carry out research and engage [52,53]. It is the third mission (engagement) that gets the most attention in terms of how universities can most effectively use the knowledge they create to further economic development [54,55] and do so in a manner that is in the economic interest of society [56,57]. To efficiently utilize their expertise in knowledge creation for the economic benefit of society, there is a need to interact, or partner,

with other organizations [34]. There is limited discussion about how universities themselves might use innovation to improve their offerings under missions one and two (see, for example [58,59]).

There is an ongoing conversation about the triple helix, where universities, government and industry interact to help drive knowledge-based economic development, particularly in an industrialized economy [51,60]. Academic interest in partnerships for innovation is increasing as reflected in Section 3.1.1. The triple helix model is largely accepted as a useful starting point to understand the changing roles of universities, industry and government to partner in innovation to drive economic development.

#### 4.1.2. Intermediaries for Innovation at the University-Industry-Government Nexus

The SLR reveals that the innovation intermediaries created by university, industry and government are created through an internal dynamic, and these are either management led (top-down) or led by an entrepreneurial individual or group (bottom-up) [54,60]. These types of intermediaries for innovation, as identified in the literature, are shown in Figure 5.



Figure 5. Innovation intermediaries in our society [54,60].

Whilst there is a focus on economic development, it is clear from the literature that the deepening knowledge-based economy is affecting not only how industry, government and universities interact, but also how consumers and citizens [34] interact with other partners in the innovation ecosystem. The pace of change affects all sectors of society (university, industry, government as well as people), and this dynamic relationship is rapidly changing, which is leading to new forms of intermediaries that are highly individualized [61]. This, in turn, leads to the opportunity for innovation at different scales and under differing dynamics and delivering different outcomes [62,63].

Various forms of intermediaries are being created as a result of internal dynamics, but also being facilitated by external opportunities and stimuli (exogenous factors). It is this overlay of changing dynamics that has led to new forms of intermediaries for emerging innovation [64] and that are being led by, or include, different actors [65] or different power dynamics and approaches [62,66]. Additionally, new models for innovation are being adopted by different sectors—including the public sector [67] or cities [68]. It is this change in dynamics, structure and power relationships that is leading to the nascent creation of innovation that is seeking to deliver economic, social and environmental enhancements at the same time [19].

#### 4.1.3. Evolution of Intermediaries for Innovation

This SLR reveals an evolution in the ecosystem of intermediaries for innovation. Intermediaries for innovation are individuals or organizations whose role it is to span the boundaries between organizations to facilitate innovation [69]. Innovation intermediaries are evolving from a simple partnership model of a technology transfer officer, through to the development of science and technology parks (STPs) and through to living labs and smart cities. This is a non-linear pathway [60], partly due to the rapid change in the nature of the knowledge-based economy—where knowledge is increasingly shared rather than owned [67], partly as a reflection of a change in knowledge production [16] and partly due to an increased focus (particularly in industrial economies) on the importance of the service-based economy, where service (experiential or simply more tailored [46]) is seen as another key to unlock economic development.

#### On Campus Structures

On-campus structures includes the creation of several organizational structures within universities to promote the development of partnerships for innovation. These forms of partnership are the simplest and are an organizational response to facilitate the creation of an increasingly entrepreneurial university [60]. In the initial phases at least, this is conducted on campus.

The structures put in place range from technology transfer offices [69], academic liaison officers [56] to act as an intermediary between the university and business, processes to facilitate access to library information for business [52] and the creation of incubators to help start-ups grow into functioning businesses [70]. The purpose of these mechanisms is to assist the transfer of knowledge from the researcher to the consumer via a business.

There is also some evidence in this SLR of the campus itself being used as an innovation, and these were underpinned by a planning perspective [71], asset management [72], by opening up the campus [73] or using the campus to drive radical innovation through institutionalization [74]. These studies show the potential for the entrepreneurial university to drive all three modes (education, research and commercialization) through the operation or development of the campus.

#### Development of Campus-Adjacent Structures

The second significant phase in the development of intermediaries for innovation is the creation of campus-adjacent structures to further the partnership between universities and business. In this SLR, there were 28 case studies looking at STPs (see Section 3.1.5). The impetus for the creation of these off-site structures seems to be university, or government driven, but there are examples of it being driven by the private real estate sector [32]. In this SLR, most papers considering STPs focused on a traditional form of STP, where a university creates spaces for businesses to occupy to deliver innovative goods and services (ideally based on or related to the intellectual output of the university).

These campus-adjacent structures have a complex nomenclature, but in this SLR, they are referred to as science and technology parks (STP). This is a generic term to take in research parks, technology parks, innovation parks and business parks. The key definition issue is that they are developed to create an environment conducive to the co-creation of economic value by business, ideally using university created knowledge. They are geographically proximate to the university and tend to focus on the research strengths of the university.

The value of geographical proximity is much debated [75]. This SLR showed strong informal connections between universities and business [52,76–78] based on geography but with less evidence of formal connections that deliver innovation based on universitycreated knowledge [75,79]. One study found that 92% of the on-park research and technology output was through private industry [80], with others considering the role of private capital to innovation success [50], the role of university finance to spin-off success [81] or the role of management [82], or the network benefit [83]. This does not, in itself, mean that STPs represent a failed policy, but that there is not strong evidence for the successful transference of knowledge from creator to consumer via a business based in the university's STP. The depth of these relationships depends upon the level of service offered by the university to its tenants—with non-core assistance (for example, human resource management functions) being valued by tenants [77] or the value of social capital to start-up success [49]. There is also a stream of work researching the connection between the university and the STP covering the role of knowledge transfer facilitated by librarians [56], the influence of the university on the STP [84], the impact of doctoral education [44] or a more holistic consideration of the STP compared to a technology transfer officer or other intermediaries (see Section 4.1.3) [69].

Although the usefulness of STPs is still subject to debate, the creation of STPs has been adopted in Europe [7,9,41,70,79,80,85–88] and North America [1], and STPs are widely emulated in the former Eastern bloc countries [11,89,90], as well as the centralized economies of China [2,55,91], Taiwan [92,93], Malaysia [53] and others in Asia; the creation of STPs is also seen as a pathway for economic development in developing nations [8,94–96] as well as being subject to international comparisons [4,34,38,47].

#### Development of Living Labs

The next phase in the evolution of intermediaries for innovation is the creation of living labs. As shown in Section 3.1.5, there is a significant body of work which investigates the role of living labs. These are partnership structures that are focused on user engagement and open innovation. The partners are varied but generally involve university, business, and government (at some level). Living labs (and derivatives) are driven by a desire to innovate within the partnership and this might be the deepening of research findings [97], creating a product or service to commercialize the research [98] or co-creating a new product or service [99].

The external change that is facilitating the development of living labs is the ability for a range of stakeholders to become freely involved in the process of innovation [100]. The service-dominant logic [48], open innovation [101], user innovation [29], user-centered design [102] or even social (rather than economic) innovation [103] have become possible due to the ability to create communities of interest for almost anyone.

Living labs (and derivatives) are widely debated in the literature and are normally considered a network that incorporates both user engagement and open innovation [35]; they have the characteristics as set out in Table 15. There are several forms of living labs, which are also evolving. Sustainability labs [104] are focused on the delivery of economic, social and environmental outcomes at a geographic location. Smart cities are developed as a higher systems level solution under which living labs enable the demonstration and prototyping of products and services. Urban living labs are a network structure within an urban environment [64].

Characteristic	Explanation		
Real life environments	Real life experimentation to test, develop, research new products, services, systems, processes		
Stakeholders	Range of partner involvement to co-create. Stakeholders are key to the outputs of the living lab		
Activities	What the living lab will focus upon. This is defined by whoever is driving the innovation (and is key to delivery of the output/outcome)		
Business models	Covers how the living lab will operate (essentially why it exists and how it will continue to exist)		
Methods and tools	The approach taken to innovation		
Challenges	Economic, social and/or environmental		
Output and/or outcomes	What the living lab delivers		
Sustainability	Emergence of innovation that moves society toward delivery of the sustainable development goals		

#### Table 15. Living lab characteristics [29].

However, both the literature and practitioners struggle to define living labs and their derivatives [29,64], or to create business cases to build them [65,104], or even best practice guides to help manage them [31]. They are a rapidly evolving creation that, in many respects, is a direct expression of the partnership that created them [104]. That said, there are structures to suit different desired outcomes, such as wicked issues [105] or radical innovation [74], and they are grouped into a genus containing 4 typologies characterized by open innovation: utilizer driven, enabler driven, provider driven, and user driven [35].

At the heart of living labs are two key elements: user engagement and open innovation [35]. These two aspects are evident in the case studies in this SLR. It is these two aspects that stand them apart as intermediaries for innovation. Because of this commitment facilitated by the knowledge economy and technological developments—living labs are footloose. They can be on campus [59,71,73,106–109], off campus [9], in an STP [48], on a high street [110], local [65,90,97,111,112], precinct scale [113], urban [100,114–118], suburban [40,66], rural [15], regional [101,105], peripheral [119] or city scale [68,99,102,116,120–123]. They can also be virtual [124].

It is partly this footloose, open and creative element that means that they are potentially difficult to harness at scale: indeed, difficult to harness by policy makers, but also difficult to harness by businesses, universities and the public. These structures are innovative in themselves; each is unique (even with common elements) and each is designed to serve a purpose. Their amorphous shape and shifting nature make them difficult to grasp and initiate at scale. Whilst STPs could be created by policy diktat [76], living labs cannot and as such are more ephemeral and can be a conundrum to universities, business and government. This transformational change of the modus operandii means that a once linear, or apparently linear evolution [60], is now beset by new branches and new forms (such as sustainability labs, urban living labs, and smart cities). These branches and forms are being created at such a pace that the literature is struggling to define them [64], or adequately develop theories to help amortize their existence [125].

#### • Living Labs and Sustainable Development

In this SLR, living labs (and derivatives) are the partnership structure that is being used successfully to drive social and economic development [62,126]. It is also the structure that is used in the limited number of studies that are using innovation to drive the delivery of sustainable development [19,29,40,73,96,107,109,115,127–132], with the emphasis on both sustainability labs and urban living labs. The literature does not provide guidance for the reasons for this. In a time when the Sustainable Development Goals have been unanimously agreed by the United Nations, it is noteworthy that the literature around developing partnerships for innovation is largely silent on the implications for innovation (an issue also noted by others [29]).

#### 4.2. Attributional Data

As discussed in Section 2, outlined in Table 16, and as defined by Massey (2011), but amended here to meet the needs of an SLR, attributional data relate to comments and discussion about a priori hypotheses or theories that the evaluator brings to the discussion. The data collected are the result of author expertise and assessment, as, in most cases, the theory that underpinned each study went unstated in the study.

Theme	Theory	Sub-Theory
		Open innovation theory [3,35-37,58,62,64,67,69,97,101,113,118,125,137-140]
		Innovation management theory [100,114,132]
	Innovation theory	User innovation theory [141,142]
	[7,11,15,46,49,53,54,60,62,63,70,78,90,94,110,123,133–136]	Collaborative knowledge production [108,118]
		Service or product dominant [48,66,72,74]
		Frugal innovation [133]
		Knowledge transfer theory [52]
	Growth theory [38,82,102]	Knowledge spill-over theory of entrepreneurship [81]
Economic development [1,8,10,57,76]		Development economics [143]
Economic geography [75,96]	Regional development [54,68,83,92,93,106,109,119,120,144]	Agglomeration economics [55,145]
		Business design concepts [33]
		Business excellence/total quality management [31]
	Management theory [4,42,50,79]	Construction management [146]
		Corporate real estate management theory [32]
		New public management theory [41]
	Socio-institutional economics [119]	
	New institutionalism [89]	
	Neo-institutional economics [116]	
		Business network theory [102]
	Network theories [34,35,47,112,147,148]	Actor network theory [149]
Systems Theory [11,91,150,151]	Self-organizing systems [71]	
	Socio technical Systems [73,107]	
	Process-based engineering [38]	

### Table 16. Categorization and paper breakdown of theories underpinning SLR.

.

Theme	Theory	Sub-Theory		
		Urban sustainability transition [99,121]		
		Transitions theory (sustainability) [45,96,108,117,129,131]		
Planning [124,125]	Iransition theory [39,88,92,117]	Transition management [107]		
-		Value of sustainable development [40,129,131]		
_	Design theory [104]	Academic capitalism [71,150]		
	Social practice theory [61,104,109]			
_	Social capital theory [40,49]			
– Social theories	Social network analysis [9,80,84-87]			
_	Social entrepreneurship [103]			
_	Social institutionalism [51]			
	Interorganizational learning [44,52]			
_	Experiential learning [56,127]			
Theories of learning –	Informed learning [124]			
	Social learning [98]			
_	Audit-based learning [59]			
-	Absorptive capacity [43,77]			

The selected papers in this SLR were underpinned by 28 different theories (as detailed in Table 16). The theories supporting the research reveal three intersecting themes which were categorized as economic development, social theories and a thinner vein on theories of learning.

Most studies have an economic theoretical underpinning (see Table 16 for a detailed disposition of the papers and their theoretical underpinning) developed through theories of innovation, economic geography, planning and transitions.

Aligned with economic development is a suite of papers dealing with social theories. This encapsulates both how society develops, but also how individuals interact with partnerships. To some degree, this is the practical element in the development of the papers, as it focuses the papers on the theory of how individuals in society interact with innovation.

The final theoretical category is around theories of learning. This is a shallower vein of research that links through to economic development and social theories but can be divided into two theoretical strands. One is how, particularly, (though not exclusively) universities can use innovation to help deliver learning to their students. The other strand relates to continuous improvement and considers how organizations (individually and collectively) can retain and improve upon their learning by doing.

#### 4.3. Emergent Data

This section of the SLR deals with emergent data. For the purposes of this SLR, emergent data were defined as information that contributes to new insights and hypothesis formulation and are the unanticipated product of connections arising from different articles. They are therefore the result of the consideration of the articulated and attributional data. They are not necessarily supported by an extant body of work, but they do highlight the areas where further research, particularly in the development of theoretical underpinnings is recommended. There are two emerging themes from this SLR that warrant further consideration.

The first relates to the evolution of partnerships for innovation and the potential relationship with globalization of the economy, society and environment. In the literature reviewed for this SLR, the development of partnerships for innovation is depicted as a process largely driven by decisions within the partnership. This SLR has proposed an evolution in the development of partnerships for innovation. However, emerging from this SLR is a nascent, but to date unresearched, relationship between the evolution of partnerships for innovation, the increasing globalization of the economy, society and environment and the development of modes of knowledge production.

The evolution of partnerships for innovation was discussed in Section 4.1.1. The increased globalization of the knowledge economy is a driver of the development of partnerships at the university–industry–government nexus. However, the globalization of society, through widespread adoption of the internet, also appears to be influencing the development of partnerships for innovation (as reflected through work on the quadruple helix [151] as well as wider moves toward open science [152]).

There is also some evidence of the impact of the globalization of the environment in the development of intermediaries for innovation. This SLR revealed a thin vein of work (as did [29]) looking at the use of partnerships for innovation to deliver sustainable development. The link between sustainable development and one form of partnerships for innovation exists, but there is currently little in the literature to provide an understanding of why living labs are the preferred partnership structure to deliver sustainable innovation.

As the same time, there is some discussion of the evolution of modes of knowledge production. Mode 1 is discipline based and produces theoretical knowledge; Mode 2 is transdisciplinary and is characterized by being applied research [14]; Mode 3 is a transdisciplinary ecosystem to enable people, culture and technology to interact across scientific and technological disciplines [151].

In this SLR, no paper explicitly addressed the relationship between the changing external landscape and the evolution of partnerships for innovation, but it would appear there is some form of relationship, and this emerging hypothesis is presented in Table 17. The hypothesis is based on the delivery of the three elements of sustainable development (economic growth, social development and environmental quality) at the same time rather than in isolation from each other. It is this balancing and integration of economic, social and environmental factors that seems to distinguish the living lab approach as the preferred partnership structure used to deliver sustainable development from the other forms of partnership (as discussed in Section 4.1.3).

Table 17.	Hypothesis of	current relationship	between	n partnerships	, globalizatior	n and mo	des of l	knowle	dge	production
-----------	---------------	----------------------	---------	----------------	-----------------	----------	----------	--------	-----	------------

Evolution of Partnerships for Innovation	Potential Partnership Response to Trends in Globalization			Changes in Knowledge Production
On campus				Modes 1 and 2
Campus adjacent	Economy	Economy		Modes 2 and 3
Living labs	_	Society	Environment	Mode 3

The second theme relates to the discussion in the papers of the triple helix [60], which are driven from a primarily economic perspective (see Section 4.2). The inclusion of a quadruple helix to include the public or citizens did take the model into a more clearly defined social territory, but the debate focused on driving the public perspective in innovation.

The papers reviewed illustrate that only living labs deal with sustainable development. The question remains unanswered as to how society can be confident that the framework driving innovation protects the interests of the global citizen. The interest of the global citizen is encompassed at the strategic level by the sustainable development goals. The relatively light body of research, revealed in [29] and in this SLR, on how to utilize partnerships for innovation to deliver the SDGs reveals a gap in the research literature.

#### 5. Conclusions and Further Research

This SLR has described an evolutionary approach to understanding the development of partnerships at the university-industry-government nexus. The evolution of the partnership structures begins with structures on campus; it then evolves into structures adjacent to campuses and finally to the development of living labs (and derivatives), which are footloose and found in a variety of locations. One reason for the evolution of living labs is the ability to create communities of interest for almost anyone. This evolutionary approach builds on the top-down/bottom-up approach currently described in the literature (see Figure 5). This insight into the evolution of partnerships for innovation is important from a theoretical and practical perspective.

From a theoretical perspective, the evolution in partnerships for innovation opens up the question of, what is driving the evolution. As discussed in Section 4.3, the partnerships for innovation are evolving at the same time as there is globalization of the economy, society and environment and a progressive change in the modes of knowledge production. This insight is helpful in understanding the evolution of partnerships for innovation at the university–industry–government nexus by developing an understanding of how factors external to the partnership may be affecting the evolution of the partnership structures. This finding demonstrates that partnerships for innovation sit within a wider innovation ecosystem that is also evolving.

The SLR also revealed that the partnership structures are evolving at a pace where the literature struggles to define them or develop theories as to their existence. There would be value in understanding the relationship, if any, between these processes to help further understand the factors that are driving the changes in partnerships for innovation.

Practitioners will also benefit from the new understanding of the evolution of partnerships for innovation into increasingly complex structures. Practitioners will be better able to understand how partnerships for innovation are changing, which may provide clearer insights into the most appropriate structure for what they are seeking to achieve which will, in turn, assist them in learning from experience rather than solely by doing.

This SLR also revealed that the predominant focus of the partnerships is economic development underpinned through theories of innovation, economic geography, planning and transitions. There is a stream of research that is focused on economic and social development, and a more limited number of research papers deal with sustainable development (this finding is also reflected elsewhere, for example, in [29]).

In this SLR, the only partnership structure that sought to further sustainable development was the living lab approach (and specifically sustainability labs and urban living labs). This SLR did not reveal the reasons why this is the case, and further research needs to be undertaken to understand why living labs might be considered the preferred structure to promote economic, social and environmental innovation. Further research is also needed to understand the barriers that are standing in the way of the other structures (on campus and campus adjacent) being used to further sustainable development objectives. The literature reviewed in this SLR is silent on the implications of the unanimously adopted Sustainable Development Goals on the development of partnerships for innovation at the university-industry-government nexus, and this gap in the research needs to be filled.

Author Contributions: Conceptualization, M.B. and G.M.M.; methodology, M.B. and G.M.M.; validation, M.B. and G.M.M.; formal analysis, M.B.; investigation, M.B.; resources, M.B.; data curation, M.B.; writing—original draft preparation, M.B.; writing—review and editing, M.B. and G.M.M.; visualization, M.B.; supervision, G.M.M.; project administration, M.B. Both authors have read and agreed to the published version of the manuscript.

**Funding:** This research is funded by the CRC (Project number NP 2001 for Low Carbon Living Ltd. supported by the Cooperative Research Centers program, an Australian Government initiative).

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- Mitchell, G.A. Research parks: Instrument, or harbinger of a new university paradigm? *Interchange* 1992, 23, 99–104. [CrossRef]
   Zhou, C. Emergence of the entrepreneurial university in evolution of the triple helix: The case of Northeastern University in
- China. J. Technol. Manag. China 2008, 3, 109–126. [CrossRef]
  Van Geenhuizen, M. From Ivory Tower to Living Lab: Accelerating the Use of University Knowledge. Environ. Plan. C Gov. Policy 2013, 31, 1115–1132. [CrossRef]
- Olvera, C.; Pique, J.M.; Cortes, U.; Nemirovsky, M. Evaluating the Success of Companies at University Science Parks: Key Performance and Innovation Indicators; Springer International Publishing: Cham, Switzerland, 2019; pp. 77–92.
- Quintas, P.D.; Wield, D.; Massey, D. Academic-Industry Links and Innovation: Questioning the Science Park Model. *Technovation* 1992, 12, 161. [CrossRef]
- Berbegal-Mirabent, J.F.; Sabaté, F.; Cañabate, A. Brokering knowledge from universities to the marketplace: The role of knowledge transfer offices. *Manag. Decis.* 2012, 50, 1285–1307. [CrossRef]
- Tschanz, R.; Cristo, S.; Delgado, L.; Hiroz, V.; Jordan, M.; Kalt, R.; Mitchell, M.; Muller, N.; Roeoesli, C.; Tamburello, V.; et al. "No Innovation Without Cooperation"—How Switzerland Innovation Promotes Cooperation between Industry, Research and Startups. *Chimia* 2020, 74, 755–757. [CrossRef] [PubMed]
- Sanni, M.; Egbetokun, A.A.; Siyanbola, W.O. A model for the design and development of a Science and Technology Park in developing countries. Int. J. Manag. Enterp. Dev. 2009, 8, 62–81. [CrossRef]
- Van der Sijde, P.; Vogelaar, G.; Hoogeveen, A.; Ligtenberg, H.; van Velzen, M. Attracting high-tech companies: The case of the University of Twente and its region. Ind. High. Educ. 2002, 16, 97–104. [CrossRef]
- Wonglimpiyarat, J. The innovation incubator, university business incubator and technology transfer strategy: The case of Thailand. Technol. Soc. 2016, 46, 18–27. [CrossRef]
- Bekniyazova, D.S.; Nurgaliyeva, A.; Korabayev, B.; Dyussembekova, G.; Altybassarova, M.; Alkeyev, M. Innovation activity in the Republic of Kazakhstan: State controlling and ways to increase management efficiency. J. Internet Bank. Commer. 2016, 53, 1–14.
- Dhewanto, W.; Lantu, D.C.; Herliana, S.; Permatasari, A. The obstacles for science technology parks in a developing country. Int. J. Technol. Learn. Innov. Dev. 2016, 8, 4–19. [CrossRef]
- 13. Etzkowitz, H.; Leydesdorff, L. The Triple Helix—University-Industry-Government Relations: A Laboratory for Knowledge Based Economic Development. *EASST Rev.* **1995**, *14*, 14–19.
- Etzkowitz, H.; Leydesdorff, L. The dynamics of innovation: From National Systems and "Mode 2" to a Triple Helix of universityindustry-government relations. Res. Policy 2000, 29, 109–123. [CrossRef]
- Guzman, J.G.; Schaffers, H.; Bilicki, V.; Merz, C.; Valenzuela, M. Living labs fostering open innovation and rural development: Methodology and results. In Proceedings of the 2008 IEEE International Technology Management Conference (ICE 2008), Lisbon, Portugal, 23–28 June 2008.
- Carayannis, E.G.; Campbell, D.F.J. 'Mode 3' and 'Quadruple Helix': Toward a 21st century fractal innovation ecosystem. Int. J. Technol. Manag. 2009, 46, 201–234. [CrossRef]
- 17. Meadows, D.H.; Meadows, D.L.; Randers, J.; Behrens, W.W., III (Eds.) The Limits to Growth: A Report for the Club of Rome's Project on the Predicament of Mankind; Universe Books: New York, NY, USA, 1972.
- Carayannis, E.G.; Barth, T.D.; Campbell, D.F.J. The Quintuple Helix innovation model: Global warming as a challenge and driver for innovation. J. Innov. Entrep. 2012, 1, 2. [CrossRef]
- Burbridge, M. If Living Labs Are the Answer—What's the Question? A Review of the Literature. In International High-Performance Built Environment Conference—A Sustainable Built Environment Conference 2016 Series; Ding, L., Fiorito, F., Osmond, P., Eds.; Springer: Berlin/Heidelberg, Germany, 2017; pp. 1725–1732.
- 20. Australia Government. National Innovation and Science Agenda; Australia Government: Canberra, Australia, 2015.
- 21. OECD. Collaboration on Innovation. In OECD Science, Technology and Industry Scoreboard 2015: Innovation for Growth and Society; OECD Publishing: Paris, France, 2015.
- 22. Petticrew, M.; Roberts, H. Systematic Reviews in the Social Sciences: A Practical Guide; Blackwell Pub.: Malden, MA, USA, 2005.
- Brien, A.M.A. The Systematic Literature Review Method: Trials and Tribulations of Electronic Database Searching at Doctoral Level; O'Brien, A.M., Mc Guckin, C., Eds.; SAGE: London, UK, 2016.
- De Medeiros, J.F.; Ribeiro, J.L.D.; Cortimiglia, M.N. Success factors for environmentally sustainable product innovation: A systematic literature review. J. Clean. Prod. 2014, 65, 76–86. [CrossRef]
- Massey, O.T. A proposed model for the analysis and interpretation of focus groups in evaluation research. *Eval. Program Plan.* 2011, 34, 21–28. [CrossRef]
- Moher, D.; Liberati, A.; Tetzlaff, J.; Altman, D. Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. J. Clin. Epidemiol. 2009, 62, 1006–1012. [CrossRef]
- Leal Filho, W.; Tripathi, S.K.; Andrade Guerra, J.B.; Gine-Garriga, R.; Orlovic Lovren, V.; Willats, J. Using the sustainable development goals towards a better understanding of sustainability challenges. Int. J. Sustain. Dev. World Ecol. 2019, 26, 179–190. [CrossRef]
- Liberati, A.; Altman, D.; Tetzlaff, J.; Mulrow, C.; Gotzsche, P.; Ioannidis, J.; Clarke, M.; Devereaux, P.; Kleijnen, J.; Moher, D. The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate healthcare interventions: Explanation and elaboration. *BMJ* 2009, 339, b2700. [CrossRef]
- Hossain, M.; Leminen, S.; Westerlund, M. A systematic review of living lab literature. J. Clean. Prod. 2019, 213, 976–988. [CrossRef]
   Cai, Y.; Etzkowitz, H. Theorizing the Triple Helix model: Past, present, and future. Triple Helix J. 2020, 7, 189–226. [CrossRef]
- Cal, F., Elzkowitz, H. Heofizing the high refix flodel. Fast, present, and future. *Inple Helk J.* 2020, 7 169–220. [CossRef]
   Katzy, B. Designing Viable Business Models for Living Labs. *Technol. Innov. Manag. Rev.* 2012, 2, 19–24. [CrossRef]
- Magdaniel, F.C.C.; de Jonge, H.; den Heijer, A. Campus development as catalyst for innovation. J. Corp. Real Estate 2018, 20, 84–102. [CrossRef]
- Schaffers, H.; Cordoba, M.G.; Hongisto, P.; Kallai, T.; Merz, C.; Van Rensburg, J. Exploring Business Models for Open Innovation in Rural Living Labs. In Proceedings of the 2007 IEEE International Technology Management Conference (ICE), Sophia Antipolis, France, 4–6 June 2007; pp. 1–8. Available online: https://ieeexplore.ieee.org/document/7458702 (accessed on 9 December 2021).
- Schoonmaker, M.G.; Carayannis, E.G. Mode 3: A Proposed Classification Scheme for the Knowledge Economy and Society. J. Knowl. Econ. 2013, 4, 556–577. [CrossRef]
- Leminen, S.; Westerlund, M.; Nyström, A.-G. Living Labs as Open-Innovation Networks. Technol. Innov. Manag. Rev. 2012, 2, 6–11. [CrossRef]
- 36. Leminen, S. Coordination and Participation in Living Lab Networks. Technol. Innov. Manag. Rev. 2013, 3, 5–14. [CrossRef]
- Leminen, S.; Nyström, A.-G.; Westerlund, M.; Kortelainen, M.J. The effect of network structure on radical innovation in living labs. J. Bus. Ind. Mark. 2016, 31, 743–757. [CrossRef]
- Aslani, A.; Eftekhari, H.; Didari, M. Comparative Analysis of the Science and Technology Parks of the US Universities and a Selected Developing Country. In Proceedings of the 11th International Conference on Innovation and Management, Vaasa, Finland, 17–19 November 2014; pp. 1–7.
- Purcell, W.M.; Henriksen, H.; Spengler, J.D. Universities as the Engine of Transformational Sustainability toward Delivering the Sustainable Development Goals: "Living Labs" for Sustainability. Int. J. Sustain. High. Educ. 2019, 20, 1343–1357. [CrossRef]
- Buhr, K.; Federley, M.; Karlsson, A. Urban Living Labs for Sustainability in Suburbs in Need of Modernization and Social Uplift. Technol. Innov. Manag. Rev. 2016, 6, 27–34. [CrossRef]
- 41. Nielsen, C. Getting value for money from your science park. Public Money Manag. 2016, 36, 539. [CrossRef]
- 42. Veeckman, C.; Schuurman, D.; Leminen, S.; Westerlund, M. Linking Living Lab Characteristics and Their Outcomes: Towards a Conceptual Framework. *Technol. Innov. Manag. Rev.* 2013, 3, 6–15. [CrossRef]

- 43. Martins, J.T. Relational capabilities to leverage new knowledge: Managing directors' perceptions in UK and Portugal old industrial regions. *Learn. Organ.* 2016, 23, 398–414. [CrossRef]
- 44. Germain-Alamartine, E.; Moghadam-Saman, S. Aligning doctoral education with local industrial employers' needs: A comparative case study. *Eur. Plan. Stud.* 2020, 28, 234–254. [CrossRef]
- Voytenko, Y.; McCormick, K.; Evans, J.; Schliwa, G. Urban living labs for sustainability and low carbon cities in Europe: Towards a research agenda. J. Clean. Prod. 2016, 123, 45–54. [CrossRef]
- Almirall, E., Lee, M.; Wareham, J. Mapping Living Labs in the Landscape of Innovation Methodologies. *Technol. Innov. Manag. Rev.* 2012, 2, 12–18. [CrossRef]
- Yun, S.; Lee, J. An innovation network analysis of science clusters in South Korea and Taiwan. Asian J. Technol. Innov. 2013, 21, 277–289. [CrossRef]
- Ådahl, M. Commercial consortia. In Living Labs: Design and Assessment of Sustainable Living; Springer: Cham, Switzerland, 2016; pp. 385–390.
- Bandera, C.; Thomas, E. The Role of Innovation Ecosystems and Social Capital in Startup Survival. IEEE Trans. Eng. Manag. 2019, 66, 542–551. [CrossRef]
- 50. Campanella, F.; Peruta, M.R.D.; del Giudice, M. Creating conditions for innovative performance of science parks in europe. How manage the intellectual capital for converting knowledge into organizational action. J. Intellect. Cap. 2014, 15, 576–596. [CrossRef]
- Hladchenko, M.; Pinheiro, R. Implementing the Triple Helix model: Means-ends decoupling at the state level? *Mineroa* 2019, *57*, 1–22. [CrossRef]
- Aportela-Rodriguez, I.M.; Pacios, A.R. University Libraries and Science and Technology Parks: Reasons for Collaboration. *Libri* 2017, 67, 235–244. [CrossRef]
- Malairaja, C.; Zawdie, G. Science parks and university-industry collaboration in Malaysia. *Technol. Anal. Strateg. Manag.* 2008, 20, 727–739. [CrossRef]
- 54. Etzkowitz, H.; Zhou, C. Innovation incommensurability and the science park. R D Manag. 2017, 48, 73–87. [CrossRef]
- Jongwanich, J.; Kohpaiboon, A.; Yang, C.-H. Science park, triple helix, and regional innovative capacity: Province-level evidence from China. J. Asia Pac. Econ. 2014, 19, 333–352. [CrossRef]
- Bakouros, Y.L.; Samara, E.T. Academic Liaison Offices vs. technology transfer units: Could they form a new joint mechanism towards the exploration of Academic/Research results? Int. J. Innov. Sci. 2010, 2, 145–157. [CrossRef]
- Parry, M. Science and Technology Parks and Universities—Facing the Next Industrial Revolution; Springer International Publishing: Cham, Switzerland, 2020; pp. 109–140.
- Kantola, T.; Hirvikoski, T.; Lehto, P.; Aholaakko, T.-J.; Kukkonen, M.-L.; Partamies, S. Towards Co-Creation of eHealth Services. Interdiscip. Stud. J. 2014, 3, 192.
- 59. Emblen-Perry, K. Auditing the University: Promoting Business Education for Sustainability through Audit-Based Learning; Springer International Publishing: Cham, Switzerland, 2019; pp. 303–322.
- 60. Etzkowitz, H. The new visible hand: An assisted linear model of science and innovation policy. *Sci. Public Policy* **2006**, *33*, 310–320. [CrossRef]
- 61. Ståhlbröst, A.; Holst, M. Reflecting on Actions in Living Lab Research. Technol. Innov. Manag. Rev. 2017, 7, 27–34. [CrossRef]
- De Silva, M.; Wright, M. Entrepreneurial co-creation: Societal impact through open innovation. R D Manag. 2019, 49, 318–342. [CrossRef]
- 63. Seppä, M. From Business Administration to Business Creation: The Case of the Kalevala Global Business Creation School. *Technol. Innov. Manag. Rev.* 2012, 2, 6–11. [CrossRef]
- Steen, K.Y.G.; Van Bueren, E.M. The Defining Characteristics of Urban Living Labs. Technol. Innov. Manag. Rev. 2017, 7, 21–33. [CrossRef]
- 65. Schaffers, H.; Turkama, P. Living Labs for Cross-Border Systemic Innovation. Technol. Innov. Manag. Rev. 2012, 2, 25–30. [CrossRef]
- 66. Ayvari, A.; Jyrama, A. Rethinking value proposition tools for living labs. J. Serv. Theory Pract. 2017, 27, 1024–1039. [CrossRef]
- 67. Gasco, M. Living labs: Implementing open innovation in the public sector. Gov. Inf. Q. 2017, 34, 90–98. [CrossRef]
- Lukkari, O.; Mustonen, J.; Tuikka, T. City of Oulu as an innovative service platform. In Proceedings of the 2008 IEEE International Technology Management Conference (ICE 2008), Lisbon, Portugal, 23 June–25 June 2008; pp. 1–4.
- Fernández-Esquinas, M.; Merchán-Hernández, C.; Valmaseda-Andía, O. How effective are interface organizations in the promotion of university-industry links? Evidence from a regional innovation system. *Eur. J. Innov. Manag.* 2016, 19, 424–442. [CrossRef]
- McAdam, M.; McAdam, R. The networked incubator: The role and operation of entrepreneurial networking with the university science park incubator (USI). Int. J. Entrep. Innov. 2006, 7, 87–98. [CrossRef]
- Bergquist, D.; Hempel, C.A.; Green, J.L. Bridging the Gap between Theory and Design: A Proposal for Regenerative Campus Development at the Swedish University of Agricultural Sciences. Int. J. Sustain. High. Educ. 2019, 20, 548–567. [CrossRef]
- Ahlava, A.; Suominen, J.; Rossi, S. Controlling Risks Through Flexibility and Urban Integration: The Regeneration of Otaniemi Campus in Finland. In *Handbook of Theory and Practice of Sustainable Development in Higher Education*; Springer International Publishing: Cham, Switzerland, 2016; pp. 21–35.
- Bracco, S.; Delfino, F.; Laiolo, P.; Morini, A. Planning & open-air demonstrating smart city sustainable districts. Sustainability 2018, 10, 4636.

- Ventura, R.; Quero, M.J.; Díaz-Méndez, M. The role of institutions in achieving radical innovation. Mark. Intell. Plan. 2019, 38, 310–324. [CrossRef]
- Vedovello, C. Science parks and university -Industry interaction: Geographical proximity between the agents as a driving force. Technovation 1997, 17, 491–502. [CrossRef]
- Motohashi, K. The role of the science park in innovation performance of start-up firms: An empirical analysis of Tsinghua Science Park in Beijing. Asia Pac. Bus. Rev. 2013, 19, 578–599. [CrossRef]
- van Oostrom, M.; Pedraza-Rodríguez, J.A.; Fernández-Esquinas, M. Does the Location in a Science and Technology Park Influence University—Industry Relationships?: Evidence From a Peripheral Region. Int. J. Knowl. Manag. 2019, 15, 66–82. [CrossRef]

78. Hung, W.C. Measuring the use of public research in firm R&D in the Hsinchu Science Park. *Scientometrics* **2012**, *92*, 63–73.

- Ramírez-Alesón, M.; Fernández-Olmos, M. Unravelling the effects of Science Parks on the innovation performance of NTBFs. J. Technol. Transf. 2018, 43, 482–505. [CrossRef]
- Minguillo, D.; Thelwall, M. Research excellence and university-industry collaboration in UK science parks. *Res. Eval.* 2015, 24, 181–196. [CrossRef]
- Corsi, C.; Prencipe, A. Improving innovation in University Spin-offs. The fostering role of university and region. J. Technol. Manag. Innov. 2016, 11, 13–21.
- Feldman, J.M. The Managerial Equation and Innovation Platforms: The Case of Linkoand Berzelius Science Park. Eur. Plan. Stud. 2007, 15, 1027–1045. [CrossRef]
- 83. Latorre-Martínez, M.P.; Navarro, E.; Pastor, T.; Iniguez-Berrozpe, T. Analysis of the network of relations of organizations set up at walqa technology park. In *Lecture Notes in Mechanical Engineering*; Springer: Cham, Switzerland, 2013; pp. 1335–1346.
- Farre-Perdiguer, M.; Sala-Rios, M.; Torres-Sole, T. Network analysis for the study of technological collaboration in spaces for innovation. Science and technology parks and their relationship with the university. Int. J. Educ. Technol. High. Educ. 2016, 13, 8. [CrossRef]
- 85. Minguillo, D.; Thelwall, M. Mapping the network structure of science parks An exploratory study of cross-sectoral interactions reflected on the web. *Aslib Proc.* 2012, *64*, 332–357. [CrossRef]
- Minguillo, D.; Thelwall, M. Which are the best innovation support infrastructures for universities? Evidence from R&D output and commercial activities. *Scientometrics* 2015, 102, 1057–1081.
- Minguillo, D.; Tijssen, R.; Thelwall, M. Do science parks promote research and technology? A scientometric analysis of the UK. Scientometrics 2015, 102, 701–725. [CrossRef]
- Steinthorsson, R.S.; Hilmarsson, E.; Janusson, H.B. Towards openness and inclusiveness: The evolution of a science park. Ind. High. Educ. 2017, 31, 388–398. [CrossRef]
- Karpińska, A. Innovation and science dilemmas. Unintended consequences of innovation policy for science. Polish experience. Cogent Soc. Sci. 2020, 6, 1718055. [CrossRef]
- Prepelita-Raileanu, B.; Pastae, O.M. Bridging the gap between higher education, academic research and Romanian business community. In Proceedings of the 9th WSEAS Conference on Education and Education Technology, Kuala Lumpur, Malaysia, 4 October 2010; pp. 4–6.
- Yan, M.-R.; Yan, H.; Zhan, L.; Yan, X.; Xu, M. Evaluation of Technological Innovations and the Industrial Ecosystem of Science Parks in Shanghai: An Empirical Study. Sci. Technol. Soc. 2020, 25, 482–504. [CrossRef]
- Tsai, C.-L.; Chang, H.-C. Evaluation of critical factors for the regional innovation system within the HsinChu science-based park. Kybernetes 2016, 45, 699–716. [CrossRef]
- Wang, W.-B.; Hung, Y.-C.; Wang, C.-C. University-Industry Research Collaboration in Taiwan. J. Inf. Optim. Sci. 2012, 33, 665–683. [CrossRef]
- Widiawan, K. Identifying the most suitable university-industry partnership model in developing countries. In Proceedings of the 2008 4th IEEE International Conference on Management of Innovation and Technology, Bangkok, Thailand, 21–24 September 2008; pp. 128–133.
- Melo, P.; Maravilhas, S. Are interaction linkages based on geographic proximity focused on development of firms innovation activities still relevant? Empirical study of the ICT firms located in the city of Salvador, Bahia, Brazil. RISUS Rev. Inovação E Sustentabilidade 2019, 10, 12–19.
- Benltoufa, A.N.H.S.; Noureddine, H.; Jaafar, F.; Maraoul, M.; Said, L.; Zili, M.; Hedfi, H.; Labidi, M.; Bouzidi, A.; Jrad, B.B.; et al. From smart campus to smart city: Monastir living lab. In Proceedings of the 2017 International Conference on Engineering and Technology (ICET), Antalya, Turkey, 21–23 August 2017.
- Martinez, S.; Bjerkas, S.; Ludvigsen, A.E.; Fensli, R. Agder Living Lab: From Ideas to Large-Scale Deployment and Long-Term User Adoption of Inclusive Health Solutions; Springer International Publishing: Cham, Switzerland, 2016; pp. 391–399.
- Hakkarainen, L.; Hyysalo, S. How Do We Keep the Living Laboratory Alive? Learning and Conflicts in Living Lab Collaboration. Technol. Innov. Manag. Rev. 2013, 3, 16–22. [CrossRef]
- Chronéer, D.; Ståhlbröst, A.; Habibipour, A. Urban Living Labs: Towards an Integrated Understanding of their Key Components. Technol. Innov. Manag. Rev. 2019, 9, 50–62. [CrossRef]
- Baccarne, B.; Mechant, P.; Schuurma, D.; De Marez, L.; Colpaert, P. Urban Socio-technical Innovations with and by Citizens. Interdiscip. Stud. J. 2014, 3, 143.

- Dhakal, S.; Mahmood, M.; Wiewora, A.; Brown, K.; Keast, R. The Innovation Potential of Living-Labs to Strengthen Small and Medium Enterprises in Regional Australia. *Australas. J. Reg. Stud.* 2013, 19, 456–474.
- Artto, K.; Kyro, R.; Ahola, T.; Peltokorpi, A.; Sandqvist, K. The Cuckoo's Nest Approach for Co-Creating Business Ecosystems in Smart Cities. Technol. Innov. Manag. Rev. 2016, 6, 26–37. [CrossRef]
- Lundström, A.; Zhou, C. Promoting innovation based on social sciences and technologies: The prospect of a social innovation park. *Innovation* 2011, 24, 133–149. [CrossRef]
- Burbridge, M.; Morrison, G.M.; van Rijn, M.; Silvester, S.; Keyson, D.V.; Baedeker, C.; Liedtke, C. Business models for sustainability in living labs. In Living Labs: Design and Assessment of Sustainable Living; Springer: Cham, Switzerland, 2016; pp. 391–403.
- 105. Zivkovic, S. Systemic innovation labs: A lab for wicked problems. Soc. Enterp. J. 2018, 14, 348–366. [CrossRef]
- 106. Mohamad, Z.F.; Kadir, S.N.A.; Nasaruddin, A.; Sakai, N.; Mohamed Zuki, F.; Hussein, H.; Sulaiman, A.-H.; Mohammad, S.A. Heartware as a driver for campus sustainability: Insights from an action-oriented exploratory case study. J. Clean. Prod. 2018, 196, 1086–1096. [CrossRef]
- 107. Berker, T.; Woods, R. Identifying and addressing reverse salients in infrastructural change. The case of a small zero emission campus in Southern Norway. *Int. J. Sustain. High. Educ.* 2020, 21, 1625–1640. [CrossRef]
- Cooper, L.; Gorman, D. A Holistic Approach to Embedding Social Responsibility and Sustainability in a University—Fostering Collaboration between Researchers, Students and Operations; Springer International Publishing: Cham, Switzerland, 2017; pp. 177–192.
- Fabregà, M.B. How Entrepreneurship in Higher Education Helps to Sustainable Development at the Local Level: The Case of Tecnocampus; Springer International Publishing: Cham, Switzerland, 2018; pp. 587–604.
- 110. van Winden, W.; Hagemans, I.; van Hemert, P. The Street-Wise University: The Amsterdam Knowledge Mile as an Intermediary and Place-Making Concept. Soc. Sci. 2019, 8, 229. [CrossRef]
- 111. Callaghan, R.; Herselman, M. Applying a Living Lab methodology to support innovation in education at a university in South Africa. J. Transdiscipl. Res. South. Afr. 2015, 11, e1–e18. [CrossRef]
- Mulvenna, M.D.; Bergvall-Kareborn, B.; Galbraith, B.; Wallace, J.; Martin, S. Living Labs Are Innovation Catalyst. In Innovation through Knowledge Transfer 2010; Howlett, R.J., Ed.; Springer: Cham, Switzerland, 2011; pp. 253–264.
- Punt, E.; Afrooz, A.; Pettit, C. Precinct scale living laboratories: Case study of Randwick living lab. ISPRS Ann. Photogramm. Remote Sens. Spat. Inf. Sci. 2020, VI-4/W2-2020, 127–134. [CrossRef]
- Baccarne, B.; Logghe, S.; Schuurman, D.; De Marez, L. Governing Quintuple Helix Innovation: Urban Living Labs and Socio-Ecological Entrepreneurship. *Technol. Innov. Manag. Rev.* 2016, 6, 22–30. [CrossRef]
- Boeri, A.; Longo, D.; Gianfrate, V.; Lorenzo, V. Resilient communities. Social infrastructures for sustainable growth of urban areas. A case study. Int. J. Sustain. Dev. Plan. 2017, 12, 227–237. [CrossRef]
- Canzler, W.; Engels, F.; Rogge, J.C.; Simon, D.; Wentland, A. From "living lab" to strategic action field: Bringing together energy, mobility, and Information Technology in Germany. *Energy Res. Soc. Sci.* 2017, 27, 25–35. [CrossRef]
- 117. Frantzeskaki, N.; van Steenbergen, F.; Stedman, R.C. Sense of place and experimentation in urban sustainability transitions: The Resilience Lab in Carnisse, Rotterdam, The Netherlands. *Sustain. Sci.* **2018**, *13*, 1045–1059. [CrossRef] [PubMed]
- Tukiainen, T.; Leminen, S.; Westerlund, M. Cities as Collaborative Innovation Platforms. *Technol. Innov. Manag. Rev.* 2015, 5, 16–23. [CrossRef]
- Antonopoulos, C.N.; Papadakis, V.G.; Stylios, C.D.; Efstathiou, M.P.; Groumpos, P.P. Mainstreaming innovation policy in less favoured regions: The case of Patras Science Park, Greece. Sci. Public Policy 2009, 36, 511–521. [CrossRef]
- Vallance, P.; Tewdwr-Jones, M.; Kempton, L. Building collaborative platforms for urban innovation: Newcastle City Futures as a quadruple helix intermediary. *Eur. Urban Reg. Stud.* 2020, 27, 325–341. [CrossRef]
- Bartelt, V.L.; Urbaczewski, A.; Mueller, A.G.; Sarker, S. Enabling collaboration and innovation in Denver's smart city through a living lab: A social capital perspective. *Eur. J. Inf. Syst.* 2020, 29, 369–387. [CrossRef]
- Coenen, T.; Robijt, S. Heading for a FALL: A Framework for Agile Living Lab Projects. Technol. Innov. Manag. Rev. 2017, 7, 37–43. [CrossRef]
- Lindberg, M.; Segerstedt, E.; Hidman, E.; Nilsson, K.; Karlberg, H.; Balogh, J. Co-Creative Place Innovation in an Arctic Town. J. Place Manag. Dev. 2020, 13, 447. [CrossRef]
- 124. Hughes, H.; Wolf, R.; Foth, M. Informed digital learning through social living labs as participatory methodology: The case of Food Rescue Townsville. *Inf. Learn. Sci.* 2017, 118, 518–534. [CrossRef]
- Leminen, S.; Rajahonka, M.; Westerlund, M. Towards Third-Generation Living Lab Networks in Cities. *Technol. Innov. Manag. Rev.* 2017, 7, 21–35. [CrossRef]
- 126. Dekker, R.; Contreras, J.F.; Meijer, A. The Living Lab as a Methodology for Public Administration Research: A Systematic Literature Review of its Applications in the Social Sciences. *Int. J. Public Adm.* 2020, *43*, 1207–1217. [CrossRef]
- Jernsand, E.M. Student living labs as innovation arenas for sustainable tourism. *Tour. Recreat. Res.* 2019, 44, 337–347. [CrossRef]
   Verhoef, L.A.; Bossert, M.; Newman, J.; Ferraz, F.; Robinson, Z.P.; Agarwala, Y.; Wolff, P.J.; Jiranek, P.; Hellinga, C. *Towards a Learning System for University Campuses as Living Labs for Sustainability*; Springer International Publishing: Cham, Switzerland, 2019; pp. 135–149.
- 129. Zen, I.S. Exploring the living learning laboratory: An approach to strengthen campus sustainability initiatives by using sustainability science approach. Int. J. Sustain. High. Educ. 2017, 18, 939–955. [CrossRef]

- Nifa, F.A.A.; Rahim, S.A.; Rani, W.N.M.; Ismail, M.N. Collaborative Procurement for Developing a Sustainable Campus; American Institute of Physics: Melville, NY, USA, 2016.
- da Silva, L.C.P.; Wright, M. Sustainable Campus Model at the University of Campinas—Brazil: An Integrated Living Lab for Renewable Generation, Electric Mobility, Energy Efficiency, Monitoring and Energy Demand Management; Springer International Publishing: Cham, Switzerland, 2018; pp. 457–472.
- Cigir, K. Creating a living lab model for tourism and hospitality businesses to stimulate CSR and sustainability innovations. WIT Trans. Ecol. Environ. 2018, 217, 569–583.
- Pisoni, A.; Michelini, L.; Martignoni, G. Frugal approach to innovation: State of the art and future perspectives. J. Clean. Prod. 2018, 171, 107–126. [CrossRef]
- Steruska, J.; Simkova, N.; Pitner, T. Do science and technology parks improve technology transfer? *Technol. Soc.* 2019, 59, 101127. [CrossRef]
- 135. Van Dierdonck, R.; Debackere, K.; Engelen, B. University-industry relationships: How does the Belgian academic community feel about it? *Res. Policy* **1990**, *19*, 551–566. [CrossRef]
- 136. Verloo, H.; Lorette, A.; Gomes da Rocha, C.; Amoussou, J.-R.; Gillès de Pélichy, E.; Matos Queiros, A.; Mendez Rubio, M.; von Gunten, A. A Comprehensive Scoping Review Protocol of Using Living Labs to Explore Needs and Solutions for Older Adults with Dementia. *Smart Homecare Technol. TeleHealth* 2020, 7, 19–27. [CrossRef]
- 137. Hagy, S.; Morrison, G.; Elfstrand, P. Co-Creation in Living Labs; Springer: Cham, Switzerland, 2016.
- Logghe, S.; Schuurman, D. Action Research as a Framework to Evaluate the Operations of a Living Lab. Technol. Innov. Manag. Rev. 2017, 7, 35–41. [CrossRef]
- Marasso, L.; Giangreco, E.; Storelli, D.; Chetta, V.; Camillò, A. Turrisi, G. Antonucci, G.; Barile, M.; Centrone, B.; Papadia, D.; Simone, F. Idea Management System for Smart City Planning. *Interdiscip. Stud. J.* 2014, 3, 227.
- Schuurman, D.; de Marez, L.; Ballon, P. Open Innovation Processes in Living Lab Innovation Systems: Insights from the LeYLab. Technol. Innov. Manag. Rev. 2013, 3, 28–36. [CrossRef]
- Veeckman, C.; van der Graaf, S. The City as Living Laboratory: Empowering Citizens with the Citadel Toolkit. Technol. Innov. Manag. Rev. 2015, 5, 6–17. [CrossRef]
- Westerlund, M.; Leminen, S.; Habib, C. Key Constructs and a Definition of Living Labs as Innovation Platforms. *Technol. Innov.* Manag. Rev. 2018, 8, 51–62. [CrossRef]
- 143. Kimatu, J.N. Evolution of strategic interactions from the triple to quad helix innovation models for sustainable development in the era of globalization. *J. Innov. Entrep.* 2016, *5*, 16. [CrossRef]
- 144. Charles, D. The rural university campus and support for rural innovation. Sci. Public Policy 2016, 43, 763–773. [CrossRef]
- 145. Kuo-Feng, H.; Chwo-Ming, Y.; Dah-Hsian, S. R&D Collaborations in a Cluster: An Empirical Study for the Taiwan's Hsinchu Science Park. *Tai Da Guan Li Lun Cong* 2010, 21, 47.
- 146. Hartkopf, V.; Loftness, V.; Mahdavi, A.; Lee, S.; Shankavaram, J. An integrated approach to design and engineering of intelligent buildings—The Intelligent Workplace at Carnegie Mellon University. *Autom. Constr.* **1997**, *6*, 401–415. [CrossRef]
- 147. Wehrmann, C.; van de Sanden, M.C.A. Universities as living labs for science communication. J. Sci. Commun. 2017, 16, C03. [CrossRef]
- Yan, M.-R.; Chien, K.-M.; Hong, L.-Y.; Yang, T.-N. Evaluating the collaborative ecosystem for an innovation-driven economy: A systems analysis and case study of science parks. Sustainability 2018, 10, 887. [CrossRef]
- 149. Sandrine, R.; Guillaume, G.; Philippe, V. Citizen Involvement in Local Environmental Governance: A Methodology Combining Human-Centred Design and Living lab Approaches. *Electron. J. E-Gov.* 2014, 12, 108.
- Hladchenko, M. Knowledge Valorisation: A Route of Knowledge That Ends In Surplus Value (An Example of The Netherlands). Int. J. Educ. Manag. 2016, 30, 668–678. [CrossRef]
- 151. Carayannis, E.G. Mode 3 Knowledge Production in Quadruple Helix Innovation Systems: 21st-Century Democracy, Innovation, and Entrepreneurship for Development; Elias, G.C., David, F.J.C., Eds.; Springer: New York, NY, USA, 2012.
- Lacey, J.; Coates, R.; Herington, M. Open science for responsible innovation in Australia: Understanding the expectations and priorities of scientists and researchers. J. Responsible Innov. 2020, 7, 427–449. [CrossRef]

### B.2 Publication 2

<u>Burbridge, M</u>., G Morrison, M. van Rijn, S. Silvester, D. Keyson, L. Virdee, & C. Baedeker
 (2016). Business models for sustainability in living labs. In: Keyson D., Guerra-Santin
 O., Lockton D. (eds) Living Labs. Springer, Cham. <u>https://doi.org/10.1007/978-3-319-33527-8_30</u>

# Chapter 30 Business Models for Sustainability in Living Labs

Mike Burbridge, Gregory M. Morrison, Menno van Rijn, Sasha Silvester, David V. Keyson, Lali Virdee, Carolin Baedeker and Christa Liedtke

Abstract There are an estimated 170 active living labs across the globe. All have common elements but not all of them contribute to the delivery of sustainable living. Here we consider the business models of sustainability in living labs (SusLabs). Specifically we review four active living laboratories that are part of the SusLab North West Europe network. We show that the business cases are different for at least two reasons. One is that each SusLab project has a specific focus even though all are seeking to develop energy efficient innovative products, services or systems. Examples of focus include demonstration projects, knowledge generation through research and business to business development. The other is

M. Burbridge e-mail: mike.burbridge@postgrad.curtin.edu.au

M. van Rijn Bax and Willems, Barcelona, Spain e-mail: m.vanrijn@baxwillems.eu

S. Silvester · D.V. Keyson Industrial Design, TU Delft, Delft, The Netherlands e-mail: S.Silvester@tudelft.nl

D.V. Keyson e-mail: d.v.keyson@tudelft.nl

L. Virdee Institute for Sustainable Futures, London, UK e-mail: lali.virdee@instituteforsustainability.org.uk

C. Baedeker · C. Liedtke Wuppertal Institute for Climate, Environment and Energy, Wuppertal, Germany e-mail: carolin.baedeker@wuppertal.org

C. Liedtke e-mail: christa.liedtke@wuppertal.org

© Springer International Publishing Switzerland 2017 D.V. Keyson et al. (eds.), *Living Labs*, DOI 10.1007/978-3-319-33527-8_30 391

M. Burbridge · G.M. Morrison (🖂) Curtin University Sustainability Policy (CUSP) Institute, Curtin University, Perth, Australia e-mail: greg.morrison@curtin.edu.au

that each came about for different reasons which might include significant public or private sponsorship, or through academia-business co-creation, and this too is reflected in the business case. We also show that the business cases are not static, but may evolve over time as opportunities are created and as partners develop a clearer understanding of the potential of each SusLab. We propose that, based on a common definition of a SusLab, theoretical considerations and societal needs, as well as insights from the cases, it should be possible to build a business case for a SusLab which draws on knowledge rather than learning-by-doing.

**Keywords** Living lab · SusLab · Business case · Triple helix · Sustainable innovation · Path dependency · Transition · Social practice theory

## **30.1 Introduction**

There are at least 170 active Living Labs across the globe (ENoLL 2015). All living labs have specific attributes in common (Salter and White 2013). One is that they aspire towards the innovation of new products or services; they involve close relationships between at least two of the following three sectors: academia, business and society or consumer; they involve co-creating the innovation bringing to bear the expertise of stakeholders on the innovation. However, not all Living Labs contribute to the delivery of sustainable development (Vanaker 2014), nor necessarily progress towards the delivery of some of the recently adopted sustainable development goals (UN 2015).

This chapter only deals with business models of sustainability in Living Labs (SusLabs). Specifically we have reviewed the experience and evidence from four active Living Labs which are part of the SusLab North West Europe (SusLab NWE) network.

## 30.2 SusLab Case Studies

The SusLab network covers facilities in North West Europe and provides acceleration of innovation for the building industry and society. SusLabs may demonstrate energy efficiency (some are even energy producers) and focus on energy use in and around the home, but they also offer innovation in terms of user and practice, as well as water and resources other than energy. They offer the opportunity for partners to undertake insight research, product prototyping and field testing in a variety of building typologies that represent society at large.

At present the network includes four linked hubs:

1. HSB Living Lab, which is a unique international facility on the Chalmers University of Technology campus in Gothenburg, where researchers and 30 Business Models for Sustainability in Living Labs

societal actors can co-create ideas and initiatives for products and services which will enable sustainable living.

- 2. SusLabNWE Living Lab which is on the London Sustainable Industries Park (London SIP), East London and has been designed, supplied and built by Climate Energy Homes. SusLabNWE commissioned the Institute for Sustainability, working in partnership with the Royal College of Art (RCA) and Imperial College London to procure and run the SusLab.
- 3. The Concept House Village which is an urban area in Rotterdam dedicated for prototype houses that has been and will be used, tested and evaluated by actual users from surrounding neighborhoods.
- 4. The SusLab North-Rhine Westfalia (SusLab NRW) Infrastructure which consists of a Smart Home Lab, real home environments and showcase apartments.

The SusLab network is transnational and has dealt with all aspects of valorization through partner and stakeholder dialogue. According to Carlsson (2006) there are few studies of the degree of internationalization of innovation systems so the SusLab NWE is an interesting case to study.

In the following sections we provide a brief analysis of the theoretical underpinning of the SusLab model (Sect. 28.4) before offering an analysis of the different business models that have been developed by the SusLab members (Sect. 28.5). We discuss some of the issues that are common amongst the SusLabs and some that are distinct. Although all SusLabs concern themselves with delivering evidence-based sustainable innovation they all have quite different business models. These different business models are historical and have partly arisen as a result of different motivators for the creation of the SusLab. Interestingly too, it appears that the business model can change over time as relationships and understandings of the potential benefits that can be co-created also develop (Franzen 2015).

## 30.3 Theory

The ideas behind the development and running of living laboratories vary depending on purpose and this has been reviewed by Dell'Era and Landoni (2014). Their definition of a Living lab concerns the methodology, which does not underpin a business case or model well. We have modified their definition to see the Living Lab as a place rather than a methodology.

A Living Lab is a real-life place for user¹ co-creation of innovations in knowledge, products, services and infrastructures.

Based on this definition we have identified three theories in particular that are worth further consideration in developing a business case and model for a SusLab.

¹User is used in general terms and may refer to those living in the lab, if there are any, but equally well to stakeholders from business, society and academia.

The first is the triple helix. This theory owes its existence largely to the work of Henry Etzkowitz and Loet Leyesdorff (2000) from Stanford University who, along with others, developed the notion that business, academia and government (or society in general) all share common interests (Etkowitz et al. 2000; Etzkovitz 2013; Ranga 2015; CEC 2006). Put simply academics produce research which can help business develop innovative products which, if successful, enhances the products competitive advantage, which in turn can deliver societal benefit, such as regional economic development, employment (Nyman 2015; Hessels 2008; Erosa 2012; Gebhardt 2015) and perhaps even the transition to a low carbon economy (Pohl 2008; Hadorn 2006). Although Etzkowitz and his colleagues did not initially dwell on the sustainability issues they argued that if business, society and academia can work closely together then the potential for innovation could be enhanced, delivering benefits to all partners (Etzkowitz 2000). For the partnerships to be successful it is necessary to recognise the different cultures of each organisation (Max-Neef 2005; Nicolescu 2015; Erosa 2012) and different ways of working within different organisations (Jantsch 1970; Pohl 2011). This is why Living Labs are so interesting (Sunitiyoso et al. 2012).

The second is path-dependence theory which is important to understand because it influences the creation of the different business models (Nee and Cao 1999; Bednar et al. 2015; Malm et al. 2012). Path dependency theory sets out how decisions made today are influenced by how previous decisions have been made (Christensen 1997; Senge 1995; Dolan 2015). This does not mean that people cannot change the way they do things, but rather that how they respond will be conditioned by how they have responded in the past (Dolan 2015; Senge 1995). This turns out to be an important issue when we are looking at the different business models adopted by SusLab participants (Vanaker et al. 2014).

Finally, the issue of socio-technical change is important (Geels and Kemp 2007) and in this respect, social practice theory (see Sect. 1.5 for fuller discussion) helps participants to conceptualise environmental behaviour and awareness and design sustainable product-service-systems around the home (Baedeker et al. 2014; Liedtke et al. 2015). This is important because studies in failed innovations have shown that the benefits derived from an eco-product are not fully realised if they have been designed without input from users (Spaargaren 2011). It is for this reason that co-creation is central to the SusLab approach. Involving end users in the design of the product helps to reduce negative rebound effects (i.e. making both the product and the innovation process more efficient) and it is argued that the design of product service systems (PSS) will help with the transition to a low carbon economy (Liedtke et al. 2015 and see Sect. 1.3 for further discussion of these issues). The purpose of the PSS is to focus on the service that is being delivered (for example, the outcome desired might be a warm home) rather than what is producing it (for example, the process of producing and dispersing the output, heat, is the central heating system) and so design focus is on the outcomes service users want produced by a low carbon system. (Liedtke et al. 2015).

### **30.4** The Business Case

The business case for the SusLabNWE comprises two components. One is the value of being part of SusLabNWE; the other component is the development and maintenance of the SusLab infrastructure itself.

The four SusLab partners have utilised different models, partly as a result of how the individual SusLab came into existence, and partly as the result of what the SusLab was intended to deliver.

As the labs are at the early to medium stage of development (all are active but some of the infrastructure has not yet been built) it is not possible to give details of the benefit outcomes that have been accrued. However, it is possible to set out the value that is expected to be delivered and that forms the basis of the investment that partners have committed. The business cases deal with investment (to create the SusLab), income (from the delivery of services) and value (financial and nonfinancial benefits derived from the SusLab).

Although each SusLab, and therefore each business case, is different each share some common elements (Etzkowitz and Leydesdorf 2000; Etzkowitz et al. 2000; Etzkowitz 2013). The common elements come largely from being part of the transnational SusLab North West Europe (NWE) network.

## **30.5** SusLab NWE—The Value of the Network

The SusLab network provides real value to the labs themselves but importantly also to business (both active and future) as well as society. The network was created as the result of a grant from the European Union's Regional Development Fund through INTERREG which is designed to help member states with the development and sharing of information across borders.

Some of the key shared benefits to the member SusLabs of the network is set out in Table 30.1. Business and society also benefit from the value that the network and partnership generate (Ranga 2015). Business benefits from a rigorous, shared process for development of market ready innovation and it follows that business will then be confident that the methodologies are academically robust representing the latest developments in research. Business also gains value from working with potential end-users who are not tied to any organisation in the development of their product or service at various stages (in-sight research, product prototyping and field testing) in the product lifecycle, all of which takes place in a real living place (or near-living in the case of the UK SusLab)—cf. definition of a Living Lab under 8.1.

Society gets value from the network by the development of products, services and product-service-systems that contribute to a sustainable lifestyle (the studied SusLabs are largely focussed on reducing energy usage and carbon emissions) as well as economic benefits associated with successful development of the SusLab

Communications	Document ongoing work on SusLab website, including findings from HSB Living Lab, and the Living Lab at the Institute of Sustainability Finalise SusLab book and promote methods at scientific and professional forums as well as at network events
Academic	Work with network to share and co-develop new SusLab methodologies Publication of results and methodology in academic journals Continue to develop SusLab sensor tool kit to support new work (for exam- ple Building Technologies Accelorator—building occupancy certification system (BTA-BOCS) pilot projects in office buildings) Continue to develop the SusLab tools such that the toolkit can be easily deployed in the field by sustainable building researchers and practitioners Each SusLab has a measuring and monitoring element and all four can be connected to a single data store and analysis tool to enhance rigour and ensure learnings are shared
Outreach	Maintain network of living labs linked to SusLab website Link SusLab work to regional networks (for example, TU Delft and AMS, Chalmers and HSB housing association) Leverage SusLab network combined with new partners to develop new joint projects Value/weight to partners from increased potential to attract further funding (public and private) for SusLab and product development Enhanced business and academic brand value

Table 30.1 SusLabNWE—value to partners in the SusLabs of the Triple Helix

and successful product-service-systems (Maassen and Stensaker 2010). This latter value is likely to include new jobs, economic diversification and regional development (Etzkowitz et al. 2000; Leydesdorff and Deakin 2011).

## 30.5.1 HSB Living Lab, Chalmers University of Technology

The HSB Living Lab model builds on the assumption that the costs for building the facility can be more or less returned to HSB (HSB is a Swedish national housing association) over a longer time horizon. It also builds on the facility attracting companies who are prepared to commit to knowledge generation and development for their own and societal interest. Finally, the model also builds on the facility being attractive to a whole range of researchers at Chalmers, as well as across the SusLab facility (Table 30.2).

The relationship between the core partners (Chalmers, HSB and Johanneberg Science Park) is maintained through a binding ten year agreement signed between the President of Chalmers and the CEO of HSB in Gothenburg. Further, a business to business partnership has been established with 10 partners along the value chain (including architect, building company, IT specialist, bathroom supplier and white goods specialist) who have made a financial and resource commitment to the facility.

The total cost for the building was expected to be fairly typical for a pre-fabricated modular building with a land footprint of  $420 \text{ m}^2$  and four floors. However,

### 396

30 Business Models for Sustainability in Living Labs

Business model	Knowledge to business and business to business
Partners	HSB Housing Association, Chalmers University of Technology, Johanneberg Science Park. 10 other business and societal partners
Sectoral interest	Local government, academia
Duration	10 years minimum
Purpose	Attracting companies who will commit to research and develop- ment for their own and societal interest
Motivation to create the lab	First building demonstration for Johanneberg Science Park

Table 30.2 Summary—HSB Living Lab, Chalmers University of Technology

the building has been designed as a flexible living lab and this has meant extra costs which are unlikely to be covered in the lifetime of the building. HSB as an organization was prepared to accept an annual loss for the building over the 10 year period, after which the value of the modules will be roughly 50 % of the new construction cost.

Chalmers researchers have been provided with initial co-funding which has thereby initiated the idea generation between the SusLab partners during the formative process of the HSB Living Lab. Co-funding for the sensor networks based on the SusLab winter pilot has also been an important contribution. Chalmers researchers have obtained and are seeking further national and international research funding to bring the facility up to some 20 active researchers, technical staff and PhD students working in the HSB Living Lab.

The financial sustainability of the HSB SusLab will rely in part on securing the research grants set out above but also by generating an income from working with businesses to develop products and services. This will, in all likelihood, be based on a positive feedback loop whereby successful product development will lead to wider relationships with business which will in turn lead to more grants and which will then lead to more capacity to develop more products. The accommodation itself will be rented out to students (who are happy to be part of the SusLab) at market rates which will cover the costs of running and maintaining the accommodation.

## 30.5.2 SusLab NRW

The SusLab in North-Rhine Westfalia—led by the Wuppertal Institute—builds on real home environments (in the City of Bottrop). It consists of 4 main pillars: real home environments, the Smart Home Lab, showcase apartments and a new concept for energy efficiency consulting (product-service-system) (Table 30.3).

Insight research and prototyping in real home environments is substantial in a SusLab Infrastructure. One key component for the exploitation has been the development of a methodology for testing products and prototypes in real home

Business model	Business to business and business to consumers. Embedded in existing, lived-in homes
Partners	Wuppertal Institute, Hochschule Ruhr West, Innovation City Management GmbH
Sectoral interest	Local government; academia; technical college
Duration	On-going
Purpose	Testing products and services in real home environment
Motivation	Developed as part of Ruhr region's commitment to reduce $CO_2$ emissions by 50 % as well as in response to local economic downturn

Table 30.3 Summary—SusLabNRW

environments. The handling and the accuracy of fit of products and services has to be tested within the daily practices of households. The sensor infrastructure consists of data loggers for measuring room climate (temperature, humidity,  $CO_2$  concentration) which are connected to the internet, an internet connected database at the Hochschule Ruhr West (HRW) and software for evaluation of the measured values. This infrastructure was tested in cooperation with the companies RWE and Deutsche Telekom by evaluating the usability and efficiency of their smart home products and systems. The HRW will offer this infrastructure to other companies for further testing of assistive technologies. The aim is to offer investigations in living labs for product improvement as a service to different companies, allowing full exploitation of the facility. The smart home is a 30 m² sized lab that is used for testing of the prototypes (proof-of-concept) and for tests of user-interaction before installing them in real homes.

In order to be able to fund the infrastructure, in addition to contracts for cocreating products and services, the German partners HRW, Wuppertal Institute and Innovation City Management GmbH developed the concept of energy efficiency services for communities. The product-service-system is a new concept for energy efficiency consulting for tenants and homeowners. The concept is based on a pre-analysis of the energy efficiency potential by using the evaluation methods developed in the SusLab project. The analysis of room climate delivers a first impression on user behavior, heating system functionality and building insulation characteristics. Based on the results a suitable, economically optimized proposal for increasing energy efficiency can be given to tenants and homeowners.

## 30.5.3 SusLab Living Lab in London

The SusLab Living Lab in London is a procured element of the project and the Institute for Sustainable Futures is responsible for delivering and managing the facility. Through a competitive process, Climate Energy Homes were chosen as the preferred supplier to design, supply and build the facility. The Institute has worked with others to deliver this facility and improve the exploitation (Table 30.4).

### 30 Business Models for Sustainability in Living Labs

Business model	Demonstration project
Partners	Institute for Sustainability (NGO)—responsible for build and management of the living lab, Imperial College, Royal College of Art
Sectoral interest	NGO, Academia, commercial
Duration	Through to 2024
Purpose	Procured project to demonstrate potential of collaboration
Motivation	To provide a living laboratory that is closer to a home environment

Table 30.4 Summary—SusLab Living Lab in London

The UK has very few facilities which can act as a SusLab outside academic focused facilities, which are usually science laboratory based and not near to a living environment. Hence this facility is welcomed by academics, housing developers and manufacturing companies based in the south east of England. Some wider interest from health authorities has been shown to conduct care in the community type simulations. This facility will be able to emulate a high proportion of the building typology in the UK and therefore participants can relate more easily to experiments or assessments which can be run in the SusLab. This choice is also important as many highly elaborate building systems and cutting edge technologies usually have been tested or have their own research and development back office.

The SusLab aims to attract stakeholders, locally, nationally and EU-wide who are interested to commit to research and development of innovative energy control and awareness technology for their own and the wider benefits associated with being more effective with energy use. The SusLab also aims at attracting research and design practices from local and international academic institutions and institutions that are interested in co-creation and experimentation of solutions in relation to daily life practices.

The original concept was to enable free and full use to encourage the core aspirations of the SusLab project to be delivered over the life of the facility. Therefore, encouragement to use is the key feature and interested parties will be required to reinstate the facility to its current condition on completion of their activities. Encouragement will be on extending use and maintaining the SusLab in at least its current condition, rather than generating an income from activity, other than to cover the operational costs including utilities, cleaning and routine maintenance.

## 30.5.4 Concept House Village

The Concept House Village (CHV) is an initiative from City Ports Academy Rotterdam and is executed by Rotterdam University of Applied Sciences, Delft University of Technology and the Woonbron Social Housing Corporation. The facility is financially and in-kind supported by the Municipality of Rotterdam.

Table 30.5         Summary Concept House Village				
Business model	Sponsorship			
Partners	Delft University of Technology, Woonbron Social Housing corporation			
O t 1 ! - t t	Local Covernment Academia			

Sectoral interest	Local Government, Academia
Duration	Up to 10 years
Purpose	Procured project to demonstrate potential of collaboration. Students involved in design and build
Motivation	To provide a living laboratory that is closer to a home environment

The partners have a long-lasting agreement on the exploitation of the CHV-facility signed by the boards of executive partners which provides a sustainable cost model for exploitation (Table 30.5).

During the first period of its existence in the Netherlands, CHV was financed from different sources. With the financial support from the INTERREG SusLab NWE project, the Dutch Ministry of Economic Affairs (Peaks in the Delta) and the Municipality of Rotterdam it was possible to establish the CHV-facility and to manage the acquisition of new prototypes, to guide the building processes of the prototyping, to coordinate the research program and foster the knowledge transfer.

The basic approach is that the involved consortia finance the prototypes themselves. The two involved knowledge institutes lead the development of the first two prototypes (CH Prototype 1 and CHIBB). The partners of the two consortia financed these prototypes with important additional subsidies from SusLabNWE and Peaks in the Delta.

Those two prototypes serve as a laboratory for the research programs of both knowledge institutes. The on-going research is mainly financed by the institutes' research funds. The prototypes also serve as a learning environment for the Bachelor and Master's students of the institutes. Besides the whole design and development of CHIBB by students of the Rotterdam University of Applied Sciences, students of the Albeda Polytechnic built the CHIBB prototype.

The CHV-facility has agreements with the consortia behind the prototypes about the involvement of students and research staff of both institutes in the development, monitoring, testing and validation of the dwellings.

In 2015 the CHV facility started revising its business model, to become independent from structural public funding, to accelerate the real market application of the results of the prototyping and to become a regional focal point of sustainable building to support the realization of the political ambitions.

## **30.6** Conclusion

SusLabNWE is in the early stages of development. Some of the infrastructures are still being built but to an extent that will always be the case as the SusLabs will be forever changing. It is therefore not possible to include all values in each of

400

### 30 Business Models for Sustainability in Living Labs

the SusLab's business models. In part they are benefits that have been valued and accepted as part of the development of the network, but they have not, yet, necessarily been realised. Clearly the business models are evolving. For example London SusLab and CHV are actively refreshing theirs and the SusLab NWR in Germany has identified a new stream of potential income which they are currently developing.

One interesting insight to come from this study of the business models is how the genesis of the SusLab has influenced the business model. For example, part of the motivation for the development of the SusLab NWR was to encourage economic diversification in the Ruhr. As such it initially attracted interest and assistance from Government to develop itself as a SusLab. Similarly the CHV was initiated on the backdrop of public funding, but now that it is established it is seeking to become independent of public funds. Both of these demonstrate, in real terms, the impact of path dependence theory (Bednar et al. 2015).

The other interesting, and unexpected, result of the work has been how relationships with business varies over time, and between businesses (Franzen 2015). By this we mean that different models suit different businesses.

Some are seeking involvement in research that will help to deliver sustainable lifestyles and to change social practices into more sustainable ones. All SusLabs share this opportunity, but the Concept House Village was developed with this market in mind (although it is revising its business model). Effectively CHV set the research program and then invited industry to fund it if they are interested. This is a widely adopted model to fund research in universities.

On the other hand, HSB Living Lab is looking to work with business to develop innovative products and services. The HSB model involves a closer, symbiotic, working relationship between business and academia with both sharing the same project goals but with each having slightly differentiated project deliverables. For example Chalmers researchers will also be drawn to a greater extent to developing the knowledge base within their fields of expertise; business will be drawn by the opportunity to develop and test products using the latest research. All parties seek to deliver the best co-created product, service or system and will gain individual benefits from working together. These outcomes reinforce the benefits predicted by the theory (Pohl 2008; Nicolescu 2015).

Recently completed research has also demonstrated how relationships, values and benefits can change over time (Franzen 2015). A relationship that might start on the basis of developing a new market for a business can with time enable a business the opportunity of developing a product line contributing to a sustainable lifestyle as the relationship between partners develops and matures.

Despite there being close to 170 active Living Labs there is a need to undertake further analysis of the various business models to identify common themes in the business cases, what is included and excluded and if Living Labs with the same aims share the same style of business case. There is the need to share an understanding of the business case development in order that the next generation of Living Labs do not have to re-learn the same lessons that the current generation have learnt. Further research as outlined above will also help to start to address the deficiency in studies of transnational innovation systems as identified by Carlsson (2006).

## References

- Baedeker, C., Greiff, K., Grinewitschus, V., Hasselkuß, D., Keyson, D., Knutsson, J., et al. (2014). Transition through sustainable Product and Service Innovations in Sustainable Living Labs: application of user-centred research methodology within four Living Labs in Northern Europe. Paper for presentation at the 5th International Sustainable Transitions (IST) Conference, August 27–29, 2014 Utrecht, The Netherlands.
- Bednar J., Jones-Rooy, A., & Page, S. (2015). Choosing a future based on the past: Institutions, behaviour, and path dependency. *European Journal of Political Economy* (in print).
- Carlsson, B. (2006). Internationalisation of innovation systems: a survey of the literature. *Research Policy*, 35, 56–67.
- Christensen, C. (1997). The innovators dilemma. Harvard Business Review, USA.
- Commission of the European Communities. (2006). *Delivering on the modernisation agenda for universities, education, research and innovation*. COM(2006) 208 final, Brussels.
- Dell'Era, C., & Landoni, P. (2014). Living Lab: A methodology between user-centred design and participatory design. *Creativity and Innovation Management*, 23, 137–155.
- Dolan, P. (2015). Happiness by design. London: Plume.
- ENoLL (European Network of Living Labs. Retrieved November 17, 2015, from http://www.openlivinglabs.eu/.
- Etzkowitz, H., & Leydesdorf, L. (2000). The dynamics of innovation: from national systems and mode 2 to a triple helix of university-industry-government relations. *Research Policy*, 29, 109–123.
- Etzkowitz, H., Webster, A., Gebhardt, C., & Terra, B. (2000). The future of the university and the university of the future: Evolution of ivory tower to entrepreneurial paradigm. *Research Policy*, 29, 313–320.
- Etzkowitz, H. (2013). Anatomy of the entrepreneurial university. Social Sciences Information, 52(3), 486–511.
- Erosa, V. (2012). Dealing with cultural issues in the Triple Helix Model implementation: A comparison among Government, University and Business Culture. *Social and Behavioural Sciences*, 52, 25–34.
- Franzen, S. (2015). Sustainable transitions at home: HSB living lab project member organisations' view on sustainability and the living lab itself. Dissertation, Chalmers University of Technology.
- Gebhardt, C. (2015). The spatial dimension of the triple helix: the city revisited—towards a mode 3 of innovation systems. *Triple Helix*, 2(11).
- Geels, F., & Kemp, R. (2007). Dynamics in socio-technical systems: Typology of change processes and contrasting case studies. *Technology in Society*, 29, 441–455.
- Hadorn, G., Bradley, D., Pohl, C., Rist, S., & Wiesmann, U. (2006). Implications of transdisciplinarity for sustainability research. *Ecological Economics*, 60, 119–128.
- Hessels, L., & van Lente, H. (2008). Re-thinking knowledge production: A literature review and research agenda. *Research Policy*, *37*, 740–760.
- Jantsch, E. (1970). Inter- and transdisciplinary University: A systems approach to education and innovation. *Policy Sciences*, 1, 403–428.
- Leydesdorff, L., & Deakin, M. (2011). The triple helix model of smart cities. A neo-evolutionary perspective. *Journal of Urban Technology*, 18(2), 53-63.
- Liedtke, C., Baedeker, C., Hasselkuß, M., Rohn, H., & Grinewitschus, V. (2015). User-integrated innovation in Sustainable LivingLabs: An experimental infrastructure for researching and developing sustainable product service systems. *Journal of Cleaner Production*, 97, 106–116.
- Malm, A., Ljunggren, O., Bergstedt, O., Pettersson, T., & Morrison, M. (2012). Replacement predictions for drinking water networks through historical data. *Water Research*, 46, 2149–2158.
- Maassen, P., & Stensaker, B. (2010). The knowledge triangle, European higher education policy logics and policy implications. *Higher Education*, *6*, 757–769.
- Max-Neef, M. (2005). Foundations of transiciplinarity. Ecological Economics, 53, 5-16.

- Nee, V., & Cao, Y. (1999). Path dependent societal transformation: Stratification in hybrid mixed economies. *Theory & Society*, 28, 799–834.
- Nicolescu, B. (2015). How can we enter in dialogue? Transdisciplinary methodology of the dialogue between people, cultures and spiritualities. *Dialogue and Universalism*, 1, 9–19.
- Nyman, G. (2015). University-business-government collaboration: From institutes to platforms and ecosystems. *Triple Helix*, 2(2).
- Pohl, C. (2008). From science to policy through transdisciplinary research. *Environmental Science & Policy*, 11, 46-53.
- Pohl, C. (2011). What is progress in transdisciplinary research? Futures, 43, 618-626.
- Ranga, M., & Etzkowitz, H. (2015). Triple Helix Systems: an analytical framework for innovation policy and practice in the knowledge society. H-STAR, Stanford University
- Salter, R. & White, S. (2013). *Collaborative research in the real world: review of living labs.* Living Lab Framework Project, CRC for Low Carbon Living
- Senge, P. (1995). The Fifth Discipline: The art and practice of the learning organisation. New York: Doubleday.
- Spaargaren, G. (2011). Theories of practice: Agency, technology, and culture. Exploring the relevance of practice theories for the governance of sustainable consumption practices in the new world-order. In *Global Environmental Change*, 21, 813–822.
- Sunitiyoso, Y., Wicaksono, A., Utomo, A., Putro, U., & Mangkusubroto, K. (2012). Developing strategic initiatives through triple helix interactions: Systems modelling for policy development. Social and Behavioural Sciences, 52, 140–149.
- United Nations. (2015). *Transforming our world—the 2030 agenda for sustainable development*. New York: United Nations.
- Vanaker, T., Manigart, S., & Meuleman, M. (2014). Path dependent evolution versus intentional management of investment ties in science based entrepreneurial firms. *Baylor University*, doi:10.111/etap.12007.

### B.3 Publication 3

### <u>Burbridge, M</u> (2017). If living labs are the answer – what's the question? Procedia Engineering, Vol 180 pp 1725 – 1732



industry and elsewhere. If there is to be a close relationship between business and publicly funded universities the innovation it produces needs to be based on evidence of its efficacy in the Anthropocene; if it is not it will be undermining the ability of society to deliver the sustainable development goals endorsed by the UN general assembly in 2015 [8].

#### 1.1. Background

The world's economy is transitioning away from being industry based towards becoming a knowledge economy where knowledge is the predominant factor in driving economic growth. A knowledge economy is only truly possible in a networked world where knowledge can be shared with ease due to the network effect [9, 10].

#### 1.2. Economic impacts

Developed economies are becoming increasingly knowledge intensive [11]. This is a process that has been recognised since the mid-20th century and possibly since the dawn of the Anthropocene in the 1750s [12]. The increasing reliance on, and importance of knowledge and information in product and service development has been widespread and accelerating with over 50% of GDP of major economies being knowledge-based [13]. This may be as high as 70% in some developed economies. However, the ability of economists to track the stocks and flow of knowledge through an economy is still developing [13].

The 'knowledge economy' is a widely used term to signify the intensification of knowledge use in the modern economy [13, 14, 15]). Using the term does not mean that previous economies (ie agricultural or industrial economy) were knowledge free – they were not. Those economies used the available knowledge effectively and were driven by technological innovation [12]. However, the key change was with the adoption of technology horizontally across the economy rather than vertically within trades. For example, the steam engine, and thereby access to power on demand, transformed many industries – from agriculture to textiles to public transport. It was this adoption of 'horizontal' technology (or general purpose technologies [16]) in the 1750s that started the transformation towards a knowledge-based economy; a trend Drucker noted in 1959 when he coined the phrase 'knowledge workers' [12].

The pace if the development has increased markedly in the late 20th and early 21st century with the widespread adoption of the internet [17] and the exponential growth in devices linked to the web (with Intel predicting 20billion devices connected by 2020 [18]). Some argue that proof of its development came with the rise, from the 1960s in the service economy where people with specific knowledge and skills did work for those who did not have it [19]. But the knowledge economy should not be merely defined by its initial association with the service economy. Rather it can be identified by its investments in research and development, education and training and new managerial work structures [13].

#### 1.3. Societal impacts

And while our economy is transforming so to is our society and environment. The knowledge economy is leading to the development of two types of jobs: those jobs where people tell computers what to do and those jobs where computers tell people what to do [20]. The knowledge economy has been blamed for real wage stagnation in the low to middle income groups, stalling service sector productivity growth and increased inequality within society.

In order to maximise the benefits that the new knowledge and information will bring it is essential to adopt new work practices [21, 16]. Investments in technology enable complementary organisational investments which help to improve productivity [11] and these are often associated with workforce changes enabled by the adoption of the technology [22]. The flattening of hierarchies, development of team work where everyone has a voice, and the general empowerment of the workforce can only happen with easy access to knowledge and information are evidence of this new economic era, as well as a clear break with the past industrial based economy [23]. However, the gains from the adoption of new technology are not the same in each organisation – the limiting factor is the ability of the management to create a new environment that will maximise the benefit of the technology [24]. Similarly simply adopting a technological fix without associated process changes can have the opposite effect [16].

#### 1.4. Environmental impacts

The world's environment is also being impacted which Hardin (1968) described as the "tragedy of the (un)managed commons" [25]. Today there are few if any unmanaged commons left and yet humanity's impact on our natural systems is significant and far reaching, even though positive for human well-being and economic growth [26]. The result is that the current geological epoch is referred to as the Anthropocene in view of the impact of humanity [27] – which are so significant as to be geological in scale.

#### 1.5. Emergent 'mega' trends [43] and institutional responses

At the same time as these global changes to the economy, society and environment it is anticipated that economic activity will move from the north to the south and the west to the east [18]. This puts Australia – historically suffering from the "tyranny of

distance" [28] – to being in the same time zone as a quarter of the world's universities. The middle class of this region is expected to expand by over a billion people by 2050 [18]. Politicians have argued that this means that the early 21 century is "the most exciting time to be Australian".

These are indeed exciting times and governments are taking steps to move in a more equitable and sustainable direction. In 2015 the world leaders unanimously agreed to 17 sustainable development goals (SDGs) which, unlike the Millennium Development Goals, all countries will pursue. These goals will guide the world's development over the next 15 years [8]. Further in Paris in 2015 world leaders committed climate action (goal 13) by taking steps to limiting global warming to 2 degrees centigrade (with a further commitment to plan to limit the increase of temperatures to 1.5c).

#### 2. Research aims and methodology

This paper is part of a PhD looking at how universities and industry can better collaborate for low carbon innovation. This paper is designed to give a strategic theoretical backdrop against which further practical work can be tested. It is based upon a high level review of the literature searching under collaboration, living labs, sus labs, innovation, knowledge economy and triple helix through the Web of Science. Papers were read and categorised according to their relevance to both the development of collaborative structures for innovation as well as the delivery of 'future-proofed' innovation.

Within this paper the term 'university' is used to embrace academia in general with an emphasis on research for the public good. Similarly the terms 'business' and 'industry' are used interchangeably to reflect private sector economic activity. There is no significance with the use of the different terms – they are used to reflect the production of knowledge (for the public good) as well as private sector economic activity (largely for private benefit). Also research and innovation are used in this paper as part of a continuum where innovation is seen as taking research and commercialising it to produce a new product or service.

#### 3. Universities and industry - partners in innovation?

Universities and industry have a complex relationship to each other in terms of how they engage with each other for research and collaboration. This is set out in Table 1. The relationship appears to be one of client/customer, or transactional rather than one of collaboration or partnership where each derives value, even if that value is different for each. For example, industry is seen to be weak at collaborating with either other industry or universities. Universities however are good at collaborating with each other (for research) but poor at collaborating with industry for innovation. Universities and industry do work well together however undertaking research. It is not clear, from this literature review, the degree to which such research partnership produces research that derives public benefit or only research that derives private benefit. Industry funds universities to undertake research at a level that is greater than the OECD average. This stands in contrast to Australian Universities and industry being at the foot of the OECD league table for collaborating for innovation. In a knowledge economy collaboration will be the key to success as it is unlikely that any one organisation will hold all the knowledge and business skills to deliver innovative products and services in a networked economy.

#### Table 1: Australian Universities and business: customers or collaborators?

	Industry	Universities	Customers/students	Suppliers	Use foresight	Research user?
Industry	Collaboration between industry for innovation is weak and localized [29]	Industry fund universities to undertake research at level greater than the OECD and EU28 average. [30]	Industry is weak at collaborating for innovation with its customers [31]	Industry is weak at collaborating for innovation with its suppliers [31]	Industry weak at using strategic foresight [7]	User, and active funder of research
Universities	Collaboration for innovation is amongst the worst in the OECD [15]	Universities are good at collaborating with universities [32]	Investment by universities and students/alumni is the focus of increased attention [33]	Little evidence of universities collaborating with their suppliers for innovation	Unclear as to whether universities use foresight (although they do research scenarios)	Unclear if universities use the research they produce. Initial findings suggest they do not

One interesting absentee from the literature reviewed is the degree to which universities are willing to use their own resources (eg investment in campus development) to either imbed research outcomes or innovation in the p process. In such a case Universities collectively could use their buying power to partner with organisations who are interested in innovation or producing product ready for the Anthropocene. The fact that such a discussion is absent from the literature reviewed suggests this area could be fruitful for developing the next canvas upon which to draw university and industry collaborations.

3.1. Barriers facing collaborators

Universities and business need to become better at collaborating [34], yet resistance to change is deeply engrained within humans [35, 36], as well as the organisations we create [37, 38] and this is, in part, because of the evolutionary desire to not wasting resources by conserving energy.

This has influenced the way we think. Kahneman (2011) argues there are only two ways humans think: one is reactive unconscious thought and the other is contemplative and conscious thought [35]. Being able to react when you are being chased by a hungry tiger is very helpful for survival. However, it is not necessarily helpful when faced with a new situation. New situations possibly need new solutions, and new solutions can only be developed deliberately through conscious thought. Conscious thought takes time, and importantly (in evolutionary terms) energy. Evolution has developed the brain to limit this energy intensive response as much as possible. If new thought is needed the brain seeks to take the conscious thought and convert (through learning) into unconscious action.

The importance of this for organisations is that the same conditions apply. Businesses have processes to efficiently deal with the known situations they face. This is comparable to unconscious thought. If a new situation faces them they need to develop ways to deal with it and this is the equivalent of conscious thought. The new situation will be dealt with, and when it reoccurs it will also be dealt with on the basis of what the organisation did before. In the social sciences literature this is path dependence theory [39] where decisions made today are influenced by how successful decisions were made yesterday. This is also learning by doing [40].

Evidence from neuroscience supports this hypothesis with the discipline of neuroplasticity demonstrating how particularly damaged brains can relearn activities and then, in order to save energy, turn them into reactive thought [41]. The difference that neuroplasticity brings to this stream of research is that it suggests that path dependence theory must, and always will, be a self-fulfiling prophesy as neither humans nor organisations have the necessary resources of energy (or finance) to consciously respond to each situation they face as if it were new. They need an armoury of (unconscious) responses in order to use their energy (or financial resources) efficiently and this limits their ability to respond to new opportunities effectively. Businesses ability to respond is strictly limited by two finite resources: leadership and financial.

Although all organisations are capable of learning, those that do not have established process, or those where processes can be quickly established to match the issue they are confronting, are the most likely to respond the quickest. In innovation – particularly in a situation of disruptive innovation (ie the scenario being faced is entirely new) age is important. Typically young, or small organisations, being the most adept at adopting the change quickly [38] as from their perspective it is not really change in as much as everything is new to them. Importantly, in a disruptive environment first-mover advantage is the key to success [38]. And the knowledge economy is based on disruption – disruption to organisations, society and to technology.

The importance of this insight is that change is and always will be difficult and resisted for evolutionary reasons. However, to make is less entrenched there are key steps organisations and people can take to maximise the chance of successful implementation of change. It is less difficult to deliver when all stakeholders understand what motivates each other and there is a clear understanding of the motivation behind each proposal and what the action is seeking to achieve [21, 42, 43]. Hakkarainen, and Hyysalo (2013) argue for the need to "learn how to interact before interacting for innovation" [67]. Living labs provide the opportunity to address such organisational and cultural barriers as long as there is strong governance model between the partners [44].

In transitioning to a deeper knowledge-based economy will impact technology, society and organisational structures responding effectively to the changes will be key.

#### 4. What is the role for universities in the knowledge economy?

The knowledge economy brings with it technological, organisational and societal innovations and change [45]. In a society dominated by the exploitation of knowledge the role of entrepreneurial universities can be significant at it gives universities the opportunity to partner with business to deliver benefits to society, business and universities. There are many models for such partnership, and the need for such partnerships have been identified for some time [14].

#### 4.1. Living labs

"The OECD science system is facilitating the challenge of reconciling its traditional functions of producing new knowledge through basic research and educating new generations of scientists and engineers with its newer role of collaborating with industry in the transfer of knowledge and technology" [13]

Within Europe the response has been the creation of Living Labs to tackle Europe's declining economic competitiveness and societal challenges [46]. The proposal was developed and agreed under the Finnish Presidency in 2006. Living Labs were developed in response to a shift in the strategic discourse between government and business/universities. In Europe this discourse has been about driving innovation and thence economic development by more effective leverage of public investment in research [14]. This has been productive with significantly higher levels of engagement between business and academia than in Australia [5].

#### 4.2. Evolution of living labs

A Living Lab is a real-life place for user co-creation of innovations in knowledge, products, services and infrastructures. [40] where user is used in general terms and may refer to those living in the lab, if there are any, but equally well to stakeholders from business, society and academia.

There are 170 active living labs in 20 of the 27 EU countries. Most are focused on delivering traditional economic returns (such as jobs, new products and services, regional economic development). But there are sub sets, the so-called Sus Labs [40] and Urban Living Labs [47], Smart Cities [48] that seek to deliver benefits for traditional economic partners, as well as the public good or meeting societal needs [49, 50]. Even though it has been argued that the sustainable development community and living lab communities are "hardly intertwined" [51], increasingly living labs have been seen as an approach to deliver innovation based on societal and end user needs by producing knowledge, goods, services and infrastructure that is fit for the Anthropocene.

Stahlbrost (2012) proposes five key principles, which should permeate all living lab operations: value, sustainability, influence, realism and openness [52]. In the living lab literature there is a discourse about the need to reduce the environmental impact of economic activity in order to deliver value for all stakeholders, with a cleaner environment – rightly - being seen as benefit to society (for example 47, 52, 53]. There is clearly much in this argument. However, this misses the potential power of a living lab to showcase for society, government and business how innovation can be undertaken to meet evidence-based societal and end user needs though partnership between knowledge creators and knowledge users.

The key difference being reducing environmental impact of a product or service does not necessarily equate with meeting science based needs. The clear case in point is the scientific agreement behind the need to reduce carbon emissions. The research community is clear about what society needs to do to prevent anthropogenic climate change exceeding 2 degrees centigrade [54]. A key partner in the living lab community is the research community and the research community should respond to and implement its own research and deliver innovation and mitigation at the same time. It is also for this reason that leadership is so important.

Therefore, in table 2 Stahlbrost's (2012) 5 key principles have been reframed to be explicit about the imperative of delivering the evidence based needs of society as well as helping to deliver progress towards the remaining sustainable development goals.

Table 2. Five key	principles under	pinning living la	ibs (based on	Stahlbrost, 2012).
	berread and a second se		too tomote on	

Principle	Definition
Value	Delivering value for all partners (throughout the value chain)
Sustainability	To follow scientific advice to help maintain a healthy environment as well as to help make progress towards delivery of all of the sustainable development goals
Influence	Acknowledging that all partners have influence in the product innovation process
Realism	Innovation should be conducted in as close to real life environment as possible
Openness	To have an open process to benefit from multiple perspectives

Living Labs offer the potential for universities, academia and society to co-create new knowledge and services together by bringing together various expertise to validate new products and services in a real-life environment [55]. The potential benefits are set out in table 3 below.

Tabl	e 3.	Partner n	eeds an	d potential	l partner	benefits
------	------	-----------	---------	-------------	-----------	----------

	Need	Benefit
Academia	<ul> <li>Partnering with business for innovation</li> <li>Higher impact research</li> <li>Collaboration</li> <li>Research income</li> </ul>	<ul> <li>Enhanced role in a knowledge economy Increase research and teaching output</li> <li>Higher impact research</li> <li>Evidence based research</li> <li>Combined research tracks</li> </ul>
Industry	Lack of capability in collaboration	Developing competitive edge to build on time, to budget and consistent with SDGs
Government society	Not meeting evidence based carbon reductions	<ul><li>Make progress towards delivery of SDGs</li><li>Efficient use of investment in research</li></ul>

Dell'Era and Landoni (2014) have categorised 4 underpinning models of living labs in order to help with understanding the array of different business models [56]. Whilst helpful it is the grey areas between the models that will also reveal insights and develop new models [57]. The basic models are:

Value capturing (using an existing technology in a new way (cf steam engines discussed above));

Value creating (exploring new technologies);

### Based on open innovation; and

Based on closed innovation.

Such categorization is useful since living labs cover a wide array of disciplines, products and services: from health, dementia [58], regional small and medium sized enterprises [59], energy and environmental decisions at a community level [60], people with disabilities [61], sustainable domestic technologies [50], eco-cities [62], smart cities [48], business models for successful innovation [63]; architectural [54] and enabling community innovation [64].

Living labs are not a universal panacea for linking business, universities and society and will not be appropriate for all circumstances. For example, highly innovative university spin-offs may not benefit due to the more homogenous nature of their social networks since real value is derived from a diverse social network [65]. But they are a highly flexible, simple and adaptable model for knowledge based innovation.

And more recently living labs have been created to develop professional competencies - this extending the product or service focus to an individual's development [66]. This latter methodology could offer insights to address the need for system changes to complement technology changes [20, 67].

#### 4.3. Business models for innovation

Whilst leadership is critical for a successful living laboratory there is no single model of effective leadership - with different styles benefiting different geographic areas [68] and structure [40]. However, the business model does need to reflect the operationalisation of the living lab (ie open or closed, exploitative or explorative [57])). The question of different business models for different geographic regions does raise issues for the transferability of methodology [69] (rather than the living lab concept), and as such will result in some mix of learning by doing and learning from experiences of others [40]. However, the problems with transferring methodology between regions (and firms) can be reduced by close attention to understanding how each method works [42].

#### 5. Conclusions and further research

In a knowledge-based economy the issue of developing living labs for innovation is a good one, which has been successful elsewhere. It offers a route to effectively commercialise new products and services in partnership with business. It is a model that can benefit all three actors at the same time (see table 2). Ultimately it can offer universities a more central role in society as their impact would be more widespread and visible. However Australian business and universities are poor collaborators. Change is demonstrably difficult to deliver effectively and continuously. It requires good leadership to create the environment that is accepting of change.

The exciting challenge for universities and business is how to collaborate for innovation whilst addressing the economic, social and environmental issues associated with the transition to a knowledge economy in a way that moves society out of the Anthropocene and towards the evidence based future that society has agreed.

Further research is needed in two areas to understand how to improve collaboration with industry. One area is around campus redevelopment. In such a scenario universities are contracting industry to refresh their campuses. As such universities have the opportunity to integrate research and teaching outcomes into the process. To date there is little evidence that this is being done.

The other is around leadership and whether universities adopting research outputs before asking others to adopt them would impact favourably the potential for business to partner with universities. The theory suggests it would.

#### Acknowledgements

This work has been supported in part through a PhD scholarship from the Low Carbon Living CRC for Mike Burbridge.

#### References

[1] H. Wilson, H, Labour's plan for science. In Labour Party (Ed.) Labour Party, London, 1963.

- ] R Colvile, The great acceleration. Bloomsbury Publishing, London, 2016.
- [3] H. Etzkowitz, & L Levdesdorff, L. The dynamics of innovation: from National Systems and "Mode 2" to a Triple Helix of university-industry-government relations. Research Policy, 29, (2000) 109-123.
- [4] National Innovation and Science Agenda, National Innovation and Science Agenda Report, Government of Australia OECD, OECD Science, Technology and Industry Outlook 2014: OECD Publishing, Paris 2015.
- [6] M. van Geenhuizen, From Ivory Tower to Living Lab: Accelerating the Use of University Knowledge. Environment and Planning C: Government and Policy, 31(6), (2013) 1115-1132.
- [7] P. Gahan, M. Adamovic, A. Bevitt, B. Harley, J. Healy, J. Olsen, & M. Theilacker, Leadership at Work: Do Australian leaders have what it takes? 2016 Retrieved from Melbourne: workplaceleadership.com.au/sal

[8] United Nations, Transforming our world. A/RES/70/1, United Nations, New York, 2015
 [9] B. Metcalfe, Metcalfe's Law after 40 Years of Ethernet. Computer, 46(12), (2014) 26-31

- [10] X. Zhang, J. Liu, & Z. Xu, Tencent and Facebook Data Validate Metcaffe's Law. Journal of Computer Science and Technology, 30(2), (2014) 246-251.
  [11] W. Powell & K. Snellman, The Knowledge Economy. Annual Review of Sociology, 30, (2014) 199-220.

1730

- [12] P. Drucker, The Rise of the knowledge society. Wilson Quarterly, 17(2) (1993)
- [13] OECD, The Knowledge Based Economy, OECD, Paris, 1996 [14] European Union, Helsinki Manifesto, Helsinki, Finland, 2006
- [15] OECD, "Collaboration on innovation", in OECD Science, Technology and Industry Scoreboard 2015; Innovation for growth and society, OECD Publishing, Paris, 2015.
- [16] E. Brynjolfsson, The productivity paradox of information technology. Comm. ACM, 36(12), (1993) 66-77
- [16] L. Byrjonsson, the productivity paradox on information technology Column ACM, 50(12), (1993) 66-77.
   [17] K. Kelly, New rules for the new economy. WIRED, 09/01/1997.
   [18] S. Hajkowicz, L. Rudd, A. Bratanova, L. Hodgers, C. Mason & N. Boughen, Tomorrow's Digitally Enabled Workforce: Megatrends and scenarios for jobs and employment in Australia over the coming twenty years. CSIRO, Brisbane, 2016.
- [19] T. Noyelle, Economic Transformation. The Annals of the American Academy of Political and Social Science, 488, (1986) 9-17
   [20] E. Brynjolfsson, & A. McAfee, Exponential Intelligence. Policy Options, 35, (2014) 66-68.
- [21] P. Senge, O. Scharmer, J. Jaworski, J and B. Flowers, Presence: Human Purpose and the Field of the Future. Bantam Doubleday, New York, 2014 [22] S. Black, & L. Lynch, How to Compete: The Impact of Workplace Practices and Information Technology on Productivity. Review of Economics and Information Technology on Productivity. Review of Economics and Information Technology on Productivity. Statistics, 83(3), (2001) 434-445.
- [23] B. Godin. (2004). The New Economy: what the concept owes to the OECD. Research Policy, 33(5), (2004) 679-690.
- [25] G. Hardin, The Tragedy of the Commons. Science, 162(3859), (1968)1243-1248.
- [26] Millennium Ecosystem Assessment, Ecosystems and human wellbeing. Island Press, Washington DC, 2005
- [27] S. Lewis & M. Maslin, Defining the Anthropocene. Nature, 519(7542), (2015) 171-180
- [28] G. Blainey, The Tyranny of Distance: How Distance Shaped Australia's History. Sydney: Macmillan, 2001.
   [29] OECD, Measuring Innovation a new perspective. OECD Publishing, Paris, 2010.
- [30] OECD, OECD Science and Technology Outlook, OECD Publishing, Paris, 2014.
   [31] OECD, Collaboration on innovation in OECD Science, Technology and Industry Scoreboard 2015: Innovation for growth and society, OECD Publishing, Paris, 2015.
- [32] M. Grayson, & S. Pincock, Nature Index Collaboration, Nature, 527, (2015), 49
- [33] Personal conversation August 2016
- [34] National Innovation and Science Agenda, National Innovation and Science Agenda Report, Government of Australia, 2016.
- [35] D. Kahneman, Thinking fast and slow. London: Penguin, 2011.
- [36] P. Dolan, Happiness by design. London: Penguin, 2014.[37] P. Senge, The fifth discipline. London: Random House, 2006.
- [38] C. Christensen, C. M. (1997). The innovator's dilemma : when new technologies cause great firms to fail. Harvard Business School Press, 1997. [10] J. Nee, & Y. Cao, (1999). Path dependent societal transformation: Stratification in hybrid mixed economies. Theory and Society, 28(6), (1999) 799-834.
- [40] M. Burbridge, & G. Morrison, Sustainable living labs. In print.
   [41] A. Pascual-Leone, D. Cohen, J. Brasil-Neto, A. Cammarota, & M. Hallett, Modulation of muscle responses evoked by transcranial magnetic stimulation during the acquisition of new fine motor skills. Journal of Neurophysiology, 74(3), (1995) 1037 1045.
   [42] H. Schaffers, & P. Turkama, Living Labs for Cross-Border Systemic Innovation. Technology Innovation Management Review, 2(9), (2012) 25-30.
- [43] S. Hajkowicz, H. Cook & A. Littleboy, Our future world global megatrends that will change the way we live. CSIRO, Brisbane, 2012.
   [44] L. Kyoung-Joo, From interpersonal networks to inter-organizational alliances for university-industry collaborations in Japan: the case of the Tokyo Institute
- of Technology. R&D Management, 41(2), (2011) 190-201. [45] A. Gaziulusoy, C. Ryan, S. McGrail, P. Chandler & P. Twomey, Identifying and addressing challenges faced by transdisciplinary research teams in climate change research. Journal of Cleaner Production. (2015)
- [46] B. Dutilleul, F. Birrer & W. Mensink, Unpacking European Living Labs: Analysing Innovation's Social Dimensions. 2010, 4(1), (2010) 26 34.
  [47] Y. Voytenko, K. McCormick, J. Evans, & G. Schliwa, Urban living labs for sustainability and low carbon cities in Europe: towards a research agenda.
- Journal of Cleaner Production, (2015) [48] T. Coenen, S. van der Graaf, & N. Walravens, Firing Up the City - A Smart City Living Lab Methodology. Interdisciplinary Studies Journal, 3(4), (2014),
- 118-128. [49] M. Gray, M. Mangyoku, A. Serra, L. Sánchez & F. Aragall, Integrating Design for All in Living Labs. Technology Innovation Management Review, 4(5),
- (2014), 50-59 [50] C. Liedtke, M. Welfens, H. Rohn, & J. Nordmann, Living Lab: user-driven innovation for sustainability. International Journal of Sustainability in Higher
- Education, 13(2), (2012) 106-118. [51] J. Geibler, L. Erdmann, C. Liedtke, H. Rohn, M. Stabe, S. Berner & K. Kennedy, Exploring the Potential of a German Living Lab Research Infrastructure for the Development of Low Resource Products and Services Resources 3, (2014) 575 - 598.
- [52] A. Stahlbrost, A set of key principles to assess the impact of Living Labs Int. J. Product Development, 17 (2012).
- [53] F. Paula & H. Pernilla, The Habitation Lab: Using a Design Approach to Foster Innovation for Sustainable Living. Technology Innovation Management Review, 3(11), (2013).
- [54] IPCC, Summary for Policymakers. In: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, London and New York. 2013. [55] E. Almirall, M. Lee, & J. Wareham, Mapping Living Labs in the Landscape of Innovation Methodologies. Technology Innovation Management Review,
- 2(9), (2012), 12-18.
- [56] C. Dell'Era & P. Landoni, (2014). Living Lab: A Methodology between User-Centred Design and Participatory Design. Creativity & Innovation Management, 23(2), (2014) 137-154. [57] S. Leminen, Q&A. What Are Living Labs? Technology Innovation Management Review, 5(9), (2015) 29-35.
- [58] R. Brankaert, E. Ouden & A. Brombacher, Innovate dementia: the development of a living lab protocol to evaluate interventions in context. info, 17(4), (2015) 40-52.
- [59] S. Dhakal, M. Mahmood, A. Wiewora, K. Brown & R. Keast, The Innovation Potential of Living-Labs to Strengthen Small and Medium Enterprises In Regional Australia, 2013
- [60] J. Dvarioniene, I. Gurauskiene, G. Gecevicius, D. Trummer, C. Selada, I. Marques & C. Cosmi, Stakeholders involvement for energy conscious communities: The Energy Labs experience in 10 European communities. Renewable Energy, 75, (2015) 512-518. [61] F. Kehayia, B. Swaine, C. Longo, S. Ahmed, P. Archambault, J. Fung, & T. Poldma, Creating a rehabilitation living lab to optimize participation and
- inclusion for persons with physical disabilities. ALTER European Journal of Disability Research 8(3), (2014) 151-157. [62] W. Lin, C. Lin, Y. Wang & R. Chen, The Transformation of Users in Living Lab Construction: The Case of Eco-City Living Lab. International Journal of Automation and Smart Technology, 2(3), (2012) 231 - 239.

1731

186

- [63] R. Olivier, S. Dimitri & B. Pieter, Exploring the Benefits of Integrating Business Model Research within Living Lab Projects. Technology Innovation

- [63] R. Olivier, S. Dimitri & B. Pieter, Exploring the Benefits of Integrating Business Model Research within Living Lab Projects. Technology Innovation Management Review, 5(12), (2015)
  [64] S. Sauer, Do Smart Chites Produce Smart Entrepreneurs? Journal of theoretical and applied electronic commerce research, 7, (2012) 63-73.
  [65] D. Soetanto, & M. van Geenhuizen, Social networks, university spin-off growth and promises of 'living labs'. Regional Science Policy & Practice, 3(3), (2011), 305-321.
  [66] M. Bourgault, Developing professional competencies using a Living Lab approach: an exploratory study in the field of management education International Journal of Product Development, 17(1/2), (2012), 76 93.
  [67] L. Hakkarainen & S. Hyysalo, How Do We Keep the Living Laboratory Alive? Learning and Conflicts in Living Lab Collaboration. Technology Innovation Management Review, 3(12), (2013) 16-22.
  [68] N. Hawk, M. Romine & G. Bartle, The Living Labs: Innovation in Real-Life Settings. Quarterly Review of Distance Education, 13(4), (2012), 269-270.
  [69] L. Katri-Liis, K. Merle, & T. Erik, Problems of Initiating International Knowledge Transfer: Is the Finnish Living Lab Method Transferable to Estonia? International Journal J

## B.4 Publication 4

Burbridge, M. (2018). A Study of Australian Universities' Collective Response to Climate Science. In: Sayigh A. (eds) Transition Towards 100% Renewable Energy. Innovative Renewable Energy. Springer, Cham. https://doi.org/10.1007/978-3-319-69844-1

### Do as I say; don't do as I do, let alone do as I've done. A study of Australian Universities' collective response to climate science.

### Mike Burbridge

PhD Student, Curtin University Sustainability Policy Institute, Curtin University, GPO Box U1987, Perth 6845

* Corresponding author: mike.burbridge@postgrad.curtin.edu.ai

### Abstract

*There is no longer a debate about whether climate change is happening due to anthropogenic emissions of carbon – the debate now is around the scale of the likely impact, how to mitigate further increases in temperature and what to do adapt to those impacts.* 

Not surprisingly, universities have been at the forefront of the climate science. Australian Universities are established to do three things: teach, research and to engage. To varying degrees climate change and its impacts has penetrated all three agendas. Universities are clearly at the forefront of teaching and researching climate change and engaging with society about the science based actions that need to be taken to combat to both mitigate and adapt to its impacts. However, interestingly only 3 Australian universities commit themselves to absolute reductions in carbon from their operations.

This paper will look at the performance of universities in reducing carbon emissions from their operations and seek to identify the reasons why universities do not adopt the science that they say the rest of the society should adopt. It also considers whether Universities performance in partnering internally has implications for their ability to partner with external organisations.

Keywords: climate change, leadership, environmental management, partnership

### 1. Introduction

The global science community is united about what is forcing climate change: "among papers expressing a position on anthropogenic, or human caused, global warming (AGW), an overwhelming percentage (97.2% based on self-ratings, 97.1% based on abstract ratings) endorses the scientific consensus on AGW." (Cook, 2013)

"Warming of the climate system is unequivocal, and since the 1950s, many of the observed changes are unprecedented over decades to millennia. The atmosphere and ocean have warmed, the amounts of snow and ice have diminished, and sea level has risen." (Edenhofer, Pichs-Madruga et al. 2014), 2014). In the 5th Assessment Report the IPCC states that to limit temperature increases to 2 degrees centigrade above pre-industrial levels will require a reduction in carbon (and other long lived greenhouse gases) of between 40 - 70% reduction from 2010 levels by 2050 and near-zero emissions by 2100. To restrict global temperature increases to 1.5 degrees centigrade would require between a 70 and 95% reduction in carbon and other long lived greenhouse gases on 2010 levels by 2050 (Edenhofer, Pichs-Madruga et al, 2014).

As shown in table 1 Australian Universities were widely involved within this international work with over a third of Australian Universities (excluding CSIRO and BOM) being involved as either authors or review editors for the three working groups that contributed to the final 5th Assessment Report ((IPCC 2014).

 Table 1 – Australian Universities contribution as author or review editor to IPCC 5th

 Assessment Report

Working Group I	Working Group II	Working Group III	
Physical Science Basis	Climate Change, Impact &	Mitigation of Climate Change	
	Vulnerability		
Australian National University	Griffith University	Australian National University	
Antarctic Climate and Ecosystems	Macquarie University	Curtin University	
CRC			
Monash University	University of Melbourne	Monash University	
University of New England	University of New South Wales	Murdoch University	
University of New South Wales	University of Queensland	University of New South Wales	
University of Tasmania	University of Tasmania		
	Victoria University		
Other Australian contributors			
Bureau of Meteorology	Australia/World Bank	Department of Agriculture	
CSIRO	Australian Antarctic Division		
	Bureau of Meteorology		
	Climate Risk Consultant		
	CSIRO		
	Data from IPO	CC (2014) 5th Assessment Report	

Universities in Australia exist to do three things: to teach, research and engage to get the research adopted. (Counsel 2017). Climate change penetrates all these areas – academics teach the science and social policy impacts, research the likely consequences of a changing climate and ways to adapt to and mitigate that change and, engage in public policy debate to encourage decision makers of the need to take decisions that are cognisant of the science produced by, amongst others, Australian universities.

This paper found minimal evidence of universities adopting their own research (this paper used climate science as a test case) or using their own capital spend to embed this, or any research or to deliver teaching and/or research outcomes in the operations of the university. This poses difficult questions for universities to answer – if climate science represents such an urgent threat to humanity why are Australian universities not applying it to their operations, and why should society adopt research they do not manage to adopt themselves?

It also suggests a further difficulty for Australia to embrace the opportunities of the knowledge economy. Universities in Australia are at the bottom of the OECD rankings for their ability to partner with business for innovation. (OECD 2013). This has been recognised as a key weakness of Australia's innovation framework by both universities (ATN, 2015) and by Commonwealth Government in the National Innovation and Science Agenda (2016). There is little evidence of universities being effective at internal partnering, and this lack of experience perhaps is one reason for their collective poor performance at external partnering as highlighted by OECD and recognised by Government and Universities. The Government and some Universities recognise this weakness by focussing on cross-cutting work in the national innovation strategy (Commonwealth of Australia, 2016) and in University corporate strategies.

However, it is worth noting 2 important points:

- Poor performance collectively should not be interpreted as no performance collectively. There are some universities albeit a small minority that are taking the issues seriously; and
- It is not just Universities that are poor at partnering. Whilst Australian business ranks at above the OECD average for funding research at Universities (OECD 2013) it is also collectively weak at engaging with suppliers and customers for innovation (Gahan 2016).

The paper concludes there are 2 clear benefits for universities in adopting their own research where possible – one is that they would be able to more clearly understand the difficulties organisations have in adopting or commercialising research into business models; the other is that universities would

have a stronger voice if they were in a position to encourage others to do as they have done rather than encouraging others to do what the science says, but that they have not done.

### 2. Methodology

There are 43 Universities in Australia. All Australian Universities websites were accessed between 18 -20 November 2016 to download their 2015 annual report and financial statements and their forward looking corporate strategies (dates ranging up to 2025). Also, all Government/ University performance agreements were accessed from Department of Education and Training website covering 2014 -2016 Mission-based Compacts. A limit of 30 minutes was allocated to each university to find and download the data. Where the data was unavailable online (either not found or not published) it was requested by email; follow-up emails were not sent.

All available annual reports and strategies were read and the public commitments in those reports (for example published statements, targets and performance) to partnership and adopting research were recorded and tabulated. Research on climate change was taken as the example for adoption of research due to the agreed status of the science (Cook, 2013), and the applicability of it to campus development (and therefore to capital campus projects). The documents were also searched by keywords (carbon, sustainable, greenhouse, climate change and variants) to understand the importance of sustainability and climate change to each university. They were also searched to understand the importance of collaboration within the university, to the university (words searched were cross-, inter-, multi-, trans-disciplinary and variants) as well as externally with industry, society and government. Each Universities capital spend committed for 2016 was recorded to understand the potential of the sector to use their own capital spend to adopt their own research (for example a university with no forecast capital spend could find it more difficult to integrate teaching and research into campus development).

Due to the incomplete nature of performance data, publicly available performance data was accessed to develop an indicative understanding of the rate of change of emissions of Australian Universities and thereby to understand their success in translating research into impact.

### 3. Results

Australia has 43 universities, of which 3 committed themselves to absolute reductions in greenhouse gas emissions up to 2020. Based on evidence from Mission-based Compacts covering 2014 - 16, annual report and financial statements for 2016, strategies and corporate strategies (covering a variety of start and end dates but all accessed cover years after 2016) there were essentially 4 groupings of universities:

- 1. Those with commitments to carbon constraint as well a target for absolute reductions in carbon;
- 2. Those with commitments to carbon constraint as well a target for reductions in carbon;
- 3. Those with commitments to carbon constraint but no evidence as to how to deliver the commitment; and
- 4. Those universities that do not mention carbon or emissions in any of their documentation.

### Table 2: Australian University commitments to carbon constraint.

	No.
Universities with a science based commitment ¹	3
Universities with a carbon commitment and published target	12
Universities who make statements but do not provide evidence as to what success looks like	15
Universities who do not mention carbon or emissions	11

¹ This includes RMIT, UTS and Charles Sturt University. CSU is Australia's only carbon neutral university, but there was a lack of clear emission reduction targets at the same time as maintaining carbon neutrality

Total		41 ²
	Data from University annual and financial report, strategic plans and mission-base	d compacts

Table 2 may portray an unduly positive picture of universities' commitment to embracing climate science. Some commitments would not pass academic, societal or government scrutiny or commentary. Statements like "we will improve energy efficiency between 2015 – 2020", we will "adhere firmly to the principles of sustainability in all we do" or "we will reduce [the university's] operational carbon footprint" are insufficient to be convincing that such universities' are genuinely tackling the issue.

There was also differences within University groupings. RMIT has "since 2008, committed to a target in partnership with Australian Technology Network of Universities to reduce greenhouse gas emissions by 25% by 2020 compared to 2007 as a base". However, this target is not shared by other ATN members (see table 3)

Tuble of Commitment of fills, memoris to curben foundation the zoto	Table 3.	Commitment	of ATN	members to	carbon	reduction	targets.
---------------------------------------------------------------------	----------	------------	--------	------------	--------	-----------	----------

ATN member	Target		
RMIT	reduce greenhouse gas emissions by 25% by 2020 on 2007 as a base		
University of Technology Sydney	30% reduction in GHG based on 2007 by 2020/21		
Queensland University of	11% reduction in $CO_2$ per gross floor area (m ² ) by 2019 on 2013		
Technology			
Curtin University of Technology	No target		
University of South Australia	No target		
Data from University annual and financial report, strategic plans and mission-based compacts.			

Of the nine Universities that scored the highest ranking of 5 in 2-digit Field of Research code 05 – Environmental sciences in the Australian Research Council's Excellence in Research Australia (Australian Research Council, 2016) seven do not mention carbon or emissions in their corporate strategies; in the same vein 10 of the 12 universities involved in the International Panel on Climate Change 5th Assessment Report (IPCC, 2013) did not include reference to emissions or carbon in their corporate strategies.

There are 10 universities that are required to publish their emission date by the National Greenhouse and Energy Reporting Act (Office of Parliamentary Counsel, 2017a). Analysis of the data shows an increase in total emissions of 4.6% between 2010/11 and 2014/15 as set out in Table 4. Currently this is the only data available.

	Saona 1 CHC	Saona 2 CHC	Total CHC
	(t CO2-e)	(t CO2-e)	(t CO2-e)
2010 - 11	103,853	800,006	903,859
2011 - 12	94,758	821,337	916,095
2012 - 13	106,674	826,123	932,797
2013 - 14	103,502	839,198	942,700
2014 - 15	104,316	841,373	945,689
% change over 2010	+0.45%	+5.17%	+4.63%

Table 4: Greenhouse gas emission from 10 largest University emitters.

Data from:

http://www.cleanenergyregulator.gov.au/NGER/National%20greenhouse%20and%20energy%20reporting%20d ata

² Documentation was not available for Torrens University or UCL (Australia). Incomplete documentation for Notre Dame, University of Tasmania.

It is interesting that the pre-eminent research universities working in the environmental science or climate change fields do not apply the science to their own organisations, but are committed to be part of the public debate about such grand challenges.

All universities made commitments to collaborate, debate or work in partnership with a combination of industry, government, society or the communities they serve to develop new research, translate existing research or to improve the wellbeing of society. This commitment to partner with organisations external to the university is a universal desire in the university sector. However, as the OECD, the University sector and Government acknowledge Australian Universities are worse than their OECD counterparts in partnering with external organisations.

The same appears to be the case when considering internal partnerships across the university or to using university spend to deliver teaching or research outcomes or commercialise research.

Australian Universities committed themselves to spend over \$1.5bn on capital works during 2016 (see table 5). The Australian Research Council – Australia's principle grant provider – between 2011 and 2013 granted funding worth under \$100m to universities to research the built environment and design under Field of Research code 012 (ARC, 2016 p86). There are 8 universities who have committed to using their campuses as a living lab, apply their knowledge to the campus or to help deliver teaching and research outcomes. Those universities will be spending \$452m on property, plant and equipment in 2016. There is no evidence in the documents about how the living labs will operate, or how the spend will contribute to teaching and research outcomes. The University of Queensland's corporate strategy is the only one that contains a forward commitment to use the campus as a living lab. (UQ Strategic Plan, 2014 - 17 p21). There are 17 universities that commit themselves, through their corporate strategic plans to cross-silo working. The remaining 26 universities do not make commitments to either cross, inter, multi, or trans disciplinary work.

### **Table 5: Capital commitments of Australian Universities**

	Capital expenditure ³
	during 2016 (,000)
Total spend on property, plant and equipment payable within 12 months (all	\$1,542,878
universities)	
Of which capital spend of Universities committed to either using campus as a	\$452,737
living lab or applying their knowledge to the campus (UQ, UniSA, UMelbourne,	
CurtinU, UCanberra, UAdelaide, RMIT, U Tasmania)	

### 4. Conclusions

The evidence from this study suggests that Universities have some work to do to put themselves in a position where they can partner effectively with the industry or government to innovate or commercialise their research. Less than 20% of Australian universities are using their campuses to deliver teaching and research outcomes or as a living lab to innovate. Only one university is committed to doing this in the future. Yet Universities will be spending over \$1.5bn during 2016 on capital infrastructure. If this infrastructure spend is not used to also drive teaching and research outcomes, or to showcase how to adopt research then it is being spent inefficiently.

Only 3 universities make a commitment for an absolute reduction in carbon emissions, in line with climate science. 90% of universities do not make this commitment (and over 25% of universities do not mention carbon or emissions in their documentation). Yet it is the same Universities that research the subject and pressure Ministers and society at large to take action on climate change.

³ Capital expenditure contracted for at the reporting date but not recognised as liabilities

Finally, the apparent struggle universities have to partner with themselves could be dismissed as irrelevant. However, Universities are being pressured into developing more effective partnerships with business; if they have little or no experience in either partnering (rather than contracting) nor in implementing or commercialising research then they will not understand the problems businesses face, nor how to overcome such challenges. In an environment where Government is committed to attaching an increasing share of funding to working in partnership with business it would seem adroit for universities to get as much experience as possible in working in partnership across organisations and disciplines. This is not just a job for academics but rather for institutions.

### Acknowledgement

This work has been funded by a PhD scholarship from the CRC Low Carbon Living. The project is looking to develop a business model to help universities, business and government understand the benefits of high-performance, evidence based innovation. With thanks to Abigail Burbridge for assistance in finding documentation.

### References

Australian Research Council, (2016). 2015-16 State of Australian University Research Volume 1 ERA National Report. ARC, Canberra,

Department of Education and Training, Commonwealth of Australia, 2016 Universities' 2014-16 Mission-based Compacts. Accessed at https://docs.education.gov.au/node/34873

Commonwealth of Australia (2015). National Innovation and Science Agenda. Commonwealth of Australia, Canberra.

Edenhofer, O., R. Pichs-Madruga, Y. Sokona, E. Farahani, S. Kadner, K. Seyboth, A. Adler, I. Baum, S. Brunner, P. Eickemeier, B. Kriemann, J. Savolainen, S. Schlömer, C. von Stechow, T. Zwickel and J.C. Minx (eds.) (2014). Summary for Policymakers. In:Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

Gahan, P., Adamovic, M., Bevitt, A., Harley, B., Healy, J., Olsen, J.E., Theilacker, M. (2016). Leaership at work: do Australian leaders have what it takes? Australia, Centre for Workplace Leadership, University of Melbourne.

IPCC (2014). "Fifth Assessment report - authors and reviewers." Retrieved 1 March 2017 from http://www.ipcc.ch/pdf/ar5/ar5_authors_review_editors_updated.pdf.

John, C., et al. (2013). "Quantifying the consensus on anthropogenic global warming in the scientific literature." <u>Environmental Research Letters</u> 8(2): 024024.

OECD (2013). Science, Technology and Innovation scoreboard. Paris, OECD.

Office of Parliamentary Counsel (2017). Higher Education Support Act. <u>Compilation 62</u>. OoPC, Canberra.

Office of Parliamentary Counsel (2017a) National Greenhouse Energy and Reporting Act, 2007. Compilation 18, Office of Parliamentary Counsel, Canberra.

University of Queensland (2014), Strategic Plan: 2014 – 2017. University of Queensland, Brisbane Australia http://www.uq.edu.au/about/docs/strategicplan/StrategicPlan2014.pdf

Source data.

Emissions data:

http://www.cleanenergyregulator.gov.au/NGER/National%20greenhouse%20and%20energy%20reporting%20d ata

All Mission based compacts can be found at: https://docs.education.gov.au/node/34873

University data.

Uni	Annual report – 1 st entry
	Strategic plan – $2^{nd}$ entry
	http://www.acu.edu.au/about_acu/our_university/publications/annual_reports
ACU	https://www.acu.edu.au/about_acu/our_university/mission_and_profile/strategic_plan_201
	<u>5-2020</u>
ANU	http://www.anu.edu.au/about/plans-reviews
	http://www.anu.edu.au/files/review/anu-2020-strategy.pdf
Bond	https://bond.edu.au/about-bond/university/introducing-bond/bond-university-annual-report
Uni	https://bond.edu.au/about-bond/university/introducing-bond/bond-mission-strategic-plan
CMU	Not found on website
COLL	https://www.cqu.edu.au/about-us/structure/governance/annual-report
CQU	http://content.cqu.edu.au/Policy/policy_file.do?policyid=2988
CDU	http://www.cdu.edu.au/about/annual-report
CDU	https://www.cdu.edu.au/about/strategic-plan
CSU	http://www.csu.edu.au/about/publications/annualreports
CSU	https://www.csu.edu.au/unistats/university-strategy [2015-16]
CUT	http://about.curtin.edu.au/policy-governance/annual-reports/
COI	http://trove.nla.gov.au/version/196899916
Dee	http://www.deakin.edu.au/adi/strategic-plan/annual-reports
Dea	http://www.deakin.edu.au/adi/strategic-plan
ECU	http://www.ecu.edu.au/about-ecu/reports-and-plans/annual-reports
	http://intranet.ecu.edu.au/ data/assets/pdf file/0005/730328/ECU12439 StrategicPlan A
	4 WEB.pdf
Fad	https://federation.edu.au/staff/governance/plans-publications-policies/organisational-data
геа	https://federation.edu.au/ data/assets/pdf file/0010/284248/FedUni StrategicPlan.pdf
FI	http://www.flinders.edu.au/about/governance/annual-reports/
1.11	http://2025.flinders.edu.au/
Gri	Https://www.griffith.edu.au/about-griffith/governance/plans-publications/annual-report
UII	https://www.griffith.edu.au/about-griffith/governance/plans-publications
ICU	https://www.jcu.edu.au/about-jcu/annual-report
JCU	https://www.jcu.edu.au/about-jcu/university-plan-2013-2017
LaT	http://www.latrobe.edu.au/council/resources
Lai	http://www.latrobe.edu.au/about/downloads/La-Trobe-Strategic-Plan-November-2015.pdf
Mac	http://www.mq.edu.au/about/about-the-university/governance/annual-reports
	http://mq.edu.au/our-
	university/pdf/Macquarie_University_A_Framing_of_Futures_revised.pdf
Mon	http://www.monash.edu/about/who/publications/annual-report
IVION	http://www.monash.edu/about/who/strategic-plan
Mur	http://www.murdoch.edu.au/About-us/Annual-report/
	http://www.murdoch.edu.au/About-us/Strategic-plan/
OUT	https://www.qut.edu.au/about/governance-and-policy/annual-report
QUI	https://www.qut.edu.au/about/strategic-ambitions/blueprint-for-the-future
RMIT	http://www.rmit.edu.au/about/governance-and-management/governance/annual-reports
	https://www.rmit.edu.au/about/our-strategy
SCU	http://scu.edu.au/docs/annual_report/
300	http://scu.edu.au/strategicplan/
Swin	http://www.swinburne.edu.au/about/strategy-initiatives/annual-report/

	http://www.swinburne.edu.au/about/strategy-initiatives/2020-plan/			
Tor	Not found on website, requested 20/10/16. No response			
LiaA	https://www.adelaide.edu.au/publications/corporate/			
UUA	https://www.adelaide.edu.au/VCO/beacon/			
	https://www.canberra.edu.au/about-uc/policy-and-legislation/key-university-			
UoC	documents/annual-reports			
	http://www.canberra.edu.au/about-uc/strategic-plan			
UCL	Not found on website, requested 20/10/16. No response			
UDiv	https://www.divinity.edu.au/news-events/2016/04/14/2016-annual-report/			
	https://www.divinity.edu.au/university-of-divinity/about-us/our-vision-and-mission/			
UMel	http://publications.unimelb.edu.au/			
	https://ihouse.unimelb.edu.au/home/features/our-response-to-growing-esteem			
IDE	http://www.une.edu.au/about-une/annual-reports			
UNE	http://www.une.edu.au/about-une/executive/vice-chancellor/strategic-plan/strategic-plan-			
LINIC	2016-2020/strategic-pian-2016-2020			
W	https://annuaireport.unsw.edu.au/2013/			
vv	http://www.2025.unsw.edu.au/			
UNe	https://www.newcastle.edu.au/about-uon/our-university/ainual-report			
w	direction/new-futures-strategic-nlan-2016-2025			
UND	Not found on Notre Dame's website, requested 20/10/16. No response			
UoQ	http://www.uq.edu.au/about/annual-reports			
	http://www.ug.edu.au/about/docs/strategicplan/StrategicPlan2014.pdf			
Uni	http://www.unisa.edu.au/About-UniSA/Corporate-publications/			
SA	http://www.unisa.edu.au/About-UniSA/Strategic-action-plan-2013-2018/			
	https://www.usq.edu.au/about-usq/about-us/plans-reports/annual-report			
USQ	https://www.usq.edu.au/about-usq/about-us/plans-reports/strategic-plan			
	http://www.usc.edu.au/explore/reports-and-publications/annual-report			
USC	http://www.usc.edu.au/explore/vision/strategy-quality-and-planning/strategic-plan-2016-			
	2020			
USyd	http://sydney.edu.au/about-us/vision-and-values/annual-report.html			
	http://sydney.edu.au/about-us/vision-and-values/strategy.html			
UTas UTS	http://www.utas.edu.au/university-council/university-reports			
	Link to strategy not working. Copy requested 20/10/16			
	http://www.uts.edu.au/about/uts-governance/official-publications/uts-annual-report			
	http://www.uts.edu.au/about/university/uts-strategic-direction			
UWA	http://www.uts.edu.au/about/university/uts-strategic-direction			
	http://www.web.uwa.edu.au/data/assets/pdf_file/0010/2538343/114085-VICCHA-			
	StrategicPlan-V3.pdf			
UOW	http://www.uow.edu.au/about/publications/index.ntml			
	https://www.uow.edu.au/ourpurpose/			
WSU	https://www.westernsydney.edu.au/about_uws/leadership/mission_goals_strategic_plan			
	https://www.westernsydney.edu.au/about_uws/readersinp/governance			
VicU	https://www.vu.edu.au/about-vu/vision-mission-strategy			
1	mapon in the and a doubt i a riston mission states			

## B.5 Publication 5

 <u>Burbridge, M.</u> & G Morrison (2016). Happy Homes – the Relationship between Homes and mental wellbeing. Proceedings of the 7th International Conference on Energy and Environment of Residential Buildings, edited by Miller, W., Susilawati, C. and Manley, K. Brisbane: Queensland University of Technology, Australia.

## Happy Homes – the Relationship between Homes and Mental Wellbeing: a Review of the Literature



Mike Burbridge Curtin University Sustainability Policy Institute Australia Mike.burbridge@gmail.com

Professor Gregory Morrison, Curtin University Sustainability Policy Institute, Australia, greg.morrison@curtin.edu.au

### Abstract

**Purpose:** This paper set out to uncover the advice available to help people take effective action within our home to improve mental health. The literature and professions are virtually silent on the issue. The professional advice is often the opposite suggesting we should get out of our homes - go for a walk, exercise, play sport, go to the cinema, meet friends, socialise and don't isolate yourself. There is nary any advice about what we can do to our homes to help maintain our mental health. Our home - the physical space where we spend large amounts of energy and time is largely an empty shell for the mental health industry. The message currently presented appears to be "remember to close the door as you leave ... to get better". Safe and secure housing is a fundamental pillar of an inclusive and productive society. Yet we don't know for sure what safe, secure, or good housing looks like.

**Approach:** This paper will begin that dialogue with a comprehensive literature review. The approach adopted to investigate this literature focussed on thinking about what a policy official might experience if they were tasked to develop guidance on steps to improve housings' impact on mental health. Such an individual would not necessarily be aware of the extent of the literature, or of academic disciplines. This approach both made the literature review problematic, but also in some ways also produces a useful insight.

**Key findings**: The paper concludes that there are three issues that should shape future research: first is the need for transdisciplinary translational research; second is to focus initially on the needs of the resident before the bricks and mortar; third is to endeavour to include the social pillar of sustainable development alongside the economic and environmental.

**Originality:** This paper is original as it seeks to start a conversation about what self-help measures people can adopt within their homes to protect or enhance their mental wellbeing

Keywords: mental wellbeing, mental health, housing, homes, transdisciplinary research, translational research



HealthyHousing2016: Proceedings of the 7th International Conference on Energy and Environment of Residential Buildings, November 2016, edited by Miller, W., Susilawati, C. and Manley, K. Brisbane: Queensland University of Technology, Australia. DOI: http://dx.doi.org/10.4225/50/58107c8eb9c71

### 1. Introduction

The prevalence of mental health issues in the community is a concern. Australia suffers from 8 suicides per day with a disproportionate representation in the Aboriginal community. Society's response has been evolving. Mental health is now a topic for public debate. However, the role of the space where people live - the home - has received relatively little attention. The self-help advice that is available routinely refers to the importance of socialising, taking up a hobby and generally getting outside (eg BeyondBlue, 2014, or Better Health Victoria, 2016). Unlike the environmental and economic pillars of sustainable development, and unless you are in the care of the state, there is little to help builders, renovators, landlords, tenants, carers or home owners to take actions to improve the potential for their living space to either prevent a slide down the mental health continuum or just to enjoy better mental well-being.

Table 1: Mental health continuum (developed from Bridging the Distance (2016) and Mental Health Commission of Canada (2016)).

Self-care and	social support	Intervention by health care sector		
Healthy	Reacting	Injured	Ш	
Normal functioning	Common and reversible	Significant functional impairment	Clinical disorder Severe and persistent functional impairment	
Normal mood fluctuations. Takes things in stride. Consistent performance. Normal sleep patterns. Physically & socially active. Usual self- confidence Comfortable with others	Irritable/Impatient Nervousness, sadness, increased worrying. Procrastination, forgetfulness. Trouble falling asleep Lowered energy. Difficulty in relaxing. Intrusive thoughts. Decreased social activity	Anger, anxiety. Lingering sadness, tearfulness, hopelessness, worthlessness. Preoccupation. Decreased performance at school or work. Significantly disturbed sleep (falling asleep and staying asleep). Avoidance of social situations		

HealthyHousing2016: 20-24 November, 2016, Queensland University of Technology, Brisbane, Australia.
Happy homes - the Relationship between Homes and Mental Wellbeing: a Review of the Literature

	Possible actions to	take at each stage	
Focus on the task at	Recognise limits	Identify and	
hand	Get adequate sleep,	understand own signs	
Break problems into	food and exercise	of distress	
manageable chunks		Talk with someone	
The office and southers	Engage in helpful	Oracla hada	
support systems	coping strategies	Seeк пер	
support systems	Identify and minimise	Seek social support	
Maintain healthy	stressors	instead of	
lifestyles		withdrawing	

## 2. Background

There is longstanding acknowledgement that a person's physical and mental health can be impacted by the place where they live (Chapin, 1951; Novick, 1971). Since human's earliest history those who developed good interventions to keep the unwanted out (hungry animals, weather, enemies) and the wanted in (warmth, family, food, community) survived.

Without good housing people have little chance of maintaining meaningful activities and supportive relationships (Browne & Hemsley, 2010). Housing gives people a physical and cultural space in society and can influence how, and what, they contribute to society (Bendiner-Viani & Saegert, 2007).

# 3. Methodology

The approach adopted to conduct this literature focussed on providing support for a policy official if they had been tasked to develop guidance on possible steps to improve the houses' impact on the occupants' mental health. Such an individual would not necessarily be aware of the extent of the literature, or the academic disciplines. This approach both made the literature review problematic, and in some ways also produces a useful insight.

Papers were selected through a search of the literature using Web of Science with the terms "housing" and "mental health" (and variants) searching both paper title and content. Not surprisingly this produced a large number of papers. Only those papers that illuminated the relationship between housing and mental wellbeing were selected for deeper analysis.

'Relevant papers' were defined as those that demonstrated an evidenced link between action by a party (e.g. decision maker, policy maker, designer, carer, home owner, or occupier) and the impact on the person living in the house's mental wellbeing. This meant that the issues defined as 'self-care' and 'social support' could be interrogated as well as 'intervention by health care sector' (see Table 1).

These papers came from a range of disciplines largely within the health, planning and built environment sectors. For the purposes of this paper 'housing' was broadly defined so as not to limit by physical structure or tenure and refers to a physical built space designed for human habitation.

'Mental health' has been similarly broadly defined to capture the impacts of interventions across the mental health continuum (see Table 1) that help people move away from treatment and intervention with a particular focus on the self-care and social support end of the spectrum

A total of 96 papers were analysed in detail with findings listed and analysed. Three categories of findings emerged from the analysis relating to the impact of housing on an occupants mental health. The categories were: scope of influence over own life; quality of the home (both build and use) and quality of the neighbourhood.

Topics such as homelessness, alcohol, drugs and their impact on housing and mental health were not included in this review (despite the volume in the literature). Interestingly the large corpus of work on biophilics was only covered tangentially by this literature review in spite of the long established and positive link between impacts of nature on mental health (Sodelund & Newman, 2015).

## 4. Discussion

There is little in the literature about what healthy people can do within their homes to protect and improve their mental wellbeing (either in new, or existing homes). The main focus of the literature with regard to housing and mental health was:

- 1. Focus on those marginalised in society (be they at the 'intervention' end of the mental health continuum or in need of housing assistance);
- 2. The issues that can trigger a decline in physical and/or mental health of home dwellers; and
- 3. Descriptive research (i.e. we did this and that happened) rather than translational research (i.e. translating the evidence into policy advice).

There is a pervading tendency in the literature of viewing the house as primarily a tool for delivering policy (e.g. meeting carbon targets for the benefit of society). As such much of the discourse is about how to use the home to deliver benefits to a non-static combination of the individual, community and wider society.

In many ways this is what joined-up, or co-ordinated, Government is about – using the most effective tools to deliver policy outcomes. However it also means that the interests of the individual are not subjugated to the interest of the community or wider society. Somehow all interest need to be met – this is why policy making is not simple.

There was no unity in the review about the definition of what a quality house or local neighbourhood looked like, with the majority of the papers focussing on 'poor quality'. However, there are issues identified that are associated with a positive impact on mental health which should therefore be <u>embraced</u> by society. There are also issues identified that negatively impact mental health (which should therefore be <u>avoided</u> in future). There was a final set of issues identified that were currently too <u>complex</u> to be set into a binary function of embrace/avoid and require further work. Under each section we have included a table with a very brief summary of each paper which have been accordingly categorised as 'embrace', 'avoid' or 'complexity'.

This review revealed that there is not a single 'thing' that can be defined as good housing in terms of promoting mental health. Rather it is the result of three interconnected evidenced categories where intervention could be effective in helping to improve an individual's mental health.

These go wider than the simple fabric of the house and the categories are:

• the extent of an individual's influence over their own lives;

#### Burbridge, M

Happy homes - the Relationship between Homes and Mental Wellbeing: a Review of the Literature

- the quality of the individual's housing (which subdivides into both quality of build and of use); and
- the quality of the local neighbourhood.

This is illustrated in Figure 1 which suggests that from the perspective of the individual 'good' housing does not exist in isolation, but is probably better described as a combination of influence, location and quality. Categories may be mutually compensatory (ie. a good neighbourhood may mask a poor dwelling). However, importantly the extent of this compensation is unclear. Policy makers might not have a similar view of good housing – and therein lies the heart of problem. A Commonwealth policy maker with responsibilities for delivering climate change outcomes may view 'good' differently from a care worker at shire level.



Figure 1: Categories influencing a person's mental health

#### Category 1: Influence over decisions

Several studies pointed towards the importance of helping people take control over their lives. This was as true for those suffering from drug and alcohol dependence (Allen, 2003) as to the management and maintenance of a residential complex (Mridha, 2015). Control over one's destiny is one of the key's to self-actualisation (Henwood et al., 2015) and papers that focussed on the issue pointed towards the importance of involving residents in the decision making process. However, this is not just about residents and includes involvement of a wider network of stakeholders (Connellan et al., 2013), and further the process of involvement should not be seen as a tick box exercise but rather a continuous endeavour as its impact is additive (Shenassa, 2007). In the leadership literature this is typically referred to as empowerment, but empowerment means that someone still has the power to withdraw authority; Marquet (2012) speaks of emancipation – that is the freedom to make a decision, be it good or bad.

Influence and control comes in many forms ranging from controlling the temperature of the house to decisions about where or how to live. It can mean the development of participatory systems so that a community can have some form of control over its destiny (Shrubsole et al, 2014). The key appears to be the balance between doing something to people and doing something with people, as coercive action is unlikely to be generally successful (Allen, 2003); whereas involvement seems to be helpful (see Table 3 below). Therefore, the development of capacity to deliver individual, designer, carer, health professional and community leadership and dialogue could be an important lever to help maintain mental health across the piece. It is the lack of influence over one's future that can be counter-productive.

Table 3 provides a summary of the literature relating to an individual's influence over decisions that will impact upon them (see methodology for explanation of table headings). It shows that there was general consensus that involvement in decisions is something to be embraced because of the link to a person's mental wellbeing and that it should become pervasive as the issue is not just about 'big' decisions, but also minor irritations (such as setting the temperature of the dwelling (Walker, 2013)).

Table 3: Theme: Influence over decision – summary of the literature

Embrace	Avoid	Complexity
Respect and status of residents (Bond et al., 2012)	Designs to avoid	Multiple uses and end users to
Inclusive design and self-actualisation (Henwood et al., 2015)	(Connellan et al., 2013)	consider (Kearns et al, 2015)
Include carers in decision making (Browne & Hemsley, 2010)	Impact of pre- set thermostats (Walker, 2015)	lssues around citizenship contribution to
Management and maintenance (Mridha, 2015)	Avoid coercion	society (Sylvestre et al., 2007)
Control is additive (Shenassa et al., 2007)	(Allen, 2003)	,
Participatory systems (Shrubsole et al., 2014)		
Integration of stakeholders in the design process (Connellan et al., 2013)		
Empowerment – community integration (Nelson et al., 1998)		
Seeing from the residents perspective (Smith et al., 2015)		

### Category 2: Quality of the neighbourhood

The quality of the local neighbourhood was specified by several authors for its impact on people's mental health and wellbeing. The causation ranged from the socio-economic status of the neighbourhood (Fitzpatrick, 2007) through to the impact of new front doors (Curl et al., 2015) to design of new neighbourhoods (Jones-Rounds et al., 2014). The issue of design of something new compared to improving something old was not explicitly covered together in the papers reviewed. However, it was dealt with in separate papers (eg Galea et al., (2007), Fitzpatrick (2007), Jones-Rounds et al., (2014)). That said the provision of quality outdoor space, regardless of the local neighbourhood can have a positive impact – which Gidlof-Gunnarsson and Ohrstrom (2007) demonstrated through the provision of noise 'free' areas, and others (Bendiner-Viani and Saegert, 2007) demonstrated in terms of 'good quality' public space. Understanding the causal pathways will be important in helping to design effective interventions with only intended consequences (Dunstan et al., 2013).

Table 4 summarises the literature on the quality of the community. This is a complex area where there is much descriptive work, but – in this review – less translational work. The dynamics between local community and quality of housing emerges with Jones-Rounds et al., (2014) arguing that a quality exterior environment can offset poor interior environment.

Happy homes - the Relationship between Homes and Mental Wellbeing: a Review of the Literature

Table 4: Quality of the community – summary of the literature

Embrace	Avoid	Complexity
Community safety (Blackman & Harvey, 2001)	Poor quality neighbourhood (Galea et al., 2007)	Quality of interactions with neighbours and quality of neighbourhood (Dunstan et al., 2013)
front doors (Curl et al., 2015)	Lack of green	Planning/health linkages (Wells et al., 2010)
Sounds (Andringa & Lanser, 2013)	space (Bertram & Rehdanz, 2015)	Recovery from stress following viewing green spaces (van den Berg et al., 2015)
Noise free areas (Gidlof-	,	SES of neighbourhood (Fitzpatrick, 2007)
Gunnarsson & Ohrstrom, 2007)	Loneliness resulting from residential	Impact on TV watching (MacLeod et al., 2008)
Green space (Bertram &	structures	
Rehdanz, 2015)	(Kearns et al., 2015)	Quality of neighbourhoods is important but so too is respect and status (Bond et al., 2012)
Location important rather than dwelling type (McCarthy, 1985)		Economic regeneration alongside property led regeneration (Curl et al., 2015)
		Good external environment can offset poor interior environment (Jones-Rounds et al., 2014))

### Category 3: Quality of the home

The literature revealed much in terms of the link between mental health and housing and develops two sub-themes relating to how the house is designed and how the house is used.

Poor design impacts mental health (Guite et al., 2006); poor housing can also have intergenerational impacts as children's emotional functioning can be impacted (Coley et al., 2013). It also helpfully identified issues that were applicable to only some sections of society – for example for people on low-moderate income 'unaffordable' housing has a negative impact on their mental health (Bentley et al., 2011); or age related mental health impacts and housing (Howden-Chapman et al., 2011).

This literature review did not find what a good or "normal" (Hogan and Carling, 1992) house was in terms of helping people attain or retain a healthy mental state. This is consistent with Bonnefoy (2007) and Evans et al., (2003). However there is some discussion about the development of a housing quality assessment tool (Keall et al., 2010). Although there is not agreement on whether housing quality is more important than housing type (Kearns et al., 2012).

Table 5 summarises the literature relating to the quality of the home. There is a wealth of research on the relationship between the home and mental wellbeing, but there is little that an individual might be able to adopt to enhance their mental wellbeing. Table 5: Quality of the home – summary of the literature

## Burbridge, M

# Happy homes - the Relationship between Homes and Mental Wellbeing: a Review of the Literature

Embrace	Avoid	Complexity
Heating and new front doors (Curl et al., 2015)	Unaffordable housing (Bentley et al., 2011)	Influence of sleep quality, indoor air quality, accessibility, obesity, mould, hygrothermal conditions and energy consumption on mental health (Bonnefoy, 2007)
Lack of draughts	Poor design and social features (Guite et al., 2006)	Clustering of ailments in deprived areas (Adamkiewicz et al., 2014)
(Blackman & Harvey, 2001)	Dampness (Hopton	Indoor conditions affect physical health (Veitch, 2008)
Engage with	& Hunt, 1996)	Housing tenure (Baker et al, 2013)
nature (Maller et al., 2009)	Affordability issue as get older	Healing environment MOBE (Hoisington et al., 2015)
	(Howden-Chapman et al., 2011)	Home repossession (Pevalin et al., 2009)
	Crowded homes (Solari & Mare,	Quality of interactions with neighbours and quality of neighbourhood (Dunstan et al., 2013)
	2012) Overcrowding	Step down community housing for people coming out of care (Barr et al., 2013)
	(Shenassa et al., 2007)	Homes and mental health making the policy links (Johnson, 2005)
	Violence, housing disarray and	Affordability and homelessness (Martin, 2015)
	childhood asthma (Suglia et al., 2010)	Health implications of multiple environmental risk exposure (Evans et al., 2003)
	Crowding (Wells & Harris, 2007)	Patient physical environment (van der Schaaf et al., 2013)
	Heat stress (Maller & Strengers, 2011)	Pleasurable and annoying sounds (Andringa & Lanser, 2013)
	Poor quality housing has higher impact on mental	Little research on the positive health effects of exposure to areas of good sound quality (van Kempen et al., 2014)
	wellbeing than housing type (Grigg	Improvements produce health benefits (Pevalin, 2009)
	et al., 2008)	Poor housing & children's emotional functioning (Coley et al., 2013)
		Unexpected consequences (MacLeod et al., 2008)
		Community, family and individual influencers (Curtis et al., 2013)
		Global south might need different research methodologies (Ferguson et al., 2013)
		House design big impact on use of environment (Marcheschi et al., 2016)
		Permanent housing has a positive impact (Smith, 2005)

#### Sub-category - how the house is designed

Housing design quality was shown to be key – good design has a big impact on how people use the house and development (Marcheschi et al., 2016), and the opposite (Guite et al., 2006). Simple things like daylight, view of nature (Maller et al., 2009) and noise (Andringa & Lanser, 2013), being damp free (Hopton & Hunt, 1996) are all important but so too are other variables (such as the interesting emerging work on microbiomes (Hoisington et al., 2015)). Whilst all are part of good design: all can be devalued through occupation. Designing a house to drive behaviours that promote mental health (Marcheschi et al., 2016) or how people use the house (Brunsgaard & Fich, 2016) are clearly important and bring us back to the issues of values raised by Sylvestre et al., (2007).

However new build whilst important will always be less in quantity than the number of existing homes. Housing improvements also deliver benefits (Pevalin et al., 2008) – such as reducing drafts (Blackman & Harvey, 2001), removing dampness (Hopton & Hunt, 1996), renovating bathrooms and kitchens (Curl et al., 2015) as well as providing heating (ibid). Issues such as preset thermostats (Walker, 2015) can have the opposite impact and could result in heat stress which itself has a negative impact on mental health (Maller & Strengers, 2011). Interestingly Curl et al., (2015) showed that the provision of heating can also negatively impact occupants' physical activity which in turn can impact on mental health.

#### Sub-category - how the house is used

Not being able to afford housing or not able to afford to run the house as designed are both stress inducers and can negatively impact a person's mental wellbeing. Such impacts are clearly delineated according to the ability to pay. Unaffordable housing is seen to be a key issue relating to mental wellbeing for those who are in the low-to-moderate income bracket (Bentley et al., 2011) as is the ability to pay bills (the impact of which changes with age (Howden-Chapman et al., 2011)). Although not directly related to cost, but linked, the link between housing tenure and mental well-being is unclear with studies demonstrating both sides of a different coin. Pevalin (2009) demonstrated that the mental health impact of home repossession is greater if it is owned rather than rented. Whereas Baker et al., (2013) found "little evidence of an intrinsic relationship between tenure and mental health". However, Smith (2005) found a strong link between secure, permanent accommodation and improving mental health.

How occupants use a house can impact on mental wellbeing. Overcrowding (Solari & Mare, 2012) is one such example of how occupants' use of their house can devalue good design and lead to a negative impact on mental health (Shenassa et al., 2007). Reducing overcrowding will improve mental health (Wells & Harris, 2007). Curtis et al. (2013) similarly identifies community, family and individual behaviours as being important influencers. For example production of 'annoying' noise (Andringa & Lanser, 2013) can mitigate design and lead to stress of inhabitants or neighbours.

The behaviour of housing occupants – violence or just disarray - can also have a negative impact on health (Suglia et al., 2015). The issue of how a house is used also links to the earlier discussion over control. Lack of control, or influence over the behaviour of inhabitants or neighbours can lead to negative mental health impacts.

#### 5. Limitations of this work

As the literature is limited about the modifications that can be made to people's homes to improve their mental well-being (or help to protect them from a decline) it is not possible to say with confidence that all such literature has been captured. However the lack of evidence in the literature is a theme that has been present – and commented upon – since the research of Evans et al. (2003).

Burbridge, M

The approach of this literature review however, did mean that much literature on the role of intervention in prisons, police stations and similar institutions were not included. Similarly, there is much in the literature about 'green offices' and how green office space may or may not help to improve the productivity of, and environment for office workers (e.g. Thatcher and Milner (2012)). This paper has not sought to identify the research in the green office space and apply it to private housing but there are very likely to be transferable lessons.

Similarly this review did not pick up the body of work on biophilics. This is a rich stream of work that could well inform translational research to deliver benefits along the mental health spectrum as well as cover new and existing homes.

The vast majority of the papers reviewed were based in a western context with Marais and Cloete (2014), Marais et al., (2013) and Mridha (2015) being the exceptions. It is important to recognise that the research methodologies used in western cultures are not necessarily directly transferrable to other regions of the world (Ferguson et al., 2013; Marais et al., 2013)

# 6. Conclusion

There are three over-arching observations from this literature review.

Firstly is the need for transdisciplinary (Jantsch, 1972) and translational research (Nelson et al., (1998), Osypuk (2014) and Veitch (2008)). Transdisciplinary because the literature review revealed the difficulty faced in capturing all the relevant disciplines; and translational because there is a continued and evidenced need to start developing interventions that can help people in the self-help/social support end of the continuum. Such translational research needs to focus on as a minimum the individual and their social support as delivery pathways. Ideally such studies should be longitudinal (Pevalin et al., 2008). Either way, it cannot be acceptable from a policy perspective to leave those issues largely uncovered.

Secondly, Novick (1971) spoke about the need to focus on the person. This still applies today, but we should be focussing first on the individual and their needs. In the UK Government there is a discourse of 'policy making as if people matter'. This needs to be the case for work on housing. A focus on the person first and then the bricks and mortar is essential. If policy makers do not do this then we will continue to deliver the sorts of social unintended consequences identified by Shrubsole et al. (2014).

Thirdly – linking both the above points – housing and mental health policy development, and particularly housing building codes and guidelines seem to focus predominantly on the economic and environmental pillars of sustainable development. Inclusion of the environmental pillar over the last couple of decades represents real and genuine progress; the next step needs to more explicitly include the social pillar.

# 7. Recommendation for future work

With so many interdependencies between the three themes of influence, design and community it is difficult to identify a way to break down the vectors of causation. However, a productive transdisciplinary route is available to the innovative university. A living lab – perhaps based in either new and/or existing student accommodation could help to understand the interdependencies within and between the themes. Such a project could start the process of developing translational research to help update building codes and design guidelines with those actions that can be undertaken, particularly at the self-help social care end of the continuum, to promote mental well-being.

# 8. Declaration

This work has been supported by a PhD scholarship from the Low Carbon Living CRC.

## 9. References

- Adamkiewicz, G., Spengler, J. D., Harley, A. E., Stoddard, A., Yang, M., Alvarez-Reeves, M., & Sorensen, G. (2014). Environmental Conditions in Low-Income Urban Housing: Clustering and Associations With Self-Reported Health. American Journal of Public Health, 104(9), 1650-1656.
- Allen, M. (2003). Waking Rip van Winkle: why developments in the last 20 years should teach the mental health system not to use housing as a tool of coercion. Behavioral Sciences & the Law, 21(4), 503-521.
- Andringa, T. C., & Lanser, J. J. L. (2013). How Pleasant Sounds Promote and Annoying Sounds Impede Health: A Cognitive Approach. International Journal of Environmental Research and Public Health, 10(4), 1439-1461.
- Baker, E., Bentley, R., & Mason, K. (2013). The Mental Health Effects of Housing Tenure: Causal or Compositional? Urban Studies, 50(2), 426-442
- Barr, W., Brown, A., Quinn, B., McFarlane, J., McCabe, R., & Whittington, R. (2013). How effective is high-support community-based step-down housing for women in secure mental health care? A quasi-experimental pilot study. Journal of Psychiatric and Mental Health Nursing, 20(1), 41-49.
- Bendiner-Viani, G., & Saegert, S. (2007). Making housing home. Places Forum of Environmental Design, 19(2), 72-79.
- Bentley, R., Baker, E., Mason, K., Subramanian, S. V., & Kavanagh, A. M. (2011). Association Between Housing Affordability and Mental Health: A Longitudinal Analysis of a Nationally Representative Household Survey in Australia. American Journal of Epidemiology, 174(7), 753-760. doi:10.1093/aje/kwr161
- Bertram, C., & Rehdanz, K. (2015). The role of urban green space for human well-being. Ecological Economics, 120, 139-152. doi:10.1016/j.ecolecon.2015.10.013
- Better Health Victoria (2016) 10 tips to stay mentally healthy.
- BeyondBlue (2014) Staying Well. BeyondBlue, Melbourne, Australia
- Blackman, T., & Harvey, J. (2001). Housing renewal and mental health: a case study. Journal of Mental Health, 10(5), 571-583.
- Bond, L., Kearns, A., Mason, P., Tannahill, C., Egan, M., & Whitely, E. (2012). Exploring the relationships between housing, neighbourhoods and mental wellbeing for residents of deprived areas. BMC Public Health, 12, 48
- Bonnefoy, X. (2007). Inadequate housing and health: An overview. International Journal of Environment and Pollution, 30(3-4), 411-429. doi:10.1504/ijep.2007.014819
- Bridging the distance (2016) The mental health continuum. Bridging the Distance, Canada. Last accessed 31 May 2016 <u>http://www.bridgingthedistance.com/pages/peer.html</u>
- Browne, G., & Hemsley, M. (2010). Housing and living with a mental illness: Exploring carers' views. International Journal of Mental Health Nursing, 19(1), 22-29
- Brunsgaard, C., & Fich, L. B. (2016). 'Healthy Buildings': Toward understanding user interaction with the indoor environment. Indoor and Built Environment, 25(2), 293-295.
- Chapin, F. S. (1951). Some Housing Factors Related to Mental Hygiene. American Journal of Public Health and the Nations Health, 41(7), 839-845.
- Coley, R. L., Leventhal, T., Lynch, A. D., & Kull, M. (2013). Relations Between Housing Characteristics and the Well-Being of Low-Income Children and Adolescents. Developmental Psychology, 49(9), 1775-1789.
- Connellan, K., Gaardboe, M., Riggs, D., Due, C., Reinschmidt, A., & Mustillo, L. (2013). Stressed Spaces: Mental Health and Architecture. Herd-Health Environments Research & Design Journal, 6(4), 127-168.
- Curl, A., Kearns, A., Mason, P., Egan, M., Tannahill, C., & Ellaway, A. (2015). Physical and mental health outcomes following housing improvements: evidence from the GoWell study. Journal of Epidemiology and Community Health, 69(1), 12-19.
- Curtis, S., Pain, R., Fuller, S., Khatib, Y., Rothon, C., Stansfeld, S. A., & Daya, S. (2013). Neighbourhood risk factors for Common Mental Disorders among young people aged 10-20 years: A structured review of quantitative research. Health & Place, 20, 81-90.
- Dunstan, F., Fone, D. L., Glickman, M., & Palmer, S. (2013). Objectively Measured Residential Environment and Self-Reported Health: A Multilevel Analysis of UK Census Data. PLoS ONE, 8(7).
- Evans, G., Wells, N. M., & Moch, A. (2003). Housing and mental health a review of the evidence and a methodoliogical and conceptual critique. Journal of social Issues, 59(3), 475 - 500.
- Ferguson, K. T., Cassells, R. C., MacAllister, J. W., & Evans, G. W. (2013). The physical environment and child development: an international review. Int J Psychology 48(4) 437
- Fitzpatrick, H. C. (2007). Supportive housing for mental health recovery: A bio-psycho-social approach. (MR34304 M.Arch.), University of Waterloo (Canada), Ann Arbor.

Burbridge, M

- Galea, S., Ahern, J., Nandi, A., Tracy, M., Beard, J., & Vlahov, D. (2007). Urban neighborhood poverty and the incidence of depression in a population based cohort study. Annals of Epidemiology, 17(3), 171-179.
- Gidlof-Gunnarsson, A., & Ohrstrom, E. (2007). Noise and well-being in urban residential environments: The potential role of perceived availability to nearby green areas. Landscape and Urban Planning, 83(2-3), 115-126
- Grigg, M., Judd, F., Komiti, A., & Ryan, L. (2008). Quality of housing among patients of an area mental health service. Australian & New Zealand Journal of Psychiatry, 42(3), 244-250
- Guite, H. F., Clark, C., & Ackrill, G. (2006). The impact of the physical and urban environment on mental well-being. Public Health, 120(12), 1117-1126.
- Henwood, B. F., Derejko, K.-S., Couture, J., & Padgett, D. K. (2015). Maslow and Mental Health Recovery: A Comparative Study of Homeless Programs for Adults with Serious Mental Illness. Administration and policy in mental health, 42(2), 220-228.
- Hogan, M. F., & Carling, P. (1992). Normal housing: A key element of a supported housing approach for people with psychiatric disabilities. Community Mental Health 28(3), p215.
- Hoisington, A. J., Brenner, L. A., Kinney, K. A., Postolache, T. T., & Lowry, C. A. (2015). The microbiome of the built environment and mental health. Microbiome, 3.
- Hopton, J. L., & Hunt, S. M. (1996). Housing conditions and mental health in a disadvantaged area in Scotland. Journal of Epidemiology and Community Health, 50(1), 56-61.
- Howden-Chapman, P. L., Chandola, T., Stafford, M., & Marmot, M. (2011). The effect of housing on the mental health of older people: the impact of lifetime housing history in Whitehall II. BMC Public Health, 11, 682. Jantsch, E. (1972). Inter- and transdisciplinary university: A systems approach to education and
- innovation. Higher Education, 1(1), 7-37. doi:10.1007/BF01956879
- Johnson, R. (2005). Mental health and housing: making the links in policy, research and practice. Journal of Public Mental Health, 4(4), 21-28.
- Jones-Rounds, M. L., Evans, G. W., & Braubach, M. (2014). The interactive effects of housing and neighbourhood quality on psychological well-being. Journal of Epidemiology and Community Health, 68(2), 171-175.
- Keall, M., Baker, M. G., Howden-Chapman, P., Cunningham, M., & Ormandy, D. (2010). Assessing housing quality and its impact on health, safety and sustainability. Journal of Epidemiology and Community Health, 64(9)
- Kearns, A., Whitley, E., Mason, P., & Bond, L. (2012). 'Living the High Life'? Residential, Social and Psychosocial Outcomes for High-Rise Occupants in a Deprived Context. Housing Studies, 27(1), 97-126. doi:10.1080/02673037.2012.632080
- Kearns, A., Whitley, E., Tannahill, C., & Ellaway, A. (2015). Loneliness, social relations and health & well-being in deprived communities. Psychology Health & Medicine, 20(3), 332-344.
- MacLeod, K. E., Gee, G. C., Crawford, P., & Wang, M. C. (2008). Neighbourhood environment as a predictor of television watching among girls. Journal of Epidemiology and Community Health, 62(4), 288-292.
- Maller, C. J., Henderson-Wilson, C., & Townsend, M. (2009). Rediscovering Nature in Everyday Settings: Or How to Create Healthy Environments and Healthy People. Ecohealth, 6(4), 553-556.
- Maller, C. J., & Strengers, Y. (2011). Housing, heat stress and health in a changing climate: promoting the adaptive capacity of vulnerable households, a suggested way forward. Health Promotion International, 26(4), 492-98.
- Marais, L., & Cloete, J. (2014). "Dying to get a house?" The health outcomes of the South African low-income housing programme. Habitat International, 43, 48-60.
- Marais, L., Sharp, C., Pappin, M., Lenka, M., Cloete, J., Skinner, D., & Serekoane, J. (2013). Housing conditions and mental health of orphans in South Africa. Health & Place, 24, 23-29
- Marcheschi, E., Johansson, M., Laike, T., & Brunt, D. (2016). Housing design and people with severe mental illness: An observational approach to the investigation of supported housing facilities. Scandinavian Journal of Psychology, 57(1), 12-21.
- Marquet, D. (2012). Turn the ship around a true story of turning followers into leaders. New York: Portfolio.
- Martin, E. J. (2015). Affordable housing, homelessness and mental health: what health care policy needs to address. Journal of Health & Human Services Admin, 38(1), 67-89.
- McCarthy, P., Byrne, D., Harrison, S., & Keithley, J. (1985). Housing type, housing location and mental health. Social psychiatry, 20(3), 125-130. doi:10.1007/bf00583578
- Mental Health Commission of Canada (2016) The Working Mind: Workplace Mental Health and Wellness Summary, Ontario Canada. Accessed 31 May 2016. http://www.mentalhealthcommission.ca/English/initiatives-and-projects/workingmind#sthash.Al0uWtuQ.dpuf

- Mridha, M. (2015). Living in an apartment. Journal of Environmental Psychology, 43, 42-54.
- Nelson, G., Walsh-Bowers, R., & Hall, G. B. (1998). Housing for Psychiatric Survivors: Values, Policy, and Research. Administration and Policy in Mental Health and Mental Health Services Research, 25(4), 455-462.
- Novick, R. (1971). Physical and Mental Health Aspects of Housing Code Enforcement. Urban Lawyer, 3(4), 538-45.
- Osypuk, T. L. (2014). Shifting from policy relevance to policy translation: do housing and neighborhoods affect children's mental health? Social Psychiatry and Psychiatric Epidemiology, 50(2), 215-217. Pevalin, D. (2009). Housing repossessions, evictions and common mental illness in the UK: results from a household panel study. Journal of Epidemiology and Community Health, 63(11), 949-951. Pevalin, D. J., Taylor, M. P., & Todd, J. (2008). The dynamics of unhealthy housing in the UK: A
- panel data analysis. Housing Studies, 23(5), 679-695.
- Shenassa, E. D., Daskalakis, C., Liebhaber, A., Braubach, M., & Brown, M. (2007). Dampness and mold in the home and depression: An examination of mold-related illness and perceived control of one's home as possible depression pathways. American Journal of Public Health, 97(10), 1893-1899.
- Shrubsole, C., Macmillan, A., Davies, M., & May, N. (2014). 100 Unintended consequences of policies to improve the energy efficiency of the UK housing stock. Indoor and Built Environment, 23(3), 340-352. \
- Smith, B. T., Padgett, D. K., Choy-Brown, M., & Henwood, B. F. (2015). Rebuilding lives and identities: The role of place in recovery among persons with complex needs. Health & Place, 33, 109-117.
- Smith, J. (2005). Housing, Homelessness, and Mental Health in Great Britain. Int J of Mental Health, 34(2), 22-46.
- Sodelund, J., & Newman, P. (2015). Biophilic architecture: a review of the rationale and outcomes. AIMS Environmental Science, 2(4), 950 - 969.
- Solari, C. D., & Mare, R. D. (2012). Housing crowding effects on children's wellbeing. Social Science Research, 41(2), 464-476. doi:10.1016/j.ssresearch.2011.09.012
- Suglia, S. F., Duarte, C. S., Sandel, M. T., & Wright, R. J. (2010). Social and environmental stressors in the home and childhood asthma. J of Epidemiology and Community Health, 64(7), 636-642.
- Suglia, S. F., Chambers, E., & Sandel, M. T. (2015). Poor Housing Quality and Housing Instability. Social Determinants of Mental Health, 171-192.
- Sylvestre, J., Nelson, G., Sabloff, A., & Peddle, S. (2007). Housing for people with serious mental illness: A comparison of values and research. American Journal of Community Psychology, 40(1-2), 125-137.
- Thatcher, A., & Milner, K. (2012). The impact of a 'green' building on employees' physical and psychological wellbeing. Work-a Journal of Prevention Assessment & Rehabilitation, 41, 3816-3823.
- van den Berg, M., M. H. E., Maas, J., Muller, R., Braun, A., Kaandorp, W., van den Berg, A. E. (2015). Autonomic Nervous System Responses to Viewing Green and Built Settings: Differentiating Between Sympathetic and Parasympathetic Activity. Int J of Environmental Research and Public Health, 12(12), 15860-15874
- van der Schaaf, P. S., Dusseldorp, E., Keuning, F. M., Janssen, W. A., & Noorthoom, E. O. (2013). Impact of the physical environment of psychiatric wards on the use of seclusion. British Journal of Psychiatry, 202(2), 142-149. doi:10.1192/bjp.bp.112.118422
- van Kempen, E., Devilee, J., Swart, W., & van Kamp, I. (2014). Characterizing urban areas with good sound quality: Development of a research protocol. Noise & Health, 16(73), 380-387. doi:10.4103/1463-1741.144416
- Veitch, J. A. (2008). Investigating and Influencing How Buildings Affect Health: Interdisciplinary Endeavours. Canadian Psychology-Psychologie Canadienne, 49(4), 281-288.
- Walker, L. A. (2015). The Cost of Good Intentions: Thermal Discomfort in Traditional Public Housing Units With Preset Thermostats. Housing Policy Debate, 25(1), 152-178.
- Wells, N. M., Evans, G. W., & Yang, Y. (2010). ENVIRONMENTS AND HEALTH: PLANNING DECISIONS AS PUBLIC-HEALTH DECISIONS. Journal of Architectural and Planning Research, 27(2), 124-143.
- Wells, N. M., & Harris, J. D. (2007). Housing quality, psychological distress, and the mediating role of social withdrawal: A longitudinal study of low-income women. Journal of Environmental Psychology, 27(1), 69-78.

# Appendix C Copyright

# C.1 Publication 1

<u>Burbridge, M.</u> & G. Morrison (2021) A systematic literature review of partnership development at the university-industry-government nexus. Sustainability 13, no. 24: 13780. <u>https://doi.org/10.3390/su132413780</u>

reference sustainability	Open Access Review
Submit to this Journal	A Systematic Literature Review of Partnership
Review for this Journal	Development at the University–Industry–Government Nexus
Edit a Special Issue	by 🔒 Mike Burbridge ^{• 🖂} and 🔒 Gregory M. Morrison 🖂 💿
rticle Menu	Curtin University Sustainability Policy Institute, School of Design and the Built Environment, Curtin University, Perth 6102, Australia
rticle Overview	Author to whom correspondence should be addressed.
Abstract	Academic Editor: Andrea Pérez
Open Access and Permissions	Sustainability 2021, 13(24), 13780; https://doi.org/10.3390/su132413780
Share and Cite Article Metrics	Received: 16 October 2021 / Revised: 2 December 2021 / Accepted: 8 December 2021 / Published: 14 December 2021
Order Article Reprints	(This article belongs to the Section Economic and Business Aspects of Sustainability)
ticle Versions	View Full-Text Download PDF Browse Figures Citation Export
elated Info Links	×
ore by Authors Links	Abstract
	The increasingly entrepreneurial intent of universities implies the commercialization of knowledge and innovation through the tr helix of interactions between universities, industry and government. However, there remains a lack of clarity concerning best reactions and reactive for innovation. This endership is increased with the clark the during the discussion of the clark the during the during the discussion of the clark the during the dur
bstract Views	particle partnerships of microsoft. This systematic instantic review (SLC) polytocies magnits on one development of partnerships at the university-industry-government nexus and builds on the existing top-downbottm-up approach for the creation of intermediaries of innovation. The SLR describes the evolution of these intermediaries, which is driven both by criteri 622 set by partners and the globalization of the knowledge economy. This SLR reveals that the partnership structure most likely to further economic and thorader social noals is the living lab with the intermediaries of an or innovation and co-creation. This S
III-Text Views	639 reveals that the living lab structure (and including sustainability labs and urban living labs) is the partnership structure utilized fr innovation that addresses economic, social and environmental goals. Two areas are recommended for further research. One
Topics	concerns the development of a deeper understanding of the relationship between the evolution in the structures of partnerships innovation and how it is influenced by the globalization of the economy, society and environment, and changing modes of knowledge production. The other is to better understand why the living lab approach to partnership creation is best suited to the delivery of sustainable development objectives and how this learning can be applied to other models of partnership development at the university-industry-government nexus. View Full-Text Keywords: triple helix; innovation; partnership development; sustainable development
Construction and	(c) This is an open access article distributed under the Creative Commons Attribution License which permits unrestricted used distribution and exercise and the access article distribution and the second units of the second

# C.2 Publication 2

<u>Burbridge, M.</u>, G Morrison, M. van Rijn, S. Silvester, D. Keyson, L. Virdee, & C. Baedeker
(2016). Business models for sustainability in living labs. In: Keyson D., Guerra-Santin
O., Lockton D. (eds) Living Labs. Springer, Cham. <u>https://doi.org/10.1007/978-3-319-33527-8_30</u>

	SPRINGER NATURE LICENSE TERMS AND CONDITIONS
	Apr 12, 2022
This Agreement between Mr. N he terms and conditions provi	like Burbridge ("You") and Springer Nature ("Springer Nature") consists of your license details and ded by Springer Nature and Copyright Clearance Center.
License Number	5286800041189
License date	Apr 12, 2022
Licensed Content Publisher	Springer Nature
Licensed Content Publication	Springer eBook
Licensed Content Title	Business Models for Sustainability in Living Labs
Licensed Content Author	Mike Burbridge, Gregory M. Morrison, Menno van Rijn et al
Licensed Content Date	Jan 1, 2017
Type of Use	Thesis/Dissertation
Requestor type	academic/university or research institute
Format	print and electronic
Portion	full article/chapter
Will you be translating?	no
Circulation/distribution	1 - 29
Author of this Springer Nature content	yes
Title	Low carbon transition and innovation at the university-industry-government nexus
Institution name	Curtin University
Expected presentation date	Apr 2022
Order reference number	Publication 2
Requestor Location	Mr. Mike Burbridge 23 John Street
	Gooseberry Hill, 6076 Australia Attn: MCB
Total	0.00 AUD
Terms and Conditions	
	Springer Nature Customer Service Centre GmbH
This agreement sets out the	Terms and conditions

This agreement sets out the terms and conditions of the licence (the Licence) between you and Springer Nature Customer Service Centre GmbH (the Licensor). By clicking 'accept' and completing the transaction for the material (Licensed Material), you also confirm your acceptance of these terms and conditions.

#### 1. Grant of License

1. 1. The Licensor grants you a personal, non-exclusive, non-transferable, world-wide licence to reproduce the Licensed Material for the purpose specified in your order only. Licences are granted for the specific use requested in the order and for no other use, subject to the conditions below.

**1. 2.** The Licensor warrants that it has, to the best of its knowledge, the rights to license reuse of the Licensed Material. However, you should ensure that the material you are requesting is original to the Licensor and does not carry the copyright of another entity (as credited in the published version).

1. 3. If the credit line on any part of the material you have requested indicates that it was reprinted or adapted with permission from another source, then you should also seek permission from that source to reuse the material.

#### 2. Scope of Licence

2. 1. You may only use the Licensed Content in the manner and to the extent permitted by these Ts&Cs and any applicable laws.

**2. 2.** A separate licence may be required for any additional use of the Licensed Material, e.g. where a licence has been purchased for print only use, separate permission must be obtained for electronic re-use. Similarly, a licence is only valid in the language selected and does not apply for editions in other languages unless additional translation rights have been granted separately in the licence. Any content owned by third parties are expressly excluded from the licence.

2. 3. Similarly, rights for additional components such as custom editions and derivatives require additional permission and may be subject to an additional fee. Please apply to

Journalpermissions@springernature.com/bookpermissions@springernature.com for these rights.

2. 4. Where permission has been granted free of charge for material in print, permission may also be granted for any electronic version of that work, provided that the material is incidental to your work as a whole and that the electronic version is essentially equivalent to, or substitutes for, the print version.

2. 5. An alternative scope of licence may apply to signatories of the <u>STM Permissions Guidelines</u>, as amended from time to time.

#### 3. Duration of Licence

3. 1. A licence for is valid from the date of purchase ('Licence Date') at the end of the relevant period in the below table:

Scope of Licence	Duration of Licence
Post on a website	12 months
Presentations	12 months
Books and journals	Lifetime of the edition in the language purchased

#### 4. Acknowledgement

**4. 1.** The Licensor's permission must be acknowledged next to the Licenced Material in print. In electronic form, this acknowledgement must be visible at the same time as the figures/tables/illustrations or abstract, and must be hyperlinked to the journal/book's homepage. Our required acknowledgement format is in the Appendix below.

#### 5. Restrictions on use

5. 1. Use of the Licensed Material may be permitted for incidental promotional use and minor editing privileges e.g. minor adaptations of single figures, changes of format, colour and/or style where the adaptation is credited as set out in Appendix 1 below. Any other changes including but not limited to, cropping, adapting, omitting material that affect the meaning, intention or moral rights of the author are strictly prohibited.

5. 2. You must not use any Licensed Material as part of any design or trademark.

5. 3. Licensed Material may be used in Open Access Publications (OAP) before publication by Springer Nature, but any Licensed Material must be removed from OAP sites prior to final publication.

#### 6, Ownership of Rights

6. 1. Licensed Material remains the property of either Licensor or the relevant third party and any rights not explicitly granted herein are expressly reserved.

#### 7. Warranty

IN NO EVENT SHALL LICENSOR BE LIABLE TO YOU OR ANY OTHER PARTY OR ANY OTHER PERSON OR FOR ANY SPECIAL, CONSEQUENTIAL, INCIDENTAL OR INDIRECT DAMAGES, HOWEVER CAUSED, ARISING OUT OF OR IN CONNECTION WITH THE DOWNLOADING, VIEWING OR USE OF THE MATERIALS REGARDLESS OF THE FORM OF ACTION, WHETHER FOR BREACH OF CONTRACT, BREACH OF WARRANTY, TORT, NEGLIGENCE, INFRINGEMENT OR OTHERWISE (INCLUDING, WITHOUT LIMITATION, DAMAGES BASED ON LOSS OF PROFITS, DATA, FILES, USE, BUSINESS OPPORTUNITY OR CLAIMS OF THIRD PARTIES), AND

WHETHER OR NOT THE PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES. THIS LIMITATION SHALL APPLY NOTWITHSTANDING ANY FAILURE OF ESSENTIAL PURPOSE OF ANY LIMITED REMEDY PROVIDED HEREIN.

#### 8. Limitations

8. 1. <u>BOOKS ONLY:</u>Where 'reuse in a dissertation/thesis' has been selected the following terms apply: Print rights of the final author's accepted manuscript (for clarity, NOT the published version) for up to 100 copies, electronic rights for use only on a personal website or institutional repository as defined by the Sherpa guideline (<u>www.sherpa.ac.uk/romeo/</u>).

8.2. For content reuse requests that qualify for permission under the <u>STM Permissions Guidelines</u>, which may be updated from time to time, the STM Permissions Guidelines supersede the terms and conditions contained in this licence.

#### 9. Termination and Cancellation

9. 1. Licences will expire after the period shown in Clause 3 (above).

9. 2. Licensee reserves the right to terminate the Licence in the event that payment is not received in full or if there has been a breach of this agreement by you.

#### Appendix 1 — Acknowledgements:

#### For Journal Content:

Reprinted by permission from [the Licensor]: [Journal Publisher (e.g. Nature/Springer/Palgrave)] [JOURNAL NAME] [REFERENCE CITATION (Article name, Author(s) Name), [COPYRIGHT] (year of publication)

#### For Advance Online Publication papers:

Reprinted by permission from [the Licensor]: [Journal Publisher (e.g. Nature/Springer/Palgrave)] [JOURNAL NAME] [REFERENCE CITATION (Article name, Author(s) Name), [COPYRIGHT] (year of publication), advance online publication, day month year (doi: 10.1038/sj.[JOURNAL ACRONYM].)

#### For Adaptations/Translations:

Adapted/Translated by permission from [the Licensor]: [Journal Publisher (e.g. Nature/Springer/Palgrave)] [JOURNAL NAME] [REFERENCE CITATION (Article name, Author(s) Name), [COPYRIGHT] (year of publication)

#### Note: For any republication from the British Journal of Cancer, the following credit line style applies:

Reprinted/adapted/translated by permission from [the Licensor]: on behalf of Cancer Research UK: : [Journal Publisher (e.g. Nature/Springer/Palgrave)] [JOURNAL NAME] [REFERENCE CITATION (Article name, Author(s) Name), [COPYRIGHT] (year of publication)

#### For Advance Online Publication papers:

Reprinted by permission from The [the Licensor]: on behalf of Cancer Research UK: [Journal Publisher (e.g. Nature/Springer/Palgrave)] [JOURNAL NAME] [REFERENCE CITATION (Article name, Author(s) Name), [COPYRIGHT] (year of publication), advance online publication, day month year (doi: 10.1038/sj.[JOURNAL ACRONYM])

#### For Book content:

Reprinted/adapted by permission from [the Licensor]: [Book Publisher (e.g. Palgrave Macmillan, Springer etc) [Book Title] by [Book author(s)] [COPYRIGHT] (year of publication)

#### Other Conditions:

Version 1.3

Questions? customercare@copyright.com or +1-855-239-3415 (toll free in the US) or +1-978-646-2777.

# C.3 Publication 3

<u>Burbridge, M</u> (2017). If living labs are the answer – what's the question? Procedia Engineering, Vol 180 pp 1725 – 1732

x	If Living Labs are the Answer – What's the Question? A F	eview of the Literat	ure		
rocedia	Author: Mike Burbridge				
	Publication: Procedia Engineering				
	Publisher: Elsevier				
hanne over a	Date: 2017				
kierostiteet monorosteron	© 2017 The Author(s). Published by Elsevier Ltd.				
urnal Author	Dieber				
unial Autio	Rights				
ease note that,	as the author of this Elsevier article, you retain the right to include it in a thes please ensure that you reference the journal as the original source. For more	is or dissertation, provided information on this and o	d it is not pub n your other	lished commercia retained rights, p	ally. Permission is lease
isit: https://www	elsevier.com/about/our-business/policies/copyright#Author-rights.				

# C.4 Publication 4

Burbridge, M. (2018). A Study of Australian Universities' Collective Response to Climate Science. In: Sayigh A. (eds) Transition Towards 100% Renewable Energy. Innovative Renewable Energy. Springer, Cham. https://doi.org/10.1007/978-3-319-69844-1

	Apr 12, 20
s Agreement between Mr. N terms and conditions provid	like Burbridge ("You") and Springer Nature ("Springer Nature") consists of your license details and ded by Springer Nature and Copyright Clearance Center.
License Number	5210690034196
License date	Dec 16, 2021
Licensed Content Publisher	Springer Nature
Licensed Content Publication	Springer eBook
Licensed Content Title	Do as I Say; Don't Do as I Do, Let Alone Do as I've Done. A Study of Australian Universities' Collective Response to Climate Science
Licensed Content Author	Mike Burbridge
Licensed Content Date	Jan 1, 2018
Type of Use	Thesis/Dissertation
Requestor type	academic/university or research institute
Format	print and electronic
Portion	full article/chapter
Will you be translating?	no
Circulation/distribution	1 - 29
Author of this Springer Nature content	yes
Title	Low carbon transition and innovation at the university-industry-government nexus
Institution name	Curtin University
Expected presentation date	Dec 2021
Requestor Location	Mr. Mike Burbridge 23 John Street
	Gooseberry Hill, 6076 Australia
Tetal	
Total	0.00 AOD
ierms and Conditions	Springer Nature Customer Service Centre CmbH
	Springer Nature Customer Service Centre Ombri Torma and Conditions

1, Grant of License

**1.1.** The Licensor grants you a personal, non-exclusive, non-transferable, world-wide licence to reproduce the Licensed Material for the purpose specified in your order only. Licences are granted for the specific use requested in the order and for no other use, subject to the conditions below.

**1. 2.** The Licensor warrants that it has, to the best of its knowledge, the rights to license reuse of the Licensed Material. However, you should ensure that the material you are requesting is original to the Licensor and does not carry the copyright of another entity (as credited in the published version).

**1.3.** If the credit line on any part of the material you have requested indicates that it was reprinted or adapted with permission from another source, then you should also seek permission from that source to reuse the material.

#### 2. Scope of Licence

2. 1. You may only use the Licensed Content in the manner and to the extent permitted by these Ts&Cs and any applicable laws.

**2. 2.** A separate licence may be required for any additional use of the Licensed Material, e.g. where a licence has been purchased for print only use, separate permission must be obtained for electronic re-use. Similarly, a licence is only valid in the language selected and does not apply for editions in other languages unless additional translation rights have been granted separately in the licence. Any content owned by third parties are expressly excluded from the licence.

2. 3. Similarly, rights for additional components such as custom editions and derivatives require additional permission and may be subject to an additional fee. Please apply to <u>Journalpermissions@springernature.com/bookpermissions@springernature.com</u> for these rights.

2. 4. Where permission has been granted **free of charge** for material in print, permission may also be granted for any electronic version of that work, provided that the material is incidental to your work as a whole and that the electronic version is essentially equivalent to, or substitutes for, the print version.

2. 5. An alternative scope of licence may apply to signatories of the <u>STM Permissions Guidelines</u>, as amended from time to time.

#### 3. Duration of Licence

3. 1. A licence for is valid from the date of purchase ('Licence Date') at the end of the relevant period in the below table:

Scope of Licence	Duration of Licence
Post on a website	12 months
Presentations	12 months
Books and journals	Lifetime of the edition in the language purchased

#### 4, Acknowledgement

4. 1. The Licensor's permission must be acknowledged next to the Licenced Material in print. In electronic form, this acknowledgement must be visible at the same time as the figures/tables/illustrations or abstract, and must be hyperlinked to the journal/book's homepage. Our required acknowledgement format is in the Appendix below.

#### 5. Restrictions on use

5. 1. Use of the Licensed Material may be permitted for incidental promotional use and minor editing privileges e.g. minor adaptations of single figures, changes of format, colour and/or style where the adaptation is credited as set out in Appendix 1 below. Any other changes including but not limited to, cropping, adapting, omitting material that affect the meaning, intention or moral rights of the author are strictly prohibited.

5. 2. You must not use any Licensed Material as part of any design or trademark.

5. 3. Licensed Material may be used in Open Access Publications (OAP) before publication by Springer Nature, but any Licensed Material must be removed from OAP sites prior to final publication.

#### 6. Ownership of Rights

6. 1. Licensed Material remains the property of either Licensor or the relevant third party and any rights not explicitly granted herein are expressly reserved.

#### 7. Warranty

IN NO EVENT SHALL LICENSOR BE LIABLE TO YOU OR ANY OTHER PARTY OR ANY OTHER PERSON OR FOR ANY SPECIAL, CONSEQUENTIAL, INCIDENTAL OR INDIRECT DAMAGES, HOWEVER CAUSED, ARISING OUT OF OR IN CONNECTION WITH THE DOWNLOADING, VIEWING OR USE OF THE MATERIALS REGARDLESS OF THE FORM OF ACTION, WHETHER FOR BREACH OF CONTRACT, BREACH OF WARRANTY, TORT, NEGLIGENCE, INFRINGEMENT OR OTHERWISE (INCLUDING, WITHOUT LIMITATION, DAMAGES BASED ON LOSS OF PROFITS, DATA, FILES, USE, BUSINESS OPPORTUNITY OR CLAIMS OF THIRD PARTIES), AND

WHETHER OR NOT THE PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES. THIS LIMITATION SHALL APPLY NOTWITHSTANDING ANY FAILURE OF ESSENTIAL PURPOSE OF ANY LIMITED REMEDY PROVIDED HEREIN.

#### 8. Limitations

8. 1. <u>BOOKS ONLY</u>: Where 'reuse in a dissertation/thesis' has been selected the following terms apply: Print rights of the final author's accepted manuscript (for clarity, NOT the published version) for up to 100 copies, electronic rights for use only on a personal website or institutional repository as defined by the Sherpa guideline (<u>www.sherpa.ac.uk/romeo/</u>).

8.2. For content reuse requests that qualify for permission under the <u>STM Permissions Guidelines</u>, which may be updated from time to time, the STM Permissions Guidelines supersede the terms and conditions contained in this licence.

#### 9. Termination and Cancellation

9. 1. Licences will expire after the period shown in Clause 3 (above).

9.2. Licensee reserves the right to terminate the Licence in the event that payment is not received in full or if there has been a breach of this agreement by you.

#### Appendix 1 — Acknowledgements:

#### For Journal Content:

Reprinted by permission from [the Licensor]: [Journal Publisher (e.g. Nature/Springer/Palgrave)] [JOURNAL NAME] [REFERENCE CITATION (Article name, Author(s) Name), [COPYRIGHT] (year of publication)

#### For Advance Online Publication papers:

Reprinted by permission from [the Licensor]: [Journal Publisher (e.g. Nature/Springer/Palgrave)] [JOURNAL NAME] [REFERENCE CITATION (Article name, Author(s) Name), [COPYRIGHT] (year of publication), advance online publication, day month year (doi: 10.1038/sj.[JOURNAL ACRONYM].)

#### For Adaptations/Translations:

Adapted/Translated by permission from [the Licensor]: [Journal Publisher (e.g. Nature/Springer/Palgrave)] [JOURNAL NAME] [REFERENCE CITATION (Article name, Author(s) Name), [COPYRIGHT] (year of publication)

#### Note: For any republication from the British Journal of Cancer, the following credit line style applies:

Reprinted/adapted/translated by permission from [the Licensor]: on behalf of Cancer Research UK: : [Journal Publisher (e.g. Nature/Springer/Palgrave)] [JOURNAL NAME] [REFERENCE CITATION (Article name, Author(s) Name), [COPYRIGHT] (year of publication)

#### For Advance Online Publication papers:

Reprinted by permission from The [the Licensor]: on behalf of Cancer Research UK: [Journal Publisher (e.g. Nature/Springer/Palgrave)] [JOURNAL NAME] [REFERENCE CITATION (Article name, Author(s) Name), [COPYRIGHT] (year of publication), advance online publication, day month year (doi: 10.1038/sj.[JOURNAL ACRONYM])

#### For Book content:

Reprinted/adapted by permission from [the Licensor]: [Book Publisher (e.g. Palgrave Macmillan, Springer etc) [Book Title] by [Book author(s)] [COPYRIGHT] (year of publication)

#### Other Conditions:

#### Version 1.3

Questions? customercare@copyright.com or +1-855-239-3415 (toll free in the US) or +1-978-646-2777.

# C.5 Publication 5

 <u>Burbridge, M.</u> & G Morrison (2016). Happy Homes – the Relationship between Homes and mental wellbeing. Proceedings of the 7th International Conference on Energy and Environment of Residential Buildings, edited by Miller, W., Susilawati, C. and Manley, K. Brisbane: Queensland University of Technology, Australia.

Resource > Happy homes – the relationship between homes and mental wellbeing	: a review of the literature
CONFERENCE PAPERHappy homes – the relationship between homes and mental wellbeing: a review of the literature22 NOV 2016MikeBurdinge, Gregory MorrisonPulisHERThe International Conference on Energy and Environment of Residential Buildings, 20-264, brownen 2016, Brisbane, AustraliaHousing Housing and health (Mental health) (Urban planning)Hutidisciplinary collaboration (Translational research)	Add to favourites     EndNote citation     RIS citation     Collections     Healthy Housing 2016 Conference     Low Carbon Living     Urban Policy & Practice
RESOURCES ↓ apo-nid68813.pdf 512.61 KB	More like this
DESCRIPTION Purpose: This paper set out to uncover the advice available to help people take effective action within our home to improve mental health. The literature and professions are virtually silent on the issue. The professional advice is often the opposite suggesting we should get out of our homes - go for a walk, exercise, play sporl, go to the cinema, meet friends, socialise and don't isolate yourself. There is nary any advice about what we can do to our homes to help maintain our mental health. Our home - the physical space where we spend large amounts of energy and time is largely an empty shell for the mental health industry. The message	A place to call my own: identifying best practice in housing and mental health
<b>APO 20</b> ¹⁰⁰	
productive sources, rective contention for surrelympt spire, second, or good notising looks like.	
Approach: This paper will begin that dialogue with a comprehensive literature review. The approach adopted to investigate this literature focussed on thinking about what a policy official might experience if they were tasked to develop guidance on steps to improve housings' impact on mental health. Such an individual would not necessarily be aware of the extent of the literature, or of academic disciplines. This approach both made the literature review problematic, but also in some ways also produces a useful insight.	
Key findings: The paper concludes that there are three issues that should shape future research: first is the need for transdisciplinary translational research; second is to focus initially on the needs of the resident before the bricks and mortar; third is to endeavour to include the social pillar of sustainable	

		in the memoring	of evidence-informed policy.
			Sponsor
PUBLICATION DETA	ILS		
DOI: 10.4225/50/58107	40eae22a		
Copyright: The authors Access Rights Type: oper Post date: 24 Oct 2016	1		
CLASSIFIEDS	SERVICES	COLLABORATE	CONNECT
CLASSIFIEDS <u>Calls &amp; Notices</u>	SERVICES <u>Advertise</u>	COLLABORATE <u>APO Blog</u>	CONNECT
CLASSIFIEDS Calls & Notices Courses	SERVICES <u>Advertise</u> How to use APQ	COLLABORATE <u>APO Blog</u> <u>Contact us</u>	CONNECT Twitter LinkedIn
CLASSIFIEDS Calls & Notices Courses Events	SERVICES Advertise How to use APO	COLLABORATE APO Blog Contact us Policy & Guidelines	CONNECT Twitter LinkedIn

© 2022 Analysis & Policy Observatory (APO), licensed under a Creative Commons Attribution-Non-Commercial 3.0 Australia (CC-BY-NC 3.0 AU) License